

# KS AuthentiCare®

# <u>User Manual</u>



# Version 1.0 November 14, 2011





# **Table of Contents**

DOCU	MENT HISTORY	6
СНАРТ	TER 1 INTRODUCTION	7
1.1	COMMON TERMINOLOGY/ACRONYMS	7
1.2	OVERVIEW OF KS AUTHENTICARE	9
1.3	ORGANIZATION OF THE DATA	
1.4	KS AUTHENTICARE USERS	
1.5	SECURITY	11
1.6	USING THIS WEB APPLICATION	12
1.7	USING THIS MANUAL	14
СНАРТ	TER 2 GETTING STARTED	15
2.1	ESTABLISHING A USER ACCOUNT	15
2.1	1.1 Adding New Users	
2.2	LOGGING IN TO KS AUTHENTICARE	20
2.3	LOGGING OUT OF KS AUTHENTICARE	21
2.4	CHANGING PASSWORD	
2.5	DISABLING A USER	23
2.5	5.1 Providers	23
2.5	5.2 State Administrative Users	25
СНАРТ	<b>TER 3 AN INTRODUCTION TO NAVIGATING KS AUTHENTICARE</b>	
3.1	HOME PAGE	
3.2	PROVIDER DASHBOARD	
3.3	Worker Dashboard	
СНАРТ	TER 4 SEARCH AND VIEWING PEOPLE AND/OR AGENCIES	
4.1	SEARCHING FOR PEOPLE AND/OR AGENCIES	
4.2	VIEWING PROVIDERS	
4.3	VIEWING CLIENTS	
4.4	VIEWING WORKERS	





CHAP	TER 5 MANAGING PROVIDER INFORMATION	39
5.1	EDITING PROVIDER INFORMATION	39
5.2	835 ELECTRONIC REMITTANCE ADVICE FROM KMAP/MMIS	42
5.3	UPLOADING THE 835 INTO AUTHENTICARE	43
CHAP	TER 6 MANAGING WORKER INFORMATION	45
6.1	ADDING A WORKER	45
6.2	DELETING A WORKER	50
6.3	EDITING WORKERS	53
6.4	SUSPENDING/INACTIVATING/END DATING A WORKER	54
CHAP	TER 7 MANAGING CLIENT (CONSUMER OR CUSTOMER) INFORMATION	56
CHAPT	TER 8 SEARCH AND VIEW AUTHORIZATIONS	57
CHAP	TER 9 SCHEDULING	60
9.1	USING THE CALENDAR	60
9.1	1.1 Monthly	61
9.1	1.2 Weekly	62
9.1	1.3 Daily	64
9.2	SEARCHING FOR SCHEDULED EVENTS IN THE CALENDAR	65
9.3	SCHEDULING AN EVENT	67
9.3	3.1 Scheduling an Event Using the Authorization	67
9.3	3.2 Scheduling a Recurring Event	70
9.4	EDITING AN EVENT	74
9.5	CANCELLING AN EVENT (DELETING)	76
9.6	ACKNOWLEDGING MISSED VISITS	77
CHAP	TER 10 THE KS AUTHENTICARE IVR	81
10.1	IVR FLOW	81
10.2	EXTRA SAFEGUARD FOR WORKERS WITH MORE THAN ONE WORKER ID	82
10.3	WORKER INSTRUCTIONS FOR USING THE IVR	83
10.4	ACTIVITY CODES REQUIRED FOR SPECIFIED SERVICES	85
CHAP	TER 11 GPS ENABLED MOBILE DEVICE	86
CHAP	TER 12 MANAGING CLAIMS	87
Novembe	r 14, 2011 KS AuthentiCare User Manual	





12.1	Se	ARCHING AND VIEWING CLAIMS	
12.2	EDI	TING A CLAIM (CLAIMS CORRECTION)	94
12.3	Adi	DING AN INDIVIDUAL (STANDARD) CLAIM	95
12.4	Adi	DING MULTIPLE CLAIMS (EXPRESS ENTRY)	100
12.5	DEI	LETING A CLAIM	102
12.6	Со	NFIRMING A SINGLE CLAIM FOR BILLING	103
12.7	Со	NFIRMING CLAIMS IN BULK	107
12.8	UN	CONFIRMING CLAIMS FOR FURTHER EDITING	108
СНАРТЕ	ER 13	3 REPORTING	109
13.1	CR	EATING A REPORT	109
13.2	VIE	WING A REPORT	114
13.3	Dei	LETING A REPORT FROM THE VIEW REPORTS SECTION	116
13.4	Usi	NG REPORT TEMPLATES	117
13.4	.1	Creating a Report Template	
13.4	.2	Running a Report from a Template	
13.4	.3	Editing a Report Template	121
13.4	.4	Deleting a Report Template	122
13.5	Rei	PORT EXAMPLES	
13.5	.1	Authorizations Report without Claim Detail	123
13.5	.2	Authorizations Report with Claim Detail	124
13.5	.3	Claim Details Report	125
13.5	.4	Claim History Report	127
13.5	.5	Calendar Report	
13.5	.6	Late and Missed Visits Report	
13.5	.7	Exception Report	
13.5	.8	Unauthorized Phone Number Report	
13.5	.9	Billing Invoice Report	
13.5	.10	Time and Attendance Report	
13.5	.11	Overlapped Claim Report	
13.5	.12	Workers by Provider Report	
November 1	14, 201	1 KS AuthentiCare User Manual	





13.5.13	Provider Activity Report	146
13.5.14	Remittance Advice Report	149
13.5.15	Claim Data Listing Report	151
13.5.16	Remittance Data List Report	152
13.5.17	Eligible Client Report	153
CHAPTER 14	ONGOING USER SUPPORT AND TRAINING	155
APPENDIX A	1: SERVICE CODES	156
APPENDIX A	A2: ACTIVITY CODES FOR SELECTED SERVICES	158
	DDES FOR ATTENDANT CARE SERVICES LEVEL I, LEVEL II, AND SELF-DIRECTED IN THRLY WAIVER	
	DDES FOR PERSONAL ASSISTANT SERVICES (T1019) IN THE MR/DD WAIVER AND ERVICES (S5126) IN THE PD AND TBI WAIVERS	159
	DDES FOR LPN (T1000) AND RN (T1000TD) SPECIALIZED MEDICAL CARE SERVIC	
ACTIVITY C	DDES FOR PERSONAL CARE ATTENDANT	161
SRS – TECH	HNOLOGY ASSISTED WAIVER	161
APPENDIX A	.3: CLAIM EXCEPTION CODES	162
CRITICAL EX	(CEPTIONS	162
INFORMATIC	NAL (NON-CRITICAL) EXCEPTIONS	164
APPENDIX A	.4: DESCRIPTION OF PROVIDER SUB-ROLES	167





#### **DOCUMENT HISTORY**

This document is controlled through the Document Management Process. To verify that the document is the latest version, please contact First Data Client Support at <u>clientsupport@firstdata.com</u>.

Date	Version	Responsible	Reason for Revision
November 14, 2011	1.0	First Data	Initial Submission





## Chapter 1 INTRODUCTION

The Kansas Department of Aging (KDOA) uses KS AuthentiCare for home and community based services (HCBS) clients (consumers or customers) receiving care under the Frail Elderly Waiver.

The Kansas Department of Social and Rehabilitation Services (SRS) uses KS AuthentiCare for HCBS clients (consumers or customers) receiving care under the Physical Disability (PD) Waiver, the MR/DD Waiver, the Traumatic Brain Injury (TBI) Waiver and the Technology Assisted (TA) Waiver.

Providers of selected services (procedures) under each waiver are required to use KS AuthentiCare to capture and bill for services provided to clients. (See Appendix A.1 for the list of included services.)

KS AuthentiCare is a web-based electronic scheduling, tracking, reporting and billing system that:

- Automatically captures and invoices accurate services provided to clients,
- Provides the ability to automate scheduling, time and attendance, billing and reporting functions,
- Assists in identifying and responding to unmet client needs (missed and late visits), and
- Provides KDOA and SRS with data to assist in making policy decisions regarding the delivery of home and community based care.

#### **<u>1.1</u>** COMMON TERMINOLOGY/ACRONYMS

Users must be familiar with the terms and acronyms used in KS AuthentiCare in order to take full advantage of the functionality and follow the instructions in this User Manual. These include:

Term/Acronym	Meaning/Use
835         The electronic remittance advice (RA) that KMAP/MMIS provides to each reporting the adjudication status of each claim submitted	
837	The electronic billing file that KS AuthentiCare submits each weekday morning to KMAP/MMIS on behalf of each provider
Activities	Those specific tasks that a worker performs for the client that make up the service (procedure). For a list of services that require activity codes and the specific activity codes, see Appendix A.2 of this User Manual.





Term/Acronym	Meaning/Use
Case Manager	Responsible for assessing eligibility, referring to needed services, monitoring services, follow-up, and Plan of Care (POC) entry into KMAP/MMIS to create the authorizations for services to be provided to the client. (Includes Targeted Case Managers)
Claim	Each episode of service captured in KS AuthentiCare. Each will have a unique claim number auto-assigned by the system. Claims may be entered into KS AuthentiCare by the IVR, web or mobile device.
Client	The individual receiving services under the applicable waiver; usually known as the consumer, the customer or the beneficiary.
Client Services	The First Data help desk that can assist KS AuthentiCare users. Contact information for them is located in Chapter 14 of this User Manual.
Event	For those providers who use the Scheduling functionality in KS AuthentiCare, an event is a visit scheduled in advance for service for a client.
First Data	The company that operates KS AuthentiCare for the State of Kansas.
IVR	Interactive Voice Response system used by KS AuthentiCare; pre-recorded information that the Worker interacts with via touch tone phone when calling a designated toll-free number. For more information, see Chapter 10 of this User Manual.
Mobile Device	A GPS enabled mobile device (smart phone) can be used to record services provided for a client. For more information, see Chapter 11 of this User Manual.
MMIS	Also known as Kansas Medical Assistance Program (KMAP) or HP; the entity that maintains information on enrolled clients and providers and that processes claims submitted for payment. Much of the data in KS AuthentiCare is from the KMAP/MMIS system. KMAP/MMIS updates data in KS AuthentiCare each week night via secure file transfer.
Provider	The provider agency or FMS provider that is authorized to provide services for a client. Each provider has a unique Medicaid provider ID (9-digit number plus a letter). If a provider company has multiple locations, each is considered a unique provider as each has a unique Medicaid provider ID + service location indicator (letter suffix).
Service	The procedure provided for the client under the waiver. For a list of the services where the provider must use KS AuthentiCare, see Appendix A.1 of this User Manual.





Term/Acronym	Meaning/Use
State Staff or State	State staff who will have access to information in KS AuthentiCare are staff at the KDOA and SRS who oversee the Physical Disability (PD), MR/DD, Traumatic Brain Injury (TBI) and Technology Assisted (TA) Waivers.
Sub-Role	The KS AuthentiCare initial administrator user at each provider location/FMS provider will create additional users and assign them a sub-role which defines what information they can access and what actions they can perform on the KS AuthentiCare website. The sub-role each one uses is assigned allows them to do their designated work while assuring that all data is maintained in a private and secure manner. For a description of each sub-role, see Appendix A.4 of this User Manual.
Worker	The Worker is the employee of the provider who actually provides the service to the client. May also be known as Attendant, PA (Personal Assistant) or Direct Support Worker (in self-directed care). Each worker is assigned a unique 5-digit Worker ID number for each provider and/or provider location where they work to use when recording services via the IVR, web or mobile device.

#### **1.2** OVERVIEW OF KS AUTHENTICARE

The core of KS AuthentiCare is a database containing information on the clients, services, authorizations, providers and workers. This information is updated each week day via file transfer from KMAP/MMIS or web entry as needed by provider staff. The basic use of KS AuthentiCare requires these steps:

- 1. The worker goes to the home of the client to provide a service.
- 2. The worker uses the client's touch-tone phone to call the toll-free KS AuthentiCare number or uses a GPS enabled mobile device.
- 3. Using caller ID or GPS technology, KS AuthentiCare identifies the client and the services authorized for that client and prompts the worker to enter his/her Worker ID number and verify the service to be provided.
- 4. The system verifies the information and advises the worker that he/she is "checked in" as of the time the contact was initiated.
- 5. When the worker completes the service, the worker calls the same toll-free number or uses the GPS enabled mobile device to "check out" and record the activities performed as appropriate. The worker is advised that he/she is "checked out" as of the end of the contact.





- 6. From that telephone/mobile device interaction, KS AuthentiCare creates a claim. After the provider reviews and approves it, the claim is submitted to KMAP/MMIS electronically for adjudication.
- 7. Claims are submitted for processing in the early morning of each week day. KMAP/MMIS makes payment directly to the provider on their normal payment schedule.

In situations where the IVR system cannot be used (example: the phone is out of order) and the worker does not have a mobile device or situations where the worker makes an error (example: forgets to check out); the worker notifies his/her supervisor and provides the information about the visit. The provider enters the visit information into KS AuthentiCare via the web, thus creating a claim for the service provided.

#### **<u>1.3</u>** ORGANIZATION OF THE DATA

The data in KS AuthentiCare is organized under four main areas:

- 1. **Entities** includes people or agencies that are involved in providing care as well as individuals receiving care. In KS AuthentiCare, the Entities are: Providers, Workers and Clients.
- 2. **Services** those procedures defined by KDOA and SRA for reporting through KS AuthentiCare are listed in Appendix A.1.
- 3. **Authorizations** specify the types and amount (units) of service that providers/workers are pre-authorized to provide based on the client's Plan of Care.
- 4. **Claims** each occasion a service was provided to a client by a worker. (Also known as a visit, this does not always equate to a claim submitted to KMAP/MMIS. Under certain pre-defined circumstances several visits may be combined to create on claim for billing purposes.) Each claim created in KS AuthentiCare has a unique claim number.

#### 1.4 KS AUTHENTICARE USERS

The user of KS AuthentiCare is assigned one of several different roles. This User Manual is designed to provide each user, regardless of role, with the ability to maximize use of the system on a daily basis.

While the client is the heart of any in home and community based care system, clients do not interact directly with KS AuthentiCare. Those who do have active roles in KS AuthentiCare are:

• The **Worker** who calls the IVR or uses a GPS mobile device to check in upon arrival at a client's home and again to check out after completing services.





- The **Provider Staff** who use the secure website to
  - Manage worker information
  - o Schedule workers' visits to clients' homes (optional)
  - Add claims for services where the use of the IVR or mobile device was not feasible
  - Modify a claim that was incorrectly entered by the worker using the IVR or mobile device
  - Confirm claims for billing prior to their submission to KMAP/MMIS
  - Acknowledge and provide an explanation of missed visits
  - View reports of real-time information to assist in record keeping and management
- The **State Administrative Staff** use the secure website to monitor care being provided to clients

KS AuthentiCare helps to reduce paperwork and gives the user tools to assist in managing staff resources, service delivery and reimbursement through access to real-time information via the Internet at any time.

#### 1.5 SECURITY

KS AuthentiCare maintains extensive security protocols to protect the information available via the website while at the same time making it readily available to authorized users. There are two levels of security controlling access to the data:

The <u>first level of security</u> is dependent on **the role** that the user has in the system. The roles in KS AuthentiCare are Provider, KDOA Administrative User and SRS Administrative User.

à	Each Provider can designate sub-roles within their agency. Each sub-role has the ability to perform specific functions within the system. See Appendix A.4 for a description of the sub-roles.
---	--

The <u>second level of security</u> is referred to as **data scoping**. Data scoping restricts what information the user has access to within their role.

- A Provider user has the access only to information associated with his/her provider Medicaid ID. A provider is not able to view the data of other providers for confidentiality reasons.
- The KDOA Administrative users can view all data in KS AuthentiCare related to clients in the Frail Elderly Waiver.

November 14, 2011





• The SRS Administrative users can view all data in KS AuthentiCare related to clients in the four SRS Waivers.

#### **1.6** USING THIS WEB APPLICATION

The KS AuthentiCare web component is accessible from any computer that has a connection to the Internet. Examples of unique functions:

• **Hyperlinks** - a hyperlink, or link, is a connection from your current location in the application to another page in the application or another web site or web application. When the mouse is placed on a hyperlink, the text may change, and the cursor changes from an arrow to a hand. Hyperlinks only require a single click. Below is an example of a text hyperlink in KS AuthentiCare.



• Pictures, graphics, or icons can also be used as hyperlinks. If the cursor arrow changes to a hand, then there is a link. In the example below, the graphic does not change in appearance but a pop up box indicates what the hyperlink does and the arrow changed to a hand.



• Web applications use a **web browser**. The web browser has the ability to maintain the web addresses of your "favorite" websites, to add an address to your favorites, to enter a web address (also known as a URL), to go back to the previous page, to go forward to the next web page assuming you have viewed it previously, to reload or refresh the current page, and to cancel the loading of the current page.





🖉 Google - Windows Internet Explorer				_ 7 🛛
C S http://www.google.com/	~	fy 🗙 yah	oo! Search	P -
File Edit Wew Favorites Took then	Refresh	] [	Charl (Canada	
Snapt Back Forward Web address	Reliesh		Stop/Cancel	»
S Google		<u>⊡</u> • 6	🛯 🔹 🌲 🔹 🔂 Page	🔹 🖓 Tools 👻
AVG - @ - @ Identity Guard				

Web applications can use breadcrumbs to assist in navigation. As the name suggests, this
provides you with a history of the web pages you used in getting to your current page in the
event that you wish to return to any of the previous web pages. The web pages identified
in the breadcrumb are hyperlinks, and by clicking on them, you will be returned to that web
page.



• **Hover Text** - additional information is displayed when the mouse cursor is placed over the hover link. This functionality is utilized in KS AuthentiCare to provide additional information rather than having to navigate to another page in the application to get the same information.

informati	g the cursor over the on icon, additional bout the claim displays. 12/28/2009 - 12/28/2009			
Additional Information		ů		
Group ID: 425				
Client: McLain Ward (023	3456896)			
Provider: Sooner Healthcare (427)				
Worker: Jennifer Alfano (4	0961)			
Service: Respite In-Home	(T1005)			
Service Type: Time Based				
(WEIGY WINCE)		_		
023456896 (McLain Ward)	12/30/2009 - 12/30/2009	£		
023456896 (McLain Ward)	01/02/2010	£		

• There are also several icons unique to KS AuthentiCare that you will see on the web pages. These include:

Schedule Event icon - used on the *Authorizations* page to schedule an event for the authorization

November 14, 2011





**Looking Glass icon** 2 - provides the ability to look up values for fields linked to other existing data in the system

**Information icon 1** - provides the ability to display more detailed data when you place the mouse over the icon (hover text)

**Entity icon** - provides a link to view the associated data for the value displaying in a certain field

Exceptions icon 🚺 - provides an ability to view the exceptions on a claim

**Run Report icon** - provides the ability to run a report from a previously created report template

#### 1.7 USING THIS MANUAL

This manual is designed to provide the information you need to use KS AuthentiCare.

Each section within a chapter provides step-by-step instructions. Each section may also have one of the following boxes:



Notes – The information provided in these boxes is intended to assist and further explain the material. It may include an important tip or hint to using the system.



Important – The information provided in these boxes highlights specific rules that are critical to the proper functioning of the system.



*Caution – The information provided in these boxes highlights actions that if taken in the system may have an adverse affect.* 

The information provided in all of these boxes is essential when using the system and should not be ignored.





# Chapter 2 GETTING STARTED

This chapter is designed to help you begin using KS AuthentiCare by walking through the provider registration process, adding new users, logging in, logging out, and changing your password.

The KS AuthentiCare website is <u>www.authenticare.com/kansas</u>. The use of an Internet Explorer web browser is preferable.

#### 2.1 ESTABLISHING A USER ACCOUNT

To log into KS AuthentiCare, you must have a user account which consists of a username (email address) and a password. There are several ways that a user may obtain a username and password.

- For all **providers**, an initial user account for an Administrator user is created by First Data and provided to a manager. Proceed to Section 2.2 for instructions on logging into KS AuthentiCare. After logging in, the Administrator user can create user accounts for other users for that provider (refer to Section 2.1.2 for further instructions).
- For state administrators, user accounts are created by First Data when authorized by the KDOA or SRS. The registration process is not required. First Data will provide your username (email address) and password so that you are able to log into and use KS AuthentiCare. Proceed to Section 2.2 for instructions on logging into KS AuthentiCare.

#### 2.1.1 ADDING NEW USERS

KS AuthentiCare allows providers with the Administrator sub-role to create additional user accounts, including another user with the Administrator sub-role.



*Important* – A State Administrative User cannot add another user. To add a new State Administrative User, contact <u>KSAuthentiCare@aging.ks.gov</u>.





Upon successful completion of logging into KS AuthentiCare, the Home page displays.

Kansas		AuthentiCare	Ð
Home Create Reports Scheduling Dasi	nboards   Visits   Administration   My Account   Cust		Logged in as: acr_admin@acr.com
	Entities	Claims	
	Add New > <u>Client</u> <u>Provider</u> <u>Worker</u> CaseManager	Add New> <u>Claim (Standard)</u> Add New> <u>Claim (Express)</u>	
	Representative	⊙ Claim	
	Search > Go!	Search Type: O Confirm Billing - View O Confirm Billing - Bulk	
		Claim Group ID	
	Services and Authorizations	Claim ID	
		Claim Status:	
	Add New > Service	Claim Start	
	Search Type: O Authorization	Service	
	Service	Authorization ID	
	Authorization ID	Client:	
	Service Type	Provider	
	Authorization Start	Worker:	
	Authorization End:	CaseManager:	
	Client: Provider:	Representative:	
	Worker:	User Option:	
	Procedure Code	Go! Clear	

1. Enter the name of your provider or some portion of the name in the Entities **Search** > field.

Entities	
Add New >	<u>Client</u> <u>Provider</u> <u>Worker</u> <u>CaseManager</u> <u>Representative</u>
Search >	Go!

2. Click Go!

The *Provider Entity Settings* page displays with the information about this provider.

```
November 14, 2011
```





Provider Entity Settings	Entity Addresses/Phones
* Indicates a required field. * ID: 100027399A	Add Address
PIN: 15060	Address Type: Work Delete
First Name:	* Address Line 1: PO BOX 1417
Middle Name:	Address Line 2: 1100 E WYATT EARP BLVD
Last Name:	* City: DODGE CITY
* Company Name: ARROWHEAD WEST INC	* State: KS * Zip: 678011417
SSN:	
FID:	
Gende <del>r:</del>	Add Phone
Birth Date	Phone Type Phone Number
* Communication Email:	Work (620)227-8803 Delete
Begin Date: 7/1/1991	
End Dates	ID Card Number
Language:	Status:
Status: Active	Registered Users
* Entity Qualifier: Business	Add User
NPI:	User Name Role Registered On Enabled Delete
Personal Assistant Services - MRDD Sleep Cycle Support - MRDD Overnight Respire - MRDD	provider@arrowhead.com KS_Administrator10/31/2011

3. Click Add User.

Registered Users Add User				
User Name	Role	Registered On	Enabled	Delete
provider@arrowhead.com	KS_Administrato	r10/31/2011		

### The Register page displays.

Register Use your assigned ID and P	PIN plus your e-mail address an	d a password you ch	oose to register f	or access to the website
* Email Address				
* Password:				
* Confirm Password:				
* User Roles KS KS KS KS	S_AdminAssistant S_Administrator S_ClaimsMgt1 S_ClaimsMgt2 S_HumanResources S_IntakeReferral S_Payrol/Billing		Rights Add Claims Edit Claims View Claims Delete Claims View Clients Add Workers Edit Workers View Workers	





- 4. Choose a **Sub-Role** from the list. When you choose the sub-role, the screen will refresh to display the list of rights for that sub-role on the right. See Appendix A.4 for a description of the functions each sub-role can performed. Available provider sub-roles:
  - KS\_Administrator
  - KS\_AdminAssistant
  - KS\_Payroll/Billing
  - KS\_Human Resources
  - KS\_Payroll/Billing/Human Resources
  - KS\_Scheduler/Coordinator
  - KS\_Intake & Referral

<u>There can be only one sub-role per email address.</u> If the person needs to be assigned to more than one sub-role, he/she will need to be added again with another email address.

5. Enter an **Email Address**.



This email address serves as a User Name and is not used to contact the provider. It does not have to be a valid email address, but it must be in the **format** of an email address.

For example, a provider with a facility called. "Special Care, Inc." may wish to type in the User Name: specialcare@inc.com.

- 6. Enter a **Password**. The password must have at least 1 uppercase letter, at least 1 lower care letter, at least 1 number and must be at least 6 characters in length.
- 7. Re-enter the password in **Confirm Password**.



This password is used in the future along with the email address to access the website so it is important to write them down and store them in a secure place.

The *Register* page displays with the entered information.

November 14, 2011





Register Use your assigned ID an	d PIN plus your e-mail address and a passwo	ord you choose to register for access	to the website.
* Email Address	Betsy@redslippers.com		
* Password:	••••		
* Confirm Password:	••••		
		Rights	
* User Roles	KS_ClaimsMgt2 KS_HumanResources KS_IntakeReferral KS_PayrollBilling KS_PayrollBilling/HumanResources KS_SchedulerCoordinator State Administrator	Add Claims Edit Claims View Claims Delete Claims View Clients Edit Workers View Workers	
		View Providers Register	Cancel

8. Click Register.

You are returned to the *Provider Entity Settings* page. A message in the top left hand corner informs you that the registration was completed successfully. The user you just registered displays in the Registered Users section of the *Provider Entity Settings* page specifying the email address, the sub-role assigned and that he is enabled.

Provider Entity Settings	Entity Addresses/Phones
<sup>®</sup> Indicates a required field.	Add Address
* ID: 100027399A	Add Address
PIN: 15060	Address Type: Work Delete
First Name:	* Address Line 1: PO BOX 1417
Middle Name:	Address Line 2: 1100 E WYATT EARP BLVD
Last Name	* City: DODGE CITY
* Company Name: ARROWHEAD WEST INC	* State: KS * Zip: 678011417
SSN:	
FID:	
Gender:	Add Phone
Birth Date:	
* Communication Email:	Phone Type Phone Number
Begin Date: 7/1/1991	Work (620)227-8803 Delete
End Dates	ID Card Number:
	Status:
Language:	
Status: Active	Registered Users
* Entity Qualifier: Business	Add User
NPI:	User Name Role Registered On Enabled Dele
	■ Betsy@redslippers.com KS Payroll/Billing 11/09/2011
LPN Specialized Medical Care - MR * Provider Services: RN Specialized Medical Care - MRD	DD provider@arrowhead.com K5_Administrator 10/31/2011 🖓 🗌
Provider Services: KN Specialized Medical Care - MRL Personal Services - PD	Ruby@redslippers.com K5_HumanResources11/09/2011





#### 2.2 LOGGING IN TO KS AUTHENTICARE

1. Navigate to the KS AuthentiCare website at https://acr.fdgs.com/Kansas.

The Welcome page displays.

Kansas	Kansas AuthentiCare	Ð
Custom Links	Welcome to AuthentiCare Plesse enter your AuthentiCare email address and password to access the system Register for Access * Email Address Password: Submit	
	Copyright © 2011 by First Data Government Solutions Privacy Policy	

- 2. Enter your KS AuthentiCare **Email Address** and **Password**.
- 3. Click Submit.

Welcome to AuthentiCare Please enter your AuthentiCare email address and password to access the system.			
Register for Access			
* Email Address:	sooner@healthcare.com		
Password:	••••••		
		Submit	

The *Home* page displays and varies in appearance based on the user's sub-role. Your view of the Home page may be different from what is displayed in this User Manual as this is the Provider Administrator's view. From the Home page, you can access the functionality available in KS AuthentiCare.





Kansas Home   Create   Reports   Schedulin	Kansa	s AuthentiCare	Logged in ss: scr_sdmin@scr.com
	Entities	Claims	
	Add Naw > <u>Client</u> <u>Provider</u> Worker CaseManager	Add New > <u>Claim (Standard)</u> Add New > <u>Claim (Express)</u>	
	Representative Search >	⊙ Claim Search Type: ○ Confirm Billing - View ○ Confirm Billing - Bulk	
	-	Claim Group ID	
	Services and Authorizations	Claim ID Claim Status	
	Add New > <u>Service</u>	Claim Start 📉 💌 Claim End 📉	
	Search Type: O Authorization	Service	
	Service	Authorization ID	
	Authorization ID	Client:	
	Service Type	Worker:	
	Authorization End:	CaseManager:	
	Client	Representative:	
	Provider:	Procedure Code:	
	Worken	User Option: 💌	
	Procedure Code	Go! Clear	

#### 2.3 LOGGING OUT OF KS AUTHENTICARE

For security and privacy reasons, it is important for users to log off KS AuthentiCare when leaving the work station. After a 20 minute period of inactivity, KS AuthentiCare automatically logs the user off.

Click on Logout from the Main Menu



The Welcome page displays and you are logged out of KS AuthentiCare.

Kansas	Kansas AuthentiCare	٢
	Welcome to AuthentiCare Please enter your AuthentiCare email address and password to access the system. Realister for Access * Email Address: Password: Submit	
November 14, 2011	KS Authe	ntiCare User Manual





#### 2.4 CHANGING PASSWORD

There are several reasons why a KS AuthentiCare user would want or need to change their password:

- *Password Expiration:* The password expires every 30 days and a new one needs to be chosen. There is a password rotation of 12 passwords which means that you cannot use any of your previous 12 passwords when creating a new password.
- *Security:* A user may want to secure their password by creating a new one if they feel this information has been compromised.
- *Account Lockout:* After 10 failed attempts to log-in in a single session the account is automatically disabled. To re-establish access:
  - 1. Providers someone with an Administrator sub-role can manage registrations and change the password for the locked out user.
  - 2. State Administrators contact First Data Client Support services at 1-800-441-4667 or <u>clientsupport@firstdata.com</u>.



When creating a new user account, you cannot use the same email address as the account that has been locked out.

Follow the steps below to change your password.

1. Place your cursor on **My Account** in the Main Menu, and when **Change Password** displays below, click on it.



The Change Password page displays.

Change Password Please enter your existing password, then enter your new	password and confirm to change your current password.
* Existing Password:	
* New Password:	
* Confirm New Password:	
	Submit

2. Enter your Existing Password.





3. Enter a New Password.

X	•	The Password must have at least 1 uppercase letter, at least 1 lower case letter, at least 1 number and at least 6 characters in length.	
100	•	This password is used in the future along with the email address to access the website so it is important to write them down and store them in a secure place.	
	•	The password entered cannot be the same as any of the previous 12 passwords you have used.	

- 4. Re-enter the password in **Confirm Password**.
- 5. Click **Submit**.

Hange Password Hease enter your existing password, then enter your new	password and confirm to	o change your current password.
* Existing Password:	•••••	
* New Password:	•••••	]
* Confirm New Password:	•••••	
		Submit

The KS AuthentiCare *Home* page displays with a confirmation that you successfully changed your password.



#### 2.5 DISABLING A USER

#### 2.5.1 PROVIDERS

If an employee of the provider has left the company and that person had access to KS AuthentiCare, it is critical that the user account is disabled immediately. Because you can log in to KS AuthentiCare anywhere you can access the Internet, a former employee would still have access to your agency's information. <u>Only someone with the Administrator sub-role can disable an account.</u>

November 14, 2011





1. Navigate to the *Provider Entity Settings* page by entering the name of the provider or some portion of the name in the Entities **Search** > field and click **Go!** 

Entities	
Add New >	<u>Client</u> <u>Provider</u> <u>Worker</u> <u>CaseManager</u> <u>Representative</u>
Search >	Ja yhawk

The *Provider Entity Settings* page displays with the information about this provider.

Kansas	Kansas Aut	
Home   Create   Reports	B   Scheduling   Dashboards   Visits   Administration   My Account	:   Custom Links   Logout Logged in a
	Provider Entity Settings	Entity Addresses/Phones
	* Indicates a required field. * ID: 100260589A	Add Address
	PIN: 45706	Address Type: Work Delete
	First Name:	* Address Line 1: 2910 SW TOPEKA BLVI
	Middle Name:	Address Line 2:
	Last Name	* City: TOPEKA
	* Company Name: JAYHAWK AREA AGENCY ON AGING	3 * State: KS * Zip: 66611
	SSN:	
	FID:	
	Gender	Add Phone
	Birth Dates	Phone Type Phone Number
	* Communication Email:	Work (785)235-1367 Delete
	Begin Date: 1/1/1997	
	End Date:	ID Card Number:
	Language:	Status:
	Status: Active	Registered Users

2. Click the **Enabled** checkbox and the check mark will disappear.





Registered Users Add User				
User Name	Role	Registered On	Enabled	Delete
Betsy@redslippers.com	KS_Payroll/Billing	11/09/2011	<b></b>	
provider@arrowhead.com	KS_Administrator	10/31/2011		
Ruby@redslippers.com	KS_HumanResource	s11/09/2011		

3. Click Save.

You are returned to the *Home* page which displays a message that the provider was saved successfully.



This user will no longer be able to log into KS AuthentiCare.



Do not click the "delete" check box to disable the user. That will hamper the functions in KS AuthentiCare that automatically track by user name (email) actions that were taken in the system.

#### 2.5.2 STATE ADMINISTRATIVE USERS

If a user should no longer have access to KS AuthentiCare, contact First Data Client Support Services at <u>clientsupport@firstdata.com</u> to have the account disabled.





## Chapter 3 AN INTRODUCTION TO NAVIGATING KS AUTHENTICARE

KS AuthentiCare provides several "starting points" from which to navigate through the system. All users will see the *Home* page when first logging in to the system. From this point, the user can access all data and functionality allowed by the assigned role or sub-role.

For providers with the sub-role of Administrator, AdminAssistant, Payroll/Billing or Payroll/Billing/Human Resources, there are two additional navigation tools that are listed on the top tool bar, *Provider Dashboard* and *Worker Dashboard*. These are short cuts to data that is essential to daily operations.

#### 3.1 HOME PAGE

The *Home* page is the central location in KS AuthentiCare. From this page you can navigate to any of the other areas of the system. This page differs depending on your role, and therefore, controls the information you can access in the system.

Kansas		AuthentiCare	Ø
Home   Create   Reports   Scheduling   Das	hboards   Visits   Administration   My Account   Cust	Claims	Logged in as: acr_admin©acr.com
	Add New > <u>Client</u> <u>Provider</u> Worker <u>CaseManager</u> Representative	Add New > <u>Claim (Standard)</u> Add New > <u>Claim (Express)</u>	
	Search >	⊙ Claim Search Type: ○ Confirm Billing - View ○ Confirm Billing - Bulk	
	Services and Authorizations	Claim Group ID Claim ID Claim Status	
	Add New > <u>Service</u> Search Type: ⊙Service ⊖ Authorization	Claim Start V Claim End V Service	
	Service Authorization ID Service Type	Authorization ID Client Providen	
	Authorization Start	Worker:	
	Client Providen Worker:	Representative: Procedure Code: User Option: Go! Clear	
	Procedure Code	GO! Clear	

The user name is displayed in the upper right-hand corner of the *Home* page. All actions taken by the user are tracked in KS AuthentiCare to maintain an accurate record of activity.





Logged in as: Suzie@genHC.cor

# Kansas AuthentiCare



Administration | My Account | Custom Links | Logout

The Main Menu on the tool bar across the top of the page allows the user to navigate to different pages to perform different tasks. The Main Menu functions are discussed in detail throughout this manual. The options on the main menu differ depending on the user's role and appear on every page in the system.



All users have a **Custom Links** option on the tool bar that links to sites or information that is outside of KS AuthentiCare. These links include the KDOA and KDOA provider websites, the SRS website, the KMAP website and the Kansas Self-Directed Services website.

Kansas	Kansas AuthentiCare
Custom Links	
KDOA public website KDOA provider website KMAP Provider website SRS Self-Directed	Welcome to AuthentiCare Please enter your AuthentiCare email address and password to access the system. Register for Access
	* Email Address Password: Submit





The *Home* page is divided into three sections:

Entities	Claims
Add New > <u>Client</u> <u>Provider</u> <u>Worker</u> <u>CaseManager</u> Representative	Add New > <u>Claim (Standard)</u> Add New > <u>Claim (Express)</u>
Representative	⊙ Claim
Search >	Search Type: O Confirm Billing - View
Go!	O Confirm Billing - Bulk
	Claim Group ID
Services and Authorizations	Claim ID:
	Claim Status:
Add New > Service	Claim Start 💌
• Service	Claim End 🛛 👻
Search Type: O Authorization	Service
Service	Authorization ID
Authorization ID	Client
	Provider:
Service Type	Worker:
	CaseManager:
Authorization End:	
Client:	Representative:
Provider:	Procedure Code:
Worker:	User Option:
Procedure Code	Go! Clear

- 1. **Entities** Entities is a broad category within KS AuthentiCare that designates people and agencies involved in the care giving experience including:
  - o Clients (which is the consumer or customer in Kansas)
  - Providers (each provider agency and FMS provider)
  - Workers (both those employed by a provider agency and those working for an FMS provider)
- 2. Services and Authorizations This section of the Home page allows the user to perform detailed searches for existing services and authorizations by clicking on Go!

November 14, 2011





 Claims – While most claims are created using the IVR phone system or the GPS enabled mobile device, designated provider sub-roles can also create claims using the website. All claims, whether they were created using the IVR, mobile device or the website, can be viewed from the website. Providers may perform detailed searches for existing claims from the Home page. Refer to Chapter 11, Managing Claims.

The Search and Create functions are described in more detail throughout the manual.

#### 3.2 PROVIDER DASHBOARD

In addition to the *Home* page, KS AuthentiCare includes a Provider Dashboard where information is presented in a graphical format and provides easy navigation to the source data represented in the dashboard. The Provider Dashboard is only available to users who have the Administrator, AdminAssistant, Payroll/Billing or Payroll/Billing/Human Resources sub-roles in the system.

Place your cursor on **Dashboards** in the Main Menu and click on **Provider**.



The Provider Dashboard displays with real time information.

ABC Home Healt Provider No.: 12		April 16, 2009 11:22 am
Unsubmitted Clai	im Status 36 List 73 List 19 List	*Claims Unsubmitted Claims 100 100 100 100 100 100 100 10
Today's Schedul	e Status	Today's Schedule
Checked In On Time Scheduled Not Due Late or Missed	51 List 82 List 14 List	Checked In On Time Late or Missed 3 4 5 1 82 Scheduled Not Due





The *Unsubmitted Claim Status* section provides information on the number of claims that fall into the following categories:

- <u>Ready to export</u> This gives the number of claims that have been confirmed by the provider, but have not yet been submitted to KMAP/MMIS for adjudication. These claims may have informational exceptions, but they do not have any critical exceptions. In the next routine system export, these claims will be sent to KMAP/MMIS. Click on the **List** button to view these claims on the *Claim Search Results*. Refer to Section 12.1 for more information.
- <u>Needs Confirmation</u> This represents the number of claims that are ready to be confirmed for billing to allow submission to KMAP/MMIS. In other words the only critical exception on these claims is billing confirmation. Once a claim has been confirmed for billing, it will be moved to the "ready to export" group. Click on the List button to view the *Confirm Billing* page where you can confirm these claims. Refer to Section 12.6 for more information.
- <u>Critical exception</u> This represents the number of claims with one or more critical exceptions in addition to billing confirmation. A critical exception means the claim did not pass one of the system edits that are based on KDOA, SRS and KMAP/MMIS business rules. These exceptions prevent the claim from being confirmed for billing and exported to KMAP/MMIS. Click on the List button to view the *Claim Search Results* which will list these claims and provide a link to the claim so that details about the exception can be viewed and addressed.

The *Today's Schedule Status* section provides information for the events scheduled for the current day only if the provider is using the scheduling functionality of KS AuthentiCare. Refer to Chapter 9, Scheduling for further information. The information in this section of the dashboard is divided into the following three categories:

- <u>Checked In On Time</u> This represents the number of events (visits) where the worker has checked in on time. (On time is defined as checking in between 60 minutes prior to the scheduled event start time and 60 minutes after the scheduled event start time.) For further information about these events, click **List** to view the *Worker Dashboard* which will display all events identifying the worker, the check in time, and the check out time (if the visit has been completed). Refer to Section 3.3 for further information.
- <u>Scheduled Not Due</u> This represents the number of events scheduled for later in the day. To view these scheduled events, click **List** to view the *Worker Dashboard* which will display the scheduled events and the worker assigned to each event scheduled to occur later in the day. Refer to Section 3.3 for further information.
- <u>Late or Missed</u> This represents the number of scheduled events where the worker either checked in Late (between 60 and 120 minutes after the scheduled start time of the event), has not yet checked in and the current time is between 60 and 120 minutes after the

November 14, 2011





scheduled start time of the event, or the worker did not check in within 120 minutes of the scheduled start time of the event.

• If a worker checks in more than 60 minutes before the scheduled start time of the event or 120 minutes or later after the scheduled start time of the event, the check in will not match to the scheduled event and the event status will be "Missed". For further information about these events, click **List** to view the *Worker Dashboard* which will display all late or missed events for the current day. Refer to Section 3.3 for further information.

In your provider agency or FMS provider does not use the scheduling functionality in KS AuthentiCare this part of the Dashboard will be blank.

#### 3.3 WORKER DASHBOARD

In addition to the Provider Dashboard, KS AuthentiCare also includes a Worker Dashboard to facilitate quick access to information about the current day's schedule. The Worker Dashboard is accessible to users with the Provider Administrator, AdminAssistant, Payroll/Billing, Payroll/Billing/Human Resources, and Scheduler/Coordinator sub-roles.

In your provider agency or FMS provider does not use the scheduling functionality in KS AuthentiCare the Worker Dashboard will be blank.

Place your cursor on **Dashboards** in the Main Menu, and when **Worker** displays below, click on it.



The Worker Dashboard displays with real time information about the scheduled events for the current day.





ABC Home Healt	h				А	pril 16, 200	9
rovider No.: 12	34567					11:22 an	n
Today's		Scheduled	Actual				
Scheduled Morkers	Worker ▲ ID ▼	Check-In ▲Time ▼	Check-In ▲ Time ▼	Client ▲ Name ▼	▲ Client ID ▼	▲ Status ▼	,
<u>Boop, Betty</u>	<u>80244</u>	<u>08:30 am</u>	<u>08:35 am</u>	<u>Fudd, Elmer</u>	987654	Checked In On Time	
<u>Bunny, Bugs</u>	<u>19827</u>	<u>02:00 pm</u>		<u>Walker, Jay</u>	<u>678345</u>	Scheduled Not Due	
Cotton, Candy	<u>50822</u>	<u>07:00 am</u>	<u>06:50 am</u>	<u>Nasium, Jim</u>	256356	Checked In On Time	
)evil, Tasmanian	<u>44987</u>	<u>08:00 am</u>		<u>Padd, Lily</u>	576879	Late or Missed	
n 1 n 11		10.00		· ·	<b>/</b>	Late or	1
		Copyriç	ht © 2010 by F	irst Data			

Each column has a sort option. By default the Worker Dashboard displays the scheduled events in ascending alphabetical order of the worker's last name.

To change the order of the sort, click on the arrow in the column name on which you wish to have the information displayed. For instance if you click on the up arrow in the Worker ID column, the information will be sorted in ascending worker ID order.

ABC Home Healt Provider No.: 12						April 16, 2009 11:22 am
Today's Scheduled ▲ Workers ▼	Werker ▲ JD ▼	Scheduled Check-In ▲Time ▼	Actual Check-In ▲ Time ▼	Client ▲ Name ▼	▲ Client ID	▼ 🔺 Status ▼
<u>Boop, Betty</u>	80244	<u>08:30 am</u>	<u>08:35 am</u>	<u>Fudd, Elmer</u>	<u>987654</u>	Checked In On Time
<u>Bunny, Bugs</u>	<u>19827</u>	<u>02:00 pm</u>		<u>Walker, Jay</u>	<u>678345</u>	Scheduled Not Due
<u>Cotton, Candy</u>	<u>50822</u>	<u>07:00 am</u>	<u>06:50 am</u>	<u>Nasium, Jim</u>	<u>256356</u>	Checked In On Time

The Status column displays one of the following three statuses:

- <u>Checked In on Time</u> This means the worker checked in between 60 minutes prior to the scheduled event start time and 60 minutes after the scheduled event start time.
- <u>Scheduled Not Due</u> This means the event is scheduled to occur later in the day based on the current time in AuthentiCare.
- <u>Late or Missed</u> This means either the worker checked in late (between 60 and 120 minutes after the scheduled start time of the event), has not yet checked in and the current time is between 60 and 120 minutes after the scheduled start time of the event, or the worker did not check in within 60 minutes prior to the scheduled start time of the event.

November 14, 2011





Each of the data items displayed on the Worker Dashboard is a hyperlink.

- Clicking on the worker name or worker ID will take you to the *Worker Entity Settings* page. This allows you to view the information about the worker. Refer to Section 4.4 for further information.
- Clicking on the scheduled check in time will take you to the *Scheduled Event* page. This allows you to view the information about the event. Refer to Section 9.2 for further information.
- Clicking on the actual check in time (if there is one) will take you to the *Claim* page. This allows you to view the information captured for the claim. Refer to Section 12.1 for further information.
- Clicking on the client name or client ID will take you to the *Client Entity Settings* page. This allows you to view the information about the client for whom the event is scheduled. Refer to Section 4.3 for further information.





### Chapter 4 SEARCH AND VIEWING PEOPLE AND/OR AGENCIES

The extent to which a user is able to view people and/or agencies depends on his/her role in the system.

- The KDOA Administrative users can view all data in KS AuthentiCare related to clients in the Frail Elderly Waiver.
- The SRS Administrative users can view all data in KS AuthentiCare related to clients in the following SRS Waivers: Physical Disability, Technology Assisted, MR/DD and Traumatic Brain Injury.
- A provider user can view the information about his/her agency as well as information about all clients served by the agency and all workers who work for that agency.

#### 4.1 SEARCHING FOR PEOPLE AND/OR AGENCIES

1. Enter a name, partial name or ID in the Search field and click Go!

Entities	
Add New >	Worker
Search >	
	Go!



If nothing is entered in the **Search** field, then all entities will be returned in your search results. This could take awhile and it will also make it difficult to find the entity.





The search results display all Entities which match the search criteria entered on the *Home* page. If there is only one entity that matches the search criteria entered, then that entity is displayed immediately rather than the *Entity Search Results* page.

Entity Searc	h Results		
ID	Name	<u>User Type</u>	
<u>414</u>	Enid Worker's Co.	Provider	£
<u>56331</u>	Delmon Young	Worker	£
<u>097097097</u>	Joe Mauer	Client	â
<u>777777777</u>	Joe Medwick	Client	£
<u>418</u>	Sooner's Provider	Provider	£
<u>94964</u>	Walter Alston	Worker	â
<u>421</u>	Johnny Keane	CaseManager	£
<u>010925904</u>	Enos JaneTest Slaughter	Client	3
<u>80172</u>	Shawn Chacon	Worker	8
<u>434</u>	Teller's Maid Sevice	Provider	3
< Prev 1 <u>2</u> <u>3</u>	<u>4 5 Next &gt;</u>		

- 2. Note the columns displayed in the search results:
  - **ID** Click this link to go to the *Entity Settings* page for this entity
  - **Name** Identifies the name for the Entity
  - User Type Identifies the type of Entity: Provider, Client or Worker
- 3. Click the column heading for the search results to sort using a different column. Click the heading once to change the sort to ascending order for that column. Click the heading again to change it to descending order for that column.
- 4. Position the cursor over the **Information icon** to display a pop-up to view more detailed information.

stom Links   Lo	ogout	Additional Information			
Entity Searc	h Results	ID: Full Name: Home Address:	987654321 Wendy Brown		
ID	Name	<u>User Type</u>		Home Phone: Work Address: Work Phone: Date of Birth:	(712) 908-0998
<u>421</u>	Ryan Howard	CaseManager	ŝ		04/05/1920
<u>427</u>	Sooner Healthcare	Provider	£	Effective Date St.	art:
023456896	McLain Ward	Client	6	Effective Date End: Medicaid ID:	ıd: 987654321
<u>12156</u>	Betsy Jones	Worker	<u>1</u>	Email: SSN:	
23424	Wendy Brown	Worker	3	FID: Status:	Active
<u>987654321</u>	Wendy Brown	Client	8		





5. Click the **ID** number in the first column next to the entities name to view the *Entity Settings* page. The *Entity Settings* page that displays is dependent on the type of entity.

l	Entity Search Results								
	ID	Name	<u>User Type</u>						
	<u>414</u>	Enid Worker's Co.	Provider	â					
	<u>56331</u>	Delmon Young	Worker	ŝ					
	<u>097097097</u>	Joe Mauer	Client	<u>1</u>					
<	11111111	Joe Medwick	Client	<u>1</u>					
	418	Sooner's Provider	Provider	<u>1</u>					
	<u>94964</u>	Walter Alston	Worker	<u>1</u>					
	<u>421</u>	Johnny Keane	CaseManager	<u>1</u>					
	<u>010925904</u>	Enos JaneTest Slaughter	Client	<u>8</u>					
	<u>80172</u>	Shawn Chacon	Worker	<u>a</u>					
	<u>434</u>	Teller's Maid Sevice	Provider	<u></u>					
	< Prev 1 <u>2 3 4 5 Next &gt;</u>								

#### 4.2 VIEWING PROVIDERS

If you click on a provider, the *Provider Entity Settings* page displays.

<b>Provider Entity Setting</b>	S		Entity Addres	ses/Phones
* Indicates a required field.			Add Addres	5
* ID:	100260589A			
PIN	45706	Address Type: Work Delete		
First Name			* Address Line 1	: 2910 SW TOPEKA BLV
Middle Name			Address Line 2	-
Last Name				r TOPEKA
* Company Name SSN:	JAYHAWK AREA AGENC	Y ON AGING	* State	:KS *Zip:66611
FID:				
Gender			Add Phone	
Birth Date			Phone Type Pho	ne Number
* Communication Email				5)235-1367 Delete
Begin Date:	1/1/1997		WORK (76	5/255-1567 Delete
End Date:			ID Card Number	
Language:			Status:	
Status:	Active		Registered Us	ers
* Entity Qualifier	Business		Add User	
NPI			Holidays / Day	rs Off
* Provider Services	Level 2 Attendant Care - F Self Directed Attendant C Level 1 Attendant Care - F Adult Day Care - FE Personal Emergency Ras Wellness Monitoring - FE Nurse Evaluation Visit - F Sleep Cycle Support - FE Provider Directed Compre- Self Directed Comprehen	are -FE FE ponse - Install - FE E shensive Support - FE		
Taxonomy Code	ı		~	Add Remove
* Mobile Enabled:			From Date	To Date
Messaging Enabled:	⊖ Yes ® No			
	Number Device Id	Accimment	Work Hours	

November 14, 2011




This page provides information on the provider, the services enrolled to provide, the provider's workers, address and phone information, holidays, and work hours. This information is data scoped (restricted) to match the rights of the user who is logged in. A provider can only see information specific to that provider agency. Only the provider can edit this information via the web. State administrative staff can see (but not change) provider information.

## 4.3 VIEWING CLIENTS



If you click on a client, the *Client Entity Settings* page displays.

This page provides client information required for claims processing, scheduling and using the IVR/mobile device. The client's name and Medicaid ID must be an exact match to the information in KMAP/MMIS in order for a claim to process and be paid without exception. Only the provider can add or edit client information on the web.

## 4.4 VIEWING WORKERS

If you click on a worker, the *Worker Entity Settings* page displays.





Worker Entity Settings		Entity Addresses/Phones
* Indicates a required field.		
	23322	Add Address
PIN:		Add Phone
* First Name	Jo	ID Card Number:
Middle Name:		Status:
* Last Name	Worker	Holidays / Days Off
Company Name		
SSN:		
FID:		
Gender		
* Birth Dates	8/3/1968	
Email Address		
Begin Date:	×	Add Remove
End Date:	×	From Date To Date
Language:	✓	
Status:	Active Y	Work Hours
		StartTime End Time Def Off Cust
External Worker ID:		Sun 📃 🔍 O O O
Family Member		Mon 📃 💿 🔿 🔿
Live with a client you serve:	No M	Tue 0 0 0
	Level 2 Attendant Care - FE Self Directed Attendant Care - FE	Wed
* Worker Services:		Thu 0 0 0
	Personal Emergency Response - Install - FE	Fri 000
* Mobile Enabled:	O Ves ⊙ No	Sat 0 0 0

This page provides information about the worker that is required for scheduling and capturing service activities via the IVR. This information is entered and maintained by the provider.





# Chapter 5 MANAGING PROVIDER INFORMATION

### 5.1 EDITING PROVIDER INFORMATION

Providers are created and updated in KS AuthentiCare by a file transfer from KMAP/MMIS. Each provider can only access information associated with their Medicaid ID. State Administrative staff can also view a provider's information but cannot change any of a provider's information on the web.

When first signing on to the system, the Provider Administrator user must view their information in KS AuthentiCare to verify that it is correct. If it is not correct, contact KMAP/MMIS so that corrections can be made and then transmitted to KS AuthentiCare.

Providers must also add an **Email Address** to their provider page. This is the email address used for notification of late and/or missed visits.

You may also enter holidays and working hours for your agency if you wish. If entered, this will drive some of the pop-up messages in the Scheduling system.

Providers should verify their information in the system on a regular basis to make sure that it is up-to-date and no changes are required.

1. Follow the Search and View instruction above to access your *Provider Entity Settings* page.



2. Verify the existing data is correct.





3. Enter the **Email Address**. This is where KS AuthentiCare will automatically send alerts for missed and late visits.



Unlike the email address you use for your user name, KS AuthentiCare uses this address to automatically send an alert if a worker is late checking in for a scheduled service and/or if a worker does not check in at all to provide a scheduled service (missed visit).

There can only be one email address but it can be an individual or an address that distributes the email to multiple employees within your agency.

- 4. Schedule information.
  - A) Providers <u>may choose</u> to add Holidays/Days Off for the agency. These are days that the agency does not provide services to clients, not days that the office is closed.
    - To add a single day: Type in the **Date** OR choose a date from the calendar then click **Add**.
    - To add a date range for a multi-day agency closure: Enter a **From Date** and **To Date** OR choose the dates from the calendars, and then click **Add**.

	Holidays / Days Off										
Friday, January 01, 2010											
	1/1/	2010		~	Add	ŀ	Rem	ove			
	4	Janu	Jary	~	20:	10 🔽	] 🕨				
	Sun	Mon	Tue	Wed	Thu	Fri	Sat	Add			
						1	2				
1	3	4	5	6	7	8	9				
	10	11		13			16				
				20				Off Cust			
	24 31	25	26 2	27 3	28	29 5	30 6	0.0			
	-			30/20				0.0			

B) Providers may choose to enter **Work Hours**. These are the hours that workers from your agency may provide services for clients and does not reflect the office hours of the agency.

November 14, 2011





• To apply the **default hours (12:00 AM to 11:59 PM)**, no change is needed.

Wor	Work Hours								
	Start Time	End Time	Def	Off	Cust				
Sun	12:00 AM	11:59 PM	œ	0	0				
Mon	12:00 AM	11:59 PM	o	0	c				
Tue	12:00 AM	11:59 PM	·	0	c				
Wed	12:00 AM	11:59 PM	·	0	o				
Thu	12:00 AM	11:59 PM	·	0	C				
Fri	12:00 AM	11:59 PM	۰	o	С				
Sat	12:00 AM	11:59 PM	۲	0	C				

- o To enter specific service hours for one or more days of the week:
  - Click the radio button in the column labeled "Cust" (for Custom)
  - Enter the **Start Time** and **End Time** in the fields provided. Be sure to indicate AM or PM for each entry.

Work Hours								
Start Time	End Time	Def	Off	Cust				
Sun 12:00 AM	11:59 PM	۲	0	0				
Mon 8:00 AM	8:00 PM	0	0	c				
Tue 12:00 AM	11:59 PM	۰	0	0				
Wed 12:00 AM	11:59 PM	œ	0	0				
Thu 12:00 AM	11:59 PM	œ	0	0				
Fri 12:00 AM	11:59 PM	۲	0	0				
Sat 12:00 AM	11:59 PM	۲	0	0				

- To **establish days of the week (for every week)** that the agency's workers do not provide services (for example on Sunday):
  - Check the circle in the column labeled "Off".

Start Time         End Time         Defl/Off Cust           Sun         C         C         C           Mon         8:00 AM         8:00 PM         C         C           Tue         12:00 AM         11:59 PM         C         C         C           Wed         12:00 AM         11:59 PM         C         C         C           Thu         12:00 AM         11:59 PM         C         C         C           Fri         12:00 AM         11:59 PM         C         C         C           Sat         12:00 AM         11:59 PM         C         C         C	Work Hours									
Mon         8:00 AM         8:00 PM         C         C         C           Tue         12:00 AM         11:59 PM         C         C         C           Wed         12:00 AM         11:59 PM         C         C         C           Thu         12:00 AM         11:59 PM         C         C         C           Fri         12:00 AM         11:59 PM         C         C         C	Start Time	End Time	Def	Off	Cust					
Tue         12:00 AM         11:59 PM         C         C         C           Wed         12:00 AM         11:59 PM         C         C         C           Thu         12:00 AM         11:59 PM         C         C         C           Fri         12:00 AM         11:59 PM         C         C         C	Sun		0	o	0					
Wed         12:00 AM         11:59 PM         ©         C         O           Thu         12:00 AM         11:59 PM         ©         C         O           Fri         12:00 AM         11:59 PM         ©         C         O	Mon 8:00 AM	8:00 PM	0	0	c					
Thu         12:00 AM         11:59 PM         C         C           Fri         12:00 AM         11:59 PM         C         C	Tue 12:00 AM	11:59 PM	·	0	0					
Fri 12:00 AM 11:59 PM C C O	Wed 12:00 AM	11:59 PM	۲	0	0					
·	Thu 12:00 AM	11:59 PM	۲	0	0					
Sat 12:00 AM 11:59 PM • C O	Fri 12:00 AM	11:59 PM	œ	0	0					
	Sat 12:00 AM	11:59 PM	۰	0	0					







Entering details in this section causes warning notices to display if the provider attempts to schedule a worker for a time that falls outside of the agency's work hours. However, you are not prevented from scheduling a visit outside normal work hours. Refer to Chapter 9, Scheduling.

5. Click Save.



A successful save message displays at the top of the page indicating the provider information was saved successfully.



## 5.2 835 ELECTRONIC REMITTANCE ADVICE FROM KMAP/MMIS

KS AuthentiCare submits claims on the provider's behalf to KMAP/MMIS in a HIPAA compliant 837 electronic file in the early morning hours of each week day. KMAP/MMIS provides adjudication results to the provider in a HIPAA compliant 835 electronic remittance advice file.

Each provider must decide whether or not to load the 835 remittance advice into KS AuthentiCare. The advantage to loading it into KS AuthentiCare is that you will have a complete history of each claim in one system; from the worker's IVR call/mobile device contact or claim web entry through adjudication with the amount paid linked to each service episode.

If you choose to not load the 835 remittance advice into KS AuthentiCare, many of your reports will be incomplete; having no information about a claim beyond the date it was submitted to KMAP/MMIS for adjudication.

If you choose to load the 835 loaded into KS AuthentiCare, use the process outlined below.





### 5.3 UPLOADING THE 835 INTO AUTHENTICARE

Providers can upload the 835 files they receive from KMAP/MMIS into KS AuthentiCare whenever they want to do so, 24/7. The ability to upload the file is a right assigned to the person(s) with an Administrator role in the system. KS AuthentiCare will accept the 835 file with an ".edi" extension, as a zip file having a ".zip" extension or with an ".rsp" extension.

Once uploaded, the 835 file will be queued in the background for the remittance data to be imported. The provider will be notified by email when the processing is complete. The provider can enter up to three email addresses to be used to report file processing results.

- If the file is successfully processed, providers will be able to run the remittance reports to see the data processed from the 835 files.
- If the file is unsuccessful, the reason will be provided in the email along with contact information for Customer Support for assistance in resolving the issue.

To begin the upload process, hover over the **Administration** tab on the tool bar and click on the **835 Import** tab when it appears,



The screen below displays where you can upload your 835 file.

Kansa	as Kansas AuthentiCare	
Home   Create	Reports   Scheduling   Dashboards   Visits   Administration   My Account   Custom Links   Logout Logged in as: Suzie	s@genHC.com
	Do not close your browser or move away from this page until you get a confirmation message that the file has been successfully uploaded.	
	* Indicates a required field.	
	* Select File Attach It	





A 90 day rolling history will be displayed that shows details about the files uploaded (date, time, uploaded and processed, success/failure, payment date in the file, Medicaid ID in the file).

835 Import History											
<u>File Name</u>	<u>Date &amp;</u> <u>Time</u> <u>Uploaded</u>	Uploaded By	<u>Size</u>	<u>Time</u> Processed	<u>Status</u>	Medicaid ID	<u>Payment</u> <u>Date</u>				





# Chapter 6 MANAGING WORKER INFORMATION

A worker is all the provider's staff members who will be providing services for a client (does not include staff who use the web only). Prior to implementation, worker information for each provider is pre-populated in KS AuthentiCare. Prior to implementation, providers must verify that the worker information is correct and add additional information for workers if appropriate such as serving a family member or living with the client being served.

Workers hired by the provider after the files were sent to First Data for pre-populating must be added via the web.

Worker information requires regular maintenance to add new workers, update current worker information and inactivate workers that no longer work for your agency. The following sections describe how these maintenance activities are accomplished.

### 6.1 ADDING A WORKER

Your workers are vital to the KS AuthentiCare process. In order for the IVR system to document services provided by a worker, the worker must be in the system and have a system-generated ID number. This will enable accurate scheduling, use of the IVR and billing for services provided.

Prior to adding a new worker, it is recommended that a search be performed to determine if the worker has already been added. For details on searching, refer to Section 4.4.

1. Click Create in the menu bar and select "New Worker".

#### OR

1. Click **Worker** adjacent to "Add New >" in the Entities section of the *Home* page.

Kansas	Kansas /
Home   Create   Reports   Sche	duling   Dashboards   Visits   Administration   My Acc
	Add New Worker
	Search >Go!





The *Worker Entity Settings* page displays. The *Worker Entity Settings* page is similar to the *Client Entity Settings* page.

Worker Entity Settings		Entity Addresses/Phones
* Indicates a required field.		Add Address
ID:		Add Phone
PIN:		
* First Name		ID Card Number:
Middle Name:		Status:
* Last Name		Holidays / Days Off
Company Names		
SSN:		
FID:		
Gender	×	
* Birth Dates	×	
Email Address		
Begin Date:	×	Add Remove
End Dates	~	Add
Language:	×	
Status:	Active Y	Work Hours
		StartTime End Time Def Off Cust
External Worker ID:		Sun 📃 💿 🔿 🔿
Family Member	<u> </u>	Mon 000
Live with a client you serve:	No V	Tue 000
	Level 2 Attendant Care - FE	Wed 0 0 0
* Worker Services	Self Directed Attendant Care - FE	
	Adult Day Care - FE	Thu ⊙ ○ ○
	Personal Emergency Response - Install - FE 🗾	Fri 000
* Mobile Enabled:	O Yes ⊙ No	Sat 📃 🕘 O O

- 2. Enter the worker's **First Name** and **Last Name**.
- 3. Select the worker's **Gender** from the drop down box.
- 4. Enter the worker's **Birth Date** in mm/dd/yyyy format.
- 5. If the provider chooses, an **External Worker ID** can be entered. Many providers already have an ID number for their workers and want to capture that information here.
- 6. Consider if the worker is working for a **Family Member** (parent, adoptive parent, grandparent, spouse, ant/uncle, sibling, first cousin) and indicate the correct relationship from the drop down box if appropriate.
- 7. Consider if the worker **Lives with the Client** he/she serves. If so, change the box from the default "no" to "yes".

November 14, 2011





- 8. Select the **Worker Services**. You must select at least one service this worker may provide, but may select more than one if needed. To do so, hold down the Control key (Ctrl) on the keyboard while clicking the selections in the list.
- 9. Enter additional information in the appropriate fields if desired.



**Provider** is automatically populated with the name of the provider that corresponds to the user that is presently logged into KS AuthentiCare and creating this worker. <u>Do not use the Add Provider button.</u>

6. Enter the following schedule information, if desired.

Holidays / Days Off	
Friday, January 01, 2010	
1/1/2010 V Add Remove	
From Date To Date	
Ad	d



Entering details in this section causes warning notices to display if the provider attempts to schedule a worker for a planned day off. For details on scheduling, refer to the Chapter 9, Scheduling.

- A) Providers may choose to add **Holidays / Days Off** for each worker to assist with scheduling.
  - To add a single day: Type in the **Date** OR choose a date from the calendar then click **Add**.
  - To add a date range: Enter a **From Date** and **To Date** OR choose the dates from the calendars, and then click **Add**.





I	Ioli	day	s /	Days	s Of	f		
	Frida							
		2010			Add		Remo	ove
	•	Janu		~	_	LO 💙		
	Sun	Mon		Wed		Fri	Sat	Add
н	27	28 4	29 5	30 6	31 7	1	2	
1	10			13		15	16	
	17			20			~~	
	24	25	26	27	28	29	30	Off Cust
	31							0 0
	Q	oday	: 12/	30/20	09			~ ~

- B) Providers may choose to Work Hours for the worker.
  - To apply the **default hours** to the worker, no change is needed.

Work Hours									
	Start Time	End Time	Def	Off	Cust				
Sun			۲	0	$\circ$				
Mon			۲	0	0				
Tue			۲	0	0				
Wed			۲	0	0				
Thu			۲	0	0				
Fri			۲	0	0				
Sat			۲	0	0				

X	The default hours are the work hours of the prov automatically checks the circles in the column law			
	To see the hours set for the provider, click on the link icon next to the provider's name. Clicking this icon displays the Provider Entity Settings page.			
	Add Provider			
	Provider: ARROWHEAD WEST IN Delete			

- To **enter hours** that differ from the provider's work hours for one or more days of the week:
  - Check the circle in the column labeled "Cust" (for Custom)
  - Enter the **Start Time** and **End Time** in the fields provided. Be sure to indicate AM or PM for each entry.





Work Hours					
	Start Time	End Time	Def	Off	Cust
Sun			۲	0	$\circ$
Mon	3:00 PM	7:00 PM	0	0	۲
Tue	9:00 AM	12:00 PM	0	0	۲
Wed			۲	0	0
Thu			۲	0	0
Fri			۲	0	0
Sat			۲	0	0

- To **establish ongoing days off** (for example if the worker never works on Monday or Tuesday):
  - Check the circle in the column labeled "Off".

Wor	Work Hours					
	Start Time	End Time	Def	Off	Cust	
Sun			۲	0	0	
Mon			0	۲	0	
Tue			0	۲	0	
Wed			۲	0	0	
Thu			۲	0	0	
Fri			۲	0	0	
Sat			۲	0	0	

7. Click **Save/Create Another** at the bottom of the page, to save the worker and create another worker.

## OR

8. Click **Save**, to save the worker and return to the *Home* page.

A successful save message displays at the top of the page indicating the worker was saved successfully.







The worker ID appears in this message. The worker must use this ID when calling the IVR from the client's home. Be sure to record this ID on the Worker Instruction sheet to be supplied to the worker. The ID can also be found by viewing the Worker Entity Setting page.

### 6.2 DELETING A WORKER

If a worker is entered in error, the worker can be deleted from the system, but this must be done immediately after the mistake is made. As soon as an event is scheduled or a claim is created for this worker, the worker cannot be deleted.

1. Search for the worker you wish to delete according to the instructions in Section 4.4.

The Worker Entities Settings page displays.

Worker Entity Settings	Entity Addresses/Phones
* Indicates a required field.	Add Address
ID: 19444	Add Phone
PIN: ****	
* first Name: Poul	ID Card Number:
Middle Name:	Status:
* Lest Nome: Sryani	Holidays / Days Off
Company Name:	
55N:	
FID:	
Gender: Mala	
* Sirth Date: 4/24/1950 👻	
Email Address:	
Segin Date: 🛛 👻	Add Remove
End Date: 👻	From Data To Data
Language: N	
Status: Active 💕	Work Hours
	Start Time End Time Def Off Quat
External Worker ID:	Sun 📃 💮 🔿 🔿
Pamily Member: 💦 📎	Man 000
Live with a client you No. No.	Tue 0.0
Level 2 Attendent Care - FS	
Worker Services: Level 1 Attendent Care - FE	i wed ⊕ ⊖ ⊖ .
Adult Day Care - FE	Thu
Personal Emergency Response - Ins	±11-F5 € m
	5at 🛛 💮 🔿 🔿
* Mobile Enabled: O Yes O No	
Password:	
Mobile phone number:	
Device ID:	
Office Phone:	
Add Provider	
Provider: ARROWHERD WEST INC 💁 Delete	
Seve Centr	l Delete

2. Click Delete.

November 14, 2011





* Mobile Enabled:	⊖Yes ⊙No
Password:	
Mobile phone number:	
Device ID:	
Office Phone:	
Add Provider	•
vider: ARROWHEAD WEST	INC 💁 Delete

The system asks you to confirm the deletion. By clicking **OK**, the worker is permanently deleted from the system. If you do not wish to proceed with permanently deleting the worker, then click **Cancel**.

Windows Internet Explorer
Are you sure you want to delete this?
OK Cancel

3. Click **OK** to permanently delete this worker.

You return to the *Home* page. A message displays in the upper left hand corner confirming the deletion. If the worker you attempted to delete had any other relationships in the system, then an error message will display informing you that the deletion was not completed.



OR

1. Enter a name, partial name or ID in the **Search** field and click **Go!** 

Entities				
Add New >	Worker			
Search >				
	Go!			





The Entity Search Results page displays with the results of your search.

Entity Search Results				
ID	Name	<u>User Type</u>		Delete Selected
472	Sooner Healthcare	Provider	6	
<u>123456789</u>	McLain Ward	Client	6	
24684	Betsy Jones	Worker	6	
<u>987654321</u>	Molly Ashe	Client	£	
<u>75771</u>	Brett Johnson	Worker	6	
<u>908765432</u>	Wendy Brown	Client	6	

2. Click the checkbox in the **Delete Selected** column.

Entity Search Results				
ID	Name	User Type		Delete Selected
<u>472</u>	Sooner Healthcare	Provider	£	
<u>123456789</u>	McLain Ward	Client	£	
24684	Betsy Jones	Worker	1	
<u>987654321</u>	Molly Ashe	Client	6	
<u>75771</u>	Brett Johnson	Worker	3	
<u>908765432</u>	Wendy Brown	Client	â	

3. Click the **Delete Selected** column heading to delete the worker you wish to delete.

Entity Search Results				
ID	Name	<u>User Type</u>		Delete Selected
<u>472</u>	Sooner Healthcare	Provider	ŝ	
<u>123456789</u>	McLain Ward	Client	3	
<u>24684</u>	Betsy Jones	Worker	3	
<u>987654321</u>	Molly Ashe	Client	<u>1</u>	
<u>75771</u>	Brett Johnson	Worker	6	✓
<u>908765432</u>	Wendy Brown	Client	3	

The system asks you to confirm the deletion. By clicking **OK**, the worker is permanently deleted from the system. If you do not wish to proceed with permanently deleting the worker, then click **Cancel**.







4. Click **OK** to permanently delete this worker.

You return to the *Home* page. A message displays in the upper left hand corner confirming the deletion. If the worker you attempted to delete had any other relationships in the system, then an error message will display informing you that the deletion was not completed.



## 6.3 EDITING WORKERS

1. Search for the worker you wish to edit according to the instructions in Section 4.4.

The *Worker Entities Settings* page displays.

Worker Entity Settings	Entity Addresses/Phones
* Indicates a required field.	Add Address
ID:	Add Phone
PIN:	
* First Name Paul	ID Card Number
Middle Names	Status:
* Last Names Bryant	Holidays / Days Off
company Names	
SSN:	
FID:	
Gender: Male	
* Birth Dates 4/24/1950 V	
Email Address	
Begin Date: 💌	From Date To Date
End Date:	From Date To Date
Language: 🔽 🗸	
Status: Active	Work Hours
	Start Time End Time Def Off Cust
External Worker ID:	Sun 📃 🖲 O O
Family Member	Mon 0 0
Live with a client you No M	Tue 000
Level 2 Attendant Care - FE	Wed
* Worker Services: Level 1 Attendant Care - FE	
Adult Day Care - FE	Thu ⊙ ○ ○
Personal Emergency Response - Inst	all - FE 💌 Fri 📃 ⊙ O O

2. Verify, and if needed, update the appropriate fields such as **First Name**, **Last Name**, **Gender**, **Birth Date**, **Worker Services**, **Family Member**, **Living with a Client** and scheduling information.

November 14, 2011





3. Click **Save** at the bottom of the page to save the worker and return to the *Home* page.

A successful save message displays at the top of the page indicating the worker was saved successfully.

### 6.4 SUSPENDING/INACTIVATING/END DATING A WORKER

A worker cannot be deleted once there are any relationships created for the worker. In other words, if a worker has been scheduled for an event or if a claim has been created for which the worker provided the service, then the worker cannot be deleted from KS AuthentiCare. If the worker no longer works for the provider or simply will not be providing services any longer, there are three options for editing the worker file:

- 1. Change the worker's status to "suspend". The worker <u>cannot</u> use the IVR but claims can be entered via the web. All claims (pending and new) will show a critical exception that the worker is not eligible.
- 2. Change the worker's status to "inactive". The worker can still use the IVR and have claims entered via the web but all claims (pending and new) will show a critical exception that the worker is not eligible.
- 3. Populate the "End Date" field on the worker's screen. The worker can still use the IVR and have claims entered via the web. Claims with a Date of Service (DOS) prior to the end date will be processed normally. Claims with a DOS after the end date will generate a critical exception that the worker is not eligible.

At any time the "End Date" can be removed or the worker's status changed back to active, making the worker again eligible to provide services and/or removing the "worker not eligible" critical exception from pending claims.

To change suspend/inactive/end date a worker:

1. Search for the worker you wish to inactivate according to the instructions in Section 4.4

The Worker Entities Settings page displays.

2. Change **Status** to Suspend or Inactive or enter an End Date.





Worker Entity Settings		
* Indicates a required field.		
ID:	58745	
PIN:	••••	
* First Names	Eric	
Middle Name:		
* Last Name	Burt on	
Company Names		
SSN:		
FID:		
* Gender:	Male 💟	
* Birth Dates	1/4/1987 💌	
Email Address		
Begin Date:	<u>~</u>	
End Da	8/23/2010 💌	
Language:	×	
* Statu	Inactive 💙	

3. Click **Save** at the bottom of the page to save the worker and return to the *Home* page.

A successful save message displays at the top of the page indicating the worker was saved successfully.





If a worker is rehired, that worker should not be re-entered because the worker still exists in the system. Instead, the **Status** should be changed back to Active and/or the End Date removed.





## Chapter 7 MANAGING CLIENT (CONSUMER OR CUSTOMER) INFORMATION

Consumers or customers are referred to as clients in KS AuthentiCare. Clients are created in KS AuthentiCare by file transfer from KMAP/MMIS. You must verify your client list and confirm that a correct telephone number(s) is provided (unless the client does not have a phone).

1. Search for the client you wish to edit according to the instructions in Section 4.3.

The Client Entity Settings page displays.



2. Verify the **Phone Type** and **Phone Number** for the client. If the client does not have a phone, then this field should be populated with 999-999-9999.

You manage the client's telephone information. You can correct or delete an incorrect number and add a correct or additional numbers. The number(s) listed is the one KS AuthentiCare will use to verify if the worker is calling from the client's home.

3. Click **Save** at the bottom of the page to save the client and return to the *Home* page.

A successful save message displays at the top of the page indicating the client was saved successfully.







## Chapter 8 SEARCH AND VIEW AUTHORIZATIONS

A service to be provided for a client must have a valid authorization documented in KS AuthentiCare in order to be scheduled and for the claim to be submitted for payment.



A worker can use the IVR to record services without a valid authorization in KS AuthentiCare. A claim will be created but will have a critical exception and will not be submitted for payment until the authorization is in place.

All authorizations for clients are loaded into KS AuthentiCare by file transfer from KMAP/MMIS. The provider cannot add a new authorization or edit an existing authorization.

To search and view an authorization:

- 1. Click the **Authorization** radio button in the Services and Authorizations section of the *Home* page.
- 2. Enter search criteria in any of the fields, if desired and click Go!

Services and Authorizations				
Search Type:	<ul> <li>○ Service</li> <li>O Authorization</li> </ul>			
Service:				
Authorization ID:				
Scheduling Status:	<b>~</b>			
Service Type:	×			
Authorization Start:	✓			
Authorization End:	×			
Client:				
Provider:				
Worker:				
Procedure Code:				
Modifier:				
Billing Location Code:	12 Home			
	Go! Clear			

The search results display all authorizations which match the search criteria entered on the *Home* page.

Authorizations Event Actions	ID	Service ID	Client	Provider	Worker	Effective Dates	
Units Remaining: 3000000	<u>415</u>	Personal Assistant Services (HCDDT1019)	Mouse, Minie Z (2222222222)	General Health Care (462880000A)		11/01/2011 - 11/30/2012	ŝ
ovember 14, 2011				KS Authe	entiCare	e User Manual	





If nothing is entered in the **Search** fields, then all authorizations will be returned in your search results. This could take awhile and it may also make it difficult to find the authorization.

- 3. Note the columns displayed in the search results:
  - Event Actions Click the Schedule Event icon 2 to schedule an event for this authorization. Refer to Chapter 9, Scheduling, for further instructions. This column also indicates how many units remain in the authorization by comparing the total number of units authorized and the total number of units scheduled. This is **not** a comparison with the units actually provided to date but with those scheduled to date.
  - **ID** Click the ID hyperlink to view/edit the *Authorization Settings* page for this Authorization. Proceed to Step 7.
  - Service ID Identifies the service and its service code
  - **Client** Identifies the client and the Medicaid ID
  - **Provider** Identifies the provider assigned to the authorization
  - **Worker** Field is blank as authorizations are issued to the provider, not the provider's individual workers.
  - Effective Dates Identifies the start and end dates of the authorization
- 4. Click the column heading if you wish for the search results to sort using a different column than the default, which is the ID (Authorization) column in ascending order. Click the heading once to change the sort to descending order for that column. Click the heading again to change it to ascending order for that column.
- 5. Position the cursor over the **Information icon** to display an Additional Information pop-up about the authorization.
- 6. Click on the **ID** hyperlink in the **ID** column to view the authorization.

The Authorization Settings page displays.





Authoriza	tion Settings			
* Indicates a	a required field.			
Service Info	ormation			
Service ID:	HCDDT1019		Service Type:	Time Based
Name:	Personal Assista	nt Services	Procedure Code	: T1019
Description:	Personal Assista	nt Services	- MRDD	
	ID:	415		
	Client:	Mouse, Min	ie Z 🔄	
	Provider:	General He	alth Care 嶜	
	Worker:			
Effe	ctive Date Start:	11/1/2011		
Eff	ective Date End:	11/30/2012	2	
	* Total Units:	3000000		
	* Rate:	3.6500		
				Cancel

7. Click Cancel to return to the Authorization Search Page.





# Chapter 9 SCHEDULING

Providers may use KS AuthentiCare to schedule workers' visits to clients' homes. A scheduled visit is called an event. Each event must be linked to an authorization. Events may only be scheduled for authorized client/service/provider/worker combinations. KS AuthentiCare tracks the number of units available for scheduling in each authorization and notifies the scheduler if an event exceeds the number of units available for scheduling. KS AuthentiCare also allows providers to schedule back up workers for events.

KS AuthentiCare compares the scheduled events to the actual check in and checkout times of the worker and alerts the provider via email if a worker is late or if an event is missed.

KS AuthentiCare automatically identifies for the scheduler any conflicts the scheduled event creates with the worker's schedule. For example, if a worker is scheduled for an event on a regularly occurring day off, KS AuthentiCare displays a warning message alerting the scheduler to this conflict. The scheduler may choose to change the event or to save it even though a conflict exists. KS AuthentiCare's scheduling warnings are meant to assist schedulers but not prevent the scheduler from scheduling an event with a conflict.

This chapter is designed to help you schedule non-recurring and recurring events. Additionally, it helps you search for an event, view an event, maintain events, and acknowledge a missed event when a worker misses a scheduled event.

## 9.1 USING THE CALENDAR

KS AuthentiCare allows you to view your calendar of events by the month, week or day. This section describes the three views.

1. Select **Scheduling** from the Main Menu on the Home page.



2. The *Event Scheduling* page displays with the calendar.





Event Sc	heduling					Cr	eate New Event	Generate R	eport
Search Sch Date:	hedules Clier	ntc		Provider:	Worker:		Service		
	~		2					Search	Clear

Calendar View: <u>Monthly</u> | <u>Weekly</u> | <u>Daily</u>

4			Novembe	er 2011		×
	Mon	Tue	Wed	Thu	Fri	Set/Sun
43	Gel 24	25	28	27	25	29
44	21	Nev 1	1	5	4	5

The default view of the calendar is the monthly view.

### 9.1.1 MONTHLY

The Monthly view of your calendar enables you to view all of the events scheduled for your agency for one month at a time. This is the default setting and the view you initially see when navigating to this page.

1. Position the **cursor over the event** and a pop-up box with additional detail displays.



2. **Double click on an event** to view all of the event information.





The Scheduled Event page displays.



- 3. Click **Eventscheduling** in the top left corner of the page to return to the calendar or click the Back button on the browser.
- 4. Click the left and right arrows in the date display bar to view the month before or the month after this date.

Calendar View:	Monthly   Weekly   Daily					
4			Decembe	r 2009		
	Mon	Tue	Wed	Thu	Fri	Sat/Sun
	Nov 23	24	25	26	27	28

### 9.1.2 WEEKLY

The Weekly view of your calendar enables you to view all of the events scheduled for your agency for one week at a time.

- 1. From the *Monthly* view, **click a day in the week** you would like to view.
- 2. Click the **Weekly** link above the calendar.

(	Calendar View: <u>Monthly</u>   <u>We</u>	ekly	Daily
	4		
	Mon	1	
		Nov 23	





The week schedule that includes the day you selected displays.

Calendar View: <u>Monthly   Weekly</u>   <u>Daily</u>	
C December 2	8 - January 3 🔋
Monday, December 28	Thursday, December 31
Tuesday, December 29	Friday, January 01
Respite In-Home Worker: Wendy Brown Client: McLain Ward	4:00am - 5:00am Respite In-Home Worker: Jennifer Alfano Client: Joshua Waters
Wednesday, December 30	Saturday, January 02
	Saturday, January 02
Respite In-Home Worker: Wendy Brown Client: McLain Ward	
8:00am - 9:00am Respite In-Home Worker: Brett Johnson Client: Joshua Waters	Sunday, January 03
1:00pm - 2:00pm Respite In-Home Worker: Brett Johnson Client: McLain Ward	

3. Position the cursor over the event and a pop-up box with additional detail displays.

	Wednesday, December 30
Respite In-Home Worker: Wendy Brown Client: McLain Ward	
8:00am - 9:00am Respite In-Home Worker: Brett Johnson Client: Jos Respite In-Home 1:00pm - 2:00pm Respite In-Home Worker: Brett Johnson Client: Mc Client: McLain Ward	

4. Double click on an event to display the *Scheduled Event* page and see the details of the event.

	* Date	* Start 1	fime * End	l Time	* Duration
vent:	12/29/2009	▼ 11:00 P	M 01:0	0 AM	02:00
	Recurring	Event	12/3	0/2009	
	* Auth	orizations:			0
	Prima	Client: M Provider: S	117 🞑 AcLain Ward Gooner Health Nendy Browr		
		up Worker:	Vendy Brown		

5. Click **Eventscheduling** in the top left corner of the page to return to the calendar or click the Back button on the browser.





6. Click the left and right arrows in the date display bar to view the week before or the week after this date.

December 28 - January 3           Monday, December 28         Thur.	
Monday, December 28 Thur.	7
	sday, December 31

9.1.3 DAILY

The Daily view of your calendar enables you to view all of the events scheduled for your agency for one day at a time.

- 1. From the *Monthly* view, **click a day** you would like to view. Or from the *Weekly* view, **click a day** you would like to view.
- 2. Click the **Daily** link above the calendar.



The day displays in hourly increments.

Calendar \	View: Monthly   Weekly   Daily
4	
	Wednesday, December 30
8 00	Respite In-Home
9 <sup>00</sup>	
10 00	
11 <sup>00</sup>	
12 pm	
1 <sup>00</sup>	Respite In-Home
2 <sup>00</sup>	v
<	2

3. Position the cursor over the event and a pop-up box with additional detail displays.







4. Double click on an event, to see the details and display the *Scheduled Event* page.

Sched	uled Event									
	* Date	* Start	Time	* End Ti	ne	* Durat	ion			
Event:	12/30/2009	✓ 08:00 /	AM	09:00 A	М	01:00				
	Recurring E	Event		12/30/2	009					
	* Auth	orizations:			5	6				
420 🔕 Client: Joshua Waters										
		Provider:	Sooner	Healthca	re 🔄					
	Prima	ry Worker:								
	Back	ıp Worker:								
		* Service:	Respite	In-Home	2					
				Delet	e S	ave (	ancel			

- 5. Click **Eventscheduling** in the top left corner of the page to return to the calendar or click the Back button on the browser.
- 6. Click the left and right arrows in the date display bar to view the day before or the day after this date.

Calendar View: <u>Monthly</u>   <u>Weekly</u>   <u>Daily</u>		
*	Wednesday, December 30	<b>/</b>
12 am Respite In-Home		-

### 9.2 SEARCHING FOR SCHEDULED EVENTS IN THE CALENDAR

After an event has been scheduled, you are able to search for this event in the calendar. KS AuthentiCare provides a variety of options for searching for an event in the calendar.

1. Navigate to the *Event Scheduling* page.





Home	Create	Reports   S	cheduling   Dashboards	Visits   Administration   M	y Account   Custom Links	Logout	Logged in as: g	rant@new.com
		Event Schee	duling		Create No	ew Event	Generate Report	
		Search Sched Date:	Client	Provider: W	orker: Servic	.e	Search Clear	
		Team Assignr	nent 🔽					
Calend	lar View:	Monthly   Wee	kly   <u>Daily</u>					
4	_			Jun	≥ 2011			×
		Mon May 23	Tue	24 Wed	Thu 5 28	Fri	27 27	/Sun 28
21								29
22		:	•	31 Jun	1		3	-
22								

- 2. Enter at least one of the following **search criteria** to locate event(s) in the calendar. Each search criteria field is described below:
  - a. Enter a **Date** (mm/dd/yyyy) or click the down arrow and select the date from the calendar, then click **Search**.

Event Sch	eduling			(	ireate New Even	t	Generate	Report
Search Sch	edules							
Date:	Client	Providen	Worken		Service			
	*						Search	Clear

The calendar displays events from this date forward.

b. Enter the **Client** ID, full name or partial name and click the **Looking Glass icon** it to find the client, and then click **Search**.

Event Sch	eduling		1		(	Create New Eve	nt	Generate Re	eport
Search Sch	edules	1							
Date:	Client	-	Providen	Worken		Service			
	¥							Search	Clear

The calendar displays all the events scheduled for this individual.





c. Enter the **Primary Worker** ID, full name or partial name and click the **Looking Glass icon** ind the worker, and then click **Search**.

Event Sch	eduling			(	reate Nev Event	Gene	erate Report
Search Sche Date:	edules Client	Providen	Worker	*	Service		
	✓					Sea	rch Clear

The calendar displays all the events scheduled for this person.

d. Enter the **Service** name or partial name and click the **Looking Glass icon** in the services, and then click **Search**.

Event Sche	eduling			reate New Eve		enerate Report
Search Sche Date:	dules	Providen	Worker	Service	1	
	¥					Search Clear

The calendar displays all the events scheduled for this Service at your agency.



The more search criteria you enter, the more narrow the search results. Be aware, there is the potential no results will display because some of the criteria may conflict.



Click **Clear**, if you have entered incorrect information in the search criteria fields at any time or would like to start a new search.

## 9.3 SCHEDULING AN EVENT

Events are the visits the worker makes to the client to deliver an authorized service. Events can be of varying durations, at any time of the day and on any day of the week. An event may be a one-time service (non-recurring) or a service that is provided on an ongoing, regularly scheduled basis (recurring).

### 9.3.1 SCHEDULING AN EVENT USING THE AUTHORIZATION

An authorization has been entered in the system and now you are ready to schedule the visit for this client. Follow the instructions below to add the event. These instructions assume you

November 14, 2011





are viewing the *Authorizations* page. Refer to Chapter 8, Managing Authorizations if you need further instructions about authorizations.

1. Click the **Schedule Event icon** for the appropriate authorization on the *Authorizations* page.

Authorizations	/	e					
Event Actions	ID	Service ID	Client	Provider	<u>Worker</u>	Effective Dates	
Units Remaining: 3000000	<u>415</u>	Personal Assistant Services (HCDDT1019)	Mouse, Minie Z (2222222222)	General Health Care (462880000A)		11/01/2011 - 11/30/2012	î

The *Scheduled Event* page displays.

	* Date	* Star	t Time	* End T	ime	* Duration
Event:		<b>~</b>				
		Event				
	* Author	rizations			0	
			415			
		Client	Mouse, I	Minie Z 🛛	2	
	1	Provider	General	Health C	are	
	Primary	y Worker:			2	
	Backup	Worker			2	
	*	Service	Persona	al Assista	antServ	/ices



The Authorization ID number is populated below the **Authorizations** field. Additionally, the **Client** and **Service** are completed since you navigated to this page from the **Authorizations** page. The **Provider** is populated based on the provider agency associated with the current log in.

- 2. Enter the **Date** of the event or select a Date from the drop-down calendar.
- 3. Enter **Start and End Times**. Start and end times must include AM or PM. The **Duration** is automatically calculated by KS AuthentiCare.

November 14, 2011





Sched	uled Event				
	* Date		* Start Time	* End Time	* Duration
Event:	11/17/2011	~	9:30AM	10:30 AM	01:00
		Εv	ent	(11/17/2011)	

When a worker visits a client one time, or a variety of times not in a normal, repeated schedule, this is considered a non-recurring event. For these types of events, proceed to Step 4. When a worker visits a client on a regular basis, at the same time of the day, this is considered a recurring event. Refer to *Section 9.3.3* for further information.

- Enter the Primary Worker. You may enter the worker ID, a full name or partial name. Then click the Looking Glass icon . Select the worker from the list displayed. You may leave this field blank if you do not know who will be conducting the service.
- 5. Enter the **Backup Worker**, if applicable. Enter a worker ID, full name or partial name and click the **Looking Glass icon**.
- 6. Click Save.

If KS AuthentiCare discovers scheduling conflicts or issues, a message displays at the top of the page advising the user of the conflict. In this example the event scheduled for January 4, 2010 exceeds the total units remaining for this authorization. The authorization for this event has 1 unit remaining (15 minutes) and the event scheduled requires 4 units; therefore, the message indicates that the event will exceed the total units allowed on the authorization.

Event Acknowledgment			
Authorization 421: This event v	vill exceed the total units allow	ed on the authorization.	
<b>Start</b> Jan 04, 2010 1:00 PM		<b>End</b> Jan 04, 2010 2:00 PM	
Recurrence None	,		
Client <u>Wendy White (987654321)</u>	Provider <u>Sooner Healthcare (427)</u>	Primary Worker <u>Jennifer Alfano (40961)</u>	Backup Worker None
	AuthorizationService421Respite In-H	<u>łome (T1005)</u>	
		Discard Acce	pt Change

- Click Change to return to the event and change the information, OR
- o Click Accept to accept the event with the conflict, OR





o Click **Discard** to discard the event and return to the Scheduled Event page.

#### Messages you may receive when scheduling an event.

- This event will exceed the total units allowed on the authorization.
- This event occurs outside the effective dates of the authorization.
- This event falls on the worker's day off.
- The status of the primary worker is inactive or the event date is outside the effective date range.
- The status of the client is inactive or the event date is outside the effective date range.
- This event falls outside the range of the worker's assigned business hours.
- This event falls outside the range of the backup worker's assigned business hours.
- This event will exceed the worker's assigned hours for the week.
- This event falls on a provider holiday.
- The status of the provider is inactive or the event date is outside the effective date range.
- This event falls on the backup worker's day off.
- The status of the backup worker is inactive or the event date is outside the effective date range.

If no scheduling conflicts or issues arise, the *Authorizations* page displays a note at the top of the page indicating the event was scheduled successfully.



### 9.3.2 SCHEDULING A RECURRING EVENT

When a worker visits a client on a regular basis, at the same time of the day, the provider can establish an event that occurs repeatedly. This is considered a recurring event. This is an





efficient way to schedule as you only have to enter the basic information one time but can use it to populate multiple days.

A recurring event can be created for as long as necessary within the limits set by the authorization.

- 1. Complete **Steps 1 3** from Section 9.3.1 or Section 9.3.2.
- 2. Click the **Recurring Event** check box.

Schedu	uled Event					
:	* Date	* Star	rt Time	* End 1	ime	* Duration
Event:		~				
	Recurring	Event				
	* Autho	rizations			¢	1
			415		_	
		Client	Mouse,	Minie Z	2	
		Provider:	General	Health C	Care	
	Primar	y Worker				
	Backuj	worker:				
	3	Services	Persona	al Assist	ant Sen	vices
					S	ave Cancel

The page expands to display scheduling options.

Scheduled Event					
Event: 1	Date         * Start Time         * End Time         * Duration           //17/2011         9:30AM         10:30 AM         01:00           Recurring Event         (11/17/2011)         10:30 AM         10:30 AM				
⊙ Daily ⊙ Every =) O Weekly ○ Every =) O Monthly ○ Every weekday					
⊙ End afterrrences ○ End by					
* Authorizations					
Client: Mouse, Minie Z 💁 Provider: General Health Cars Primary Worker:					
Backup Worker:					
	Save Cancel				

November 14, 2011





3. Follow the steps outlined below to create a Daily, Weekly or Monthly recurring event.

#### Daily Recurring Event

KS AuthentiCare defaults to **Daily** recurrence. To schedule an event to occur more than once in the same week, use the Daily occurrence option.

- a. Select the **frequency** of the recurring event by entering the number of times the event should occur (every 2 days or every 3 days for example) or by choosing every weekday.
- b. Enter the **number of occurrences** of the event or enter an **End by** date for the event.

Scheduled Event						
Event:	* Date 11/17/2011	* Start Time • 9:30AM Event	* End Time 10:30 AM (11/17/2011)	* Duration 01:00		
O Baily O Weekly O Monthly O Monthly						
⊙ End after 5 rrences ○ End by						

### Weekly Recurring Event

To schedule an event to occur on the same day or days for multiple weeks, select the **Weekly** occurrence option.

- a. Select the **frequency** of the event by entering the number of times the event should occur (every week, every 2 weeks for example).
- b. Select the **day or days** of the week the event should occur.
- c. Select the **number of times** the event should occur or enter an **End by** date for the event.




Scheduled Event								
* Date * Start Time * End Time * Duration								
Event:	11/17/2011 💌	9:30AM	10:30 AM	01:00				
	✓ Recurring Event (11/17/2011)							
O Daily Recur every :(s) on: Weekly Sunday Monday Tuesday Wednesday O Monthly Thursday Friday Saturday								
<ul> <li>○ End after rrences</li> <li>⊙ End by 2/24/2012 </li> </ul>								

#### Monthly Recurring Event

To schedule an event to occur on a monthly, bi-monthly or quarterly basis, select the **Monthly** occurrence option.

- a. Select the **frequency** of the event by entering the calendar day of the month (the 15th day of every month or the 1st day of every 2nd month, for example) or by entering the weekday of the month (the 1st Monday of every month or the last Friday of every 3rd month, for example).
- b. Select the **number of times** the event should occur or enter an **End by** date for the event.

Scheduled Event						
	* Date	* Start Time	* End Time	* Duration		
Event:	11/17/2011 💌	9:30AM	10:30 AM	01:00		
	✓ Recurring Event (11/17/2011)					
O Daily O Weekly O Monthly ○ The First ♥r Mondav						
O End after rrences						
	۲	End by 2/24/20	012 🔽			

4. Proceed with Step 4 in Section 9.3.1or in Section 9.3.2.





## 9.4 EDITING AN EVENT



If the date or time of a scheduled event has already passed, the event cannot be edited or deleted.

When information changes for an event, KS AuthentiCare allows you to make adjustments up to the last minute. The following instructions discuss editing an event **Date**, **Time**, **Client**, and/or **Worker**.

- 1. Click on **Scheduling** from the Main Menu on the *Home* page.
- 2. Find the event on the calendar that needs to be changed (use Search or various views available to do so). Double click on the **event** on the calendar to display the details in the *Scheduled Event* page.

$\triangleleft$			Scheduled Event
	Mon		
	Dec 28		This is a recurring event. Do you want to edit only this occurrence or the series?
52			O Edit this occurrence
22			O Edit the series
	4		* Date * Start Time * End Time * Duration
	1:00pm Respite In-Home Worker: Betsy Jones C		Event: 1/1/2010 🕑 01:00 PM 02:30 PM 01:30
1	1:00pm - 2: 	lome	Recurring Event 01/01/2010
	Worker: Be Client: McLa		O Daily Recur every 2 week(s) on:
2			
			O Monthly     Thursday     Friday     Saturday
			<ul> <li>End after 2 occurrences</li> </ul>
		F	○ End by
			* Authorizations:
			418 🛛
			Client: McLain Ward 🛛 🗟
			Provider: Sooner Healthcare 嶜
			Primary Worker: Betsy Jones 🔄
			Backup Worker:
			* Service: Respite In-Home
			Delete Save Cancel







3. Click on the **field** for the item to be corrected. Initially it looks like the fields are not editable, but once you click on the field it changes to an editable field.



4. Enter the new information.



5. Click Save.







If KS AuthentiCare discovers scheduling conflicts or issues, a message displays at the top of the page indicating the conflict the system has found. Refer to Section 9.3.1 for further instructions. Along with the message are three options from which to choose for how the system should proceed. Select one of the following three options:

- Click Change to return to the event and change the information, OR
- o Click Accept to accept the event with the conflict, OR
- Click **Discard** to discard the event and return to the *Scheduled Event* page.

If no scheduling conflicts or issues arise, a message displays at the top of the page indicating the event was scheduled successfully.



## 9.5 CANCELLING AN EVENT (DELETING)

If you need to delete an event that means it was either entered in error or needs to be cancelled. <u>Cancellations must be done prior to the start time of the event.</u> Once the event start time has passed, the system will not allow you to cancel (delete) the event.

1. Double click on the **event** on the calendar to display the details in the *Scheduled Event* page.

	* Date		* Start Time	* End Time	* Duration	
Event:	12/30/2009	~	01:00 PM	02:00 PM	01:00	
	Recurring	Eve	nt	12/30/2009		
	* Aut	hori	zations:		-	
		ary 1 kup 1	417 Client: McLain rovider: Soone Norker: Brett : Norker: Service: Respi	n Ward 🛛 🕅 er Healthcare 🕅 Johnson 🕅	i	

2. Click Delete.







Once you click Delete, the event is permanently deleted. There is no pop up dialog box asking you to confirm the deletion, so prior to clicking Delete, be sure this is the event you want to delete.

The *Event Scheduling* page displays with a note at the top indicating the event was deleted successfully.

 Home
 Create
 Reports
 Scheduling

 Successfully deleted the scheduled event.



## 9.6 ACKNOWLEDGING MISSED VISITS

When a worker uses the IVR or the GPS mobile device either to check in or check out, the information is captured in KS AuthentiCare immediately. You may need to refresh your screen for updates to display depending on what you are doing in the system.

KS AuthentiCare links worker check ins and check outs to scheduled events continuously and provides notice of late and missed visits to providers via email so that back-up coverage can be initiated. In addition to email message, these late and missed visits are posted on the *Late and Missing Events* page of the system.

**Definition of a Late Visit:** The worker does not check in within 60 minutes of the event **start** time. The Late Visit is either removed because the worker checks in or it becomes a Missed Visit if the worker does not check in within the next 60 minutes (120 minutes after the event start time).

```
November 14, 2011
```





**Definition of a Missed Visit:** The worker does not check in within 120 minutes of the event **start** time.

The late and missed visits are recorded in three ways:

- 1. The *Late and Missing Events* page lists all scheduled events that are late or missed. This page is available from the menu bar to the following sub-roles: Administrator, AdminAssistant and Scheduler/Coordinator.
- 2. The Late/Missed Visits Report. Refer to Chapter 12, *Reporting* for more information.
- 3. An email is sent to the email address on the provider's record each time an event is late and/or missed. Examples of these emails are included below.

#### Late:

This is a KS AuthentiCare late visit notification. Worker John Jones (ID Number) for provider Senior Helpers (WMIS ID) was scheduled to provide service Level 2 Attendant Care (Service ID) on 01/05/2012 at 09:00 PM. A KS AuthentiCare visit is considered late when the service is not provided within 60 minutes of the scheduled <u>start</u> time. To view this scheduled event in more detail, log in to the AuthentiCare website at: https://acr.fdgs.com/Kansas.

Please do not reply to this email as it is intended for notification purposes only.

#### Missed:

This is a KS AuthentiCare missed visit notification. Worker John Jones (ID number) for provider Senior Helpers (WMIS ID) was scheduled to provide service Level 2 Attendant Care (Service ID) on 01/04/2012 at 09:00 PM but missed the appointment. A KS AuthentiCare visit is considered missed when the service is not provided within 120 minutes of the scheduled <u>start</u> time. To view this scheduled event in more detail, log in to the KS AuthentiCare website at: https://acr.fdgs.com/Kansas.

Please do not reply to this email as it is intended for notification purposes only.

 To view the list of late and missed visits, place your cursor on Late and Missed Visits in the toolbar of the Main Menu, and when Late and Missed Visits displays below, click on it.







You will see a selection criteria screen that allows you to choose the visits you want to work. If you do not choose, all visits will be displayed.

Start Date:	~
End Date:	~
Client:	
Provider:	
Worker:	
Service:	2

The *Late and Missing Events* page displays if there are scheduled events for this provider which have been identified as late or missed.

5. Select the appropriate missed visit code from the missed visit code dropdown list for each of the missed visits listed on the *Late and Missing Events* page. Do not choose missed visit codes for events that have a status of "Late". There is a note section for each missed visit for additional documentation if needed.

Ka	nsas	dauthaling   Dauthbour	de l'anato i i		1 March 1 / June			thentiCare	×	Ø
Løbe an	d Missing Events							1		
Status	Start Date/Time	End Date/Time	Recurring?	Client	Provider	Primary Worker	Backup Workor	Sente	Mined Vint Code	Notes
wanted.	11/03/2011 09:10	11/03/2011 09:40		STLERI, DREW	CENTRAL PLAINS	Braskars, Bill		Level 3 Attendant		

Description	Code
Hospital	1
Nursing Facility temporary	2
Consumer refusal	3
Consumer not available – justify in notes	4
Family voluntarily provided temporary service instead	5





Description	Code
Other (consumer driven) – justify in notes	6
Consumer deceased	7
No staff available	А
No staff willing	В
Natural disaster	С
Late plan from Case Manager	E
Other (provider driven) – justify in notes	F
Not a Missed Visit	Ν

 Click the Save Missed Visit Code link column heading. This saves the missed visit codes which can be viewed on the Late and Missed Visits Report (refer to Section 13.5.6 for further information). A missed visit will remain on the list until it is acknowledged by the provider.

Late an	ate and Missing Events								
<u>If you h</u>	If you have not received emails informing you of these late / missed visits, please click here and verify your email address.								
Status	Start Date/Time	End Date/Time	Recurring?	Client	Provider	Primary Worker	Backup Worker	Service	Save Missed Visit Code
Missed	02/18/2010 07:30 AM	02/18/2010 08:00 AM	No	McLain Ward	Sooner Healthcare	Betsy Jones		Personal Care	Hospital (1)
Missed	03/01/2010 01:00 PM	03/01/2010 02:00 PM	Yes	Paul McCartney	Sooner Healthcare	Betsy Jones		Personal Care	Not a Missed Visit (N)
Missed	03/08/2010 01:00 PM	03/08/2010 02:00 PM	Yes	Paul McCartney	Sooner Healthcare	Betsy Jones		Personal Care	Consumer refusal (3)

You return to the *Home* page which displays a message in the upper left hand corner that the missed visit codes have been saved successfully.







# Chapter 10 THE KS AUTHENTICARE IVR

Much of the functionality and data described in Chapters 2 through 9 leads to in-home workers using the Interactive Voice Response (IVR) to record services provided for the client by calling in from the client's home when service begins and calling out from the client's home when service is completed. Each worker has a 5-digit worker ID number that identifies him/her as a worker for a specific provider location. That worker ID is recorded in the IVR each time the worker makes a call.



Important – If a worker works for more than one provider location, he/she is assigned a different number for each location. Caution workers to make sure they use the correct Worker ID for each client visit.

## 10.1 IVR FLOW

The IVR (available in English and Spanish) is designed to capture the information required to create a claim for the service being provided. Section 10.3 below walks the worker step-by-step through what he/she can expect to hear when calling KS AuthentiCare.

If the phone number the worker is calling from matches the number for the client as recorded in KS AuthentiCare, then the client's name will be read by the IVR. If the system does not recognize the number, then the worker must enter the Medicaid ID number of the client.



When checking out, if the IVR reads more than one name, that means the worker has not checked out for previous claim(s) for which he/she had checked in. In order to resolve this, the claim(s) must be completed on the web with the appropriate check out times.

The IVR then reads the list of services that the worker could potentially be providing for this client. For consistency, the services for a specific provider are read in the same order on all calls. This same list also applies during check out calls. Additionally, during check out the worker must specify activities (tasks) completed during the visit if the services provided required activity codes.



A worker can use the IVR even if there is no authorization yet for the client. However, the client <u>must exist</u> in KS AuthentiCare in order for the call to be completed.

The IVR then reads back all of the information in order for the worker to verify its accuracy. If there are any errors, the worker has the option to start over and correct the errors. If the information is correct, then the call is completed and the worker is checked in or out depending on the option chosen at the beginning of the call.

November 14, 2011





If the worker is there to provide services for more than one client, the worker must check in for the first client and at the end of the call when given the option to return to the main menu, choose that option and check in for the second client. The worker can repeat this process as many times as necessary.

When checking out, the worker will need to follow the same process – check out for the first client, return to the main menu as needed to check out for additional clients.

Each time the worker returns to the main menu on either a check-in or check out call, the beginning time of the call is reset.

## **10.2** EXTRA SAFEGUARD FOR WORKERS WITH MORE THAN ONE WORKER ID

Workers who have more than one worker ID because they work for more than one provider cannot accidently sign in using the ID not matched to the client. If they try, the IVR will play the following message "You have entered an incorrect worker ID. Please enter a different worker ID followed by the pound sign."





## 10.3 WORKER INSTRUCTIONS FOR USING THE IVR

# Worker\_\_\_\_\_

Г

Worker ID \_\_\_\_\_

Instr	ructions to <b>Check-in</b>
1	Dial 1-800-903-4676 from the client's touch-tone phone.
2	Enter your worker ID number followed by the pound (#) sign when prompted.
3	Press 1 for Check-in
4	You will then hear the name of the client you are there to serve. If it is correct, press 1.
	If KS AuthentiCare does not recognize the phone number you are calling from, you will be asked to enter the client's ID number (Medicaid number) followed by the pound (#) sign.
5	You will hear a list of services available for the client and be asked to choose the one you are there to perform by pressing the appropriate number on the phone key pad.
6	KS AuthentiCare will then repeat back your name, your agency's name, the client's name, and the service to be provided. If this is all correct, press 1. If the information is not correct press 2 and you will be able to correct the information before you finish the call
7	If the information is correct you will be told that the check-in was successful at (states the time). At this point you will be instructed to press 2 to end the call.
Instr	uctions to Check-out
1	Dial 1-800-903-4676 from the client's touch-tone phone.
2	Enter your worker ID number followed by the pound (#) sign when prompted.
3	Press 2 for Check-out
4	If you failed to check in, the IVR will read the client back to you or, if it does not recognize the phone number you are calling from, you will be asked to enter the client's ID number (Medicaid number) followed by the pound (#) sign. You will also be asked to select a service.
5	Some services will require the entry of activity codes. You will be prompted to enter the activity codes one at a time. After the entry of each code, press the pound (#) sign. KS AuthentiCare reads the activity code, asks you confirm it is correct and then asks you to enter another activity code if needed. Once you have entered all activity codes, press 8 to continue to the next step.



П



6	KS AuthentiCare will repeat back your name, your agency's name, the client's name and the service you provided. If this is all correct, press 1. If the information is not correct press 2 and you will be able to correct the information before you finish the call
7	If the information was correct you will be told that you have successfully filed your claims and the time and press 2 to end your call.

What do I do if	
I forget my worker ID or my client's Medicaid ID?	Call your supervisor who has both numbers on file.
I check-in but forget to check-out?	Call your supervisor and let him know what client you were serving and the time you left the client's home.
I forget to check-in?	If you are near the beginning of your visit, go ahead and do a check-in. Then let your supervisor know the check-in was phoned in late and what time you started providing care. If you don't remember until the end of your visit, go ahead and check-out when you leave. Let your supervisor know you forgot to check-in and what time you arrived at the client's home.
I forget to check-in and check-out?	Call your supervisor and explain what happened.
I am in the process of checking in and realize I have made a mistake?	KS AuthentiCare will let you change the information before you complete the check-in. You can go back by pressing 2 at the confirmation heard during Step 6 of the Check-in process or Step 6 of the Check-out process. Re-enter the correct information when prompted.
I have already checked in and realize I made a mistake?	Go ahead and check out but call your supervisor and explain what happened.
I have checked in and checked out and realize I have made a mistake?	Call your supervisor and explain what happened.
the client does not have a touch-tone phone, refuses to let me use the phone, or the phone is out of order?	Call your supervisor and explain what happened.





## **10.4** ACTIVITY CODES REQUIRED FOR SPECIFIED SERVICES

The table below lists the services (procedures) by Waiver for which an activity code must be documented for the claim to be complete. The Activity Code lists are in Appendix A.2 of this manual.

Waiver	Service Code	Name
Frail Elderly	HCFES5125	Level 2 Attendant Care
Frail Elderly	HCFES5125UD	Self-Directed Attendant Care
Frail Elderly	HCFES5130	Level 1 Attendant Care
MR/DD	HCDDT1019	Personal Assistance Services
MR/DD	HCDDT1000	LPN Specialized Medical Care
MR/DD	HCDDT1000TD	RN Specialized Medical Care
Physical Disability	HCPDS5126U6	Personal Services
Traumatic Brain Injury	HCTBIS5126UB	Personal Services
Technology Assisted	HCTAT1019	Personal Service Attendant





# Chapter 11 GPS ENABLED MOBILE DEVICE

User instructions to be added





## Chapter 12 MANAGING CLAIMS

A claim contains all of the information required for submission to KMAP/MMIS via a HIPAA compliant electronic billing file (837). Every service captured by the IVR, mobile device or entered via the web automatically creates a claim. Each claim within KS AuthentiCare is assigned a unique claim number which can be fully tracked in the system.

Providers must confirm each claim before KS AuthentiCare can submit it to KMAP/MMIS for payment. This involves reviewing each claim for accuracy and approving the claim for billing. Only confirmed claims are exported to KMAP/MMIS to be adjudicated for payment. Unconfirmed claims remain in KS AuthentiCare until they are confirmed or deleted.

Claims can be confirmed one at a time (see Section 12.6) or in bulk (see Section 12.7). Bulk confirmation is a function that you initiate on the website and then it is performed by the system after hours.

KS AuthentiCare exports claims to KMAP/MMIS for adjudication in the early morning hours Monday through Friday of each week. Claims confirmed by 11:59 PM the night before will be included in that morning's file submission.

If it is determined that there is an error in the claim that requires a correction, you can make those corrections using the procedures outlined in Section 12.2.

#### 12.1 SEARCHING AND VIEWING CLAIMS

- 1. Click the **Claim** radio button in the Claims section of the *Home* page.
- 2. Enter search criteria in any of these fields, if desired. You can also choose to filter claims and confirm billing by date or user name (log in) as shown at the bottom of the screen. If you do not enter any search criteria, all claims will be listed.
- 3. Click Go!





Claims	
Add New > Add New >	
	⊙ Claim
Search Type:	○ Confirm Billing - View
	O Confirm Billing - Bulk
Claim Group ID	
Claim ID:	
Claim Status:	×
Claim Start	*
Claim End	*
Service	
Authorization ID	
Client:	
Provider:	
Worker:	
CaseManager	
Representative	
Procedure Code:	
Travel Time	
Mileage:	
User Option	▼
	Go! Clear

The *Claims* page displays with the results of the search up a maximum of 300 claims.

Claims						
Cidims						
Group ID	ID	Status	Client ID	Client Name	Date Range	
725	906	InfoExceptions	111111111	Aniston, Jennifer	03/29/2010-03/29/2010	2
728	909	ConfirmBillingForClaim	111111111	Aniston, Jennifer	03/30/2010-03/30/2010	3
733	917	InfoExceptions	111111111	Aniston, Jennifer	03/31/2010-03/31/2010	3
<u>734</u>	918	InfoExceptions	111111111	Aniston, Jennifer	03/31/2010-03/31/2010	3
<u>655</u>	754	InfoExceptions	111111111	Aniston, Jennifer	03/09/2010-03/09/2010	3
<u>724</u>	905	InfoExceptions	111111111	Aniston, Jennifer	03/30/2010-03/30/2010	3
<u>744</u>	935	Authorize, Calculate, ConfirmBillingForClaim	033006586	BOSSE, HARRY A	03/30/2010-03/30/2010	3
<u>855</u>	1093	ClientEligibility, ConfirmBillingForClaim	026352675	Brocklan, Bradley	07/02/2010-07/02/2010	3
855	1092	${\sf ClientEligibility}, {\sf ConfirmBillingForClaim}, {\sf ProviderEligibility}$	026352675	Brocklan, Bradley	07/10/2010-07/10/2010	3

4. Note the columns displayed in the search results:





- **Group ID** Identifies a claim. Each claim is assigned to a claim group ID as well as a claim ID. Individual claims may be entered into the system as part of a group which is why there is a group ID and a claim ID. If the claim is not part of the group, then the group ID and claim ID will be the same.
- ID Identifies a claim
- o Status
  - Displays as *NoExceptions* or *InfoExceptions* if the claim has been sent to KMAP/MMIS or is ready to be sent.

#### OR

- Displays the name of the critical exception (if there are any). The critical exceptions are: Authorize, Calculate, AuthExhastedBefore, AuthExhaustedOn, IneligibleWorker, IneligibleProvider, DuplicateClaim, ConfirmBillingForClaim, MissingActivityCode and UnenrolledProviderServices. You could see any combination of these critical exceptions listed under status.
- **Client ID** Identifies the client who received services
- Client Name Identifies the client who received services.
- **Date Range** The date or dates of the service. This indicates if a claim spanned more than one day.
- 5. Claims are automatically listed alphabetically by client last name. Click the column heading if you wish for the search results to sort using a different column than the default. Click the heading once to change the sort to descending order for that column. Click the heading again to change it to ascending order for that column.
- 6. Position the cursor over the **Information icon** to display an additional Information pop-up about the claim.





Claims								
Group ID	ID	Status	additional to fermion	N				
<u>725</u>	906	InfoExceptions	Additional Information Group ID: 725					
<u>728</u>	909	ConfirmBillingForClaim	Client: Aniston, Jennifer (11111111) Provider: Brangelina Inco (953) Worker: Pitt, Brad (48315) Service: Respite In-Home (T1005) Service Type: Time Based					
<u>733</u>	917	InfoExceptions						
<u>734</u>	918	InfoExceptions						
<u>655</u>	754	InfoExceptions	11111111		03/03/2010-03/03/2010			
<u>724</u>	905	InfoExceptions	111111111	Aniston, Jennifer	03/30/2010-03/30/2010	2		
						-		

Claims						
Group ID	ID	Status	Client ID	Client Name	Date Range	
725	906	InfoExceptions	111111111	Aniston, Jennifer	03/29/2010-03/29/2010	2
<u>728</u>	903	ConfirmBillingForClaim	111111111	Aniston, Jennifer	03/30/2010-03/30/2010	3
733	917	InfoExceptions	111111111	Aniston, Jennifer	03/31/2010-03/31/2010	6
<u>734</u>	918	InfoExceptions	111111111	Aniston, Jennifer	03/31/2010-03/31/2010	1
<u>655</u>	754	InfoExceptions	111111111	Aniston, Jennifer	03/09/2010-03/09/2010	8
<u>724</u>	905	InfoExceptions	111111111	Aniston, Jennifer	03/30/2010-03/30/2010	3
<u>744</u>	935	Authorize, Calculate, ConfirmBillingForClaim	033006586	BOSSE, HARRY A	03/30/2010-03/30/2010	1
<u>855</u>	1093	ClientEligibility, ConfirmBillingForClaim	026352675	Brocklan, Bradley	07/02/2010-07/02/2010	8
<u>855</u>	1092	ClientEligibility, ConfirmBillingForClaim, ProviderEligibility	026352675	Brocklan, Bradley	07/10/2010-07/10/2010	8

7. Click the **Group ID** hyperlink for the claim you wish to view.

The *Claim* page displays and includes the Client, Provider, Worker, Service, Check in date and time, Check out date and time and the time Amount (if required). The box on the right side notes whether the claim was created via the IVR, mobile device or web and identifies the calculated amount which takes the total time after applying rounding rules and computes the dollar amount using the rate identified on the authorization. This is the amount that KS AuthentiCare submits to KMAP/MMIS for payment. When the claim is adjudicated by KMAP/MMIS, the actual amount paid may be different.

* Client McLain Ward 🛛 🗟 🔂	Provider Sooner He	ealthcare 🗟	Worker Jennifer Alfand	» 🗟	Group ID: 427 Filed On: IVR
* Service Date Respite In-Home 🗟 12/29	Time /2009 V 08:53 AM Mileage: Travel Time:	Amount Date 00:23 ##:## 12/29/20 Total Lines:		Click here 1 more service(s)	Printer Friendly Show All Claims Total Claims: 1 Total Calculated Amount: \$10.00 Total Authorized Amount:
Critical Exceptions		Delete	Add Lines Above   Add Lin	nes Below   Move Up   Move Down	\$10.00 Total Hours: 00:23
Note:		Add Note			Billing Confirmed





Z	As specified by KDOA and SRS, KS AuthentiCare follows normal rounding rules for billable units of service. Normal rounding causes 0.5 or half a unit to always round up.					
	<i>For Example:</i> For a 15 minute increment time-based service, 7 minutes of acutal time rounds down to 0 minutes and 8 minutes of actual time rounds up to 15 minutes. So if a service is performed for 53 minutes (15+15+15+8,) then we round up to 60 minutes or 4 units of service.					
	KS AuthentiCare calculates based on minutes, not seconds.					
	Rounding Rule for Adult Day Care:					
	• If the length of the Adult Day Care is 0-59 minutes, the claim calculates 0 units.					
	<ul> <li>If the length of the Adult Day Care is 1-5 hours (60 – 300 minutes), the claim calculates 1 unit.</li> </ul>					
	• If the length of the Adult Care is more than 301 minutes in a 24 hour period from check in, the claim calculates 2 units.					

- 8. Click on the Entity icon adjacent to the Client, Provider, Worker or Service if you wish to view the respective record for the entity.
- 9. Note if there are any **Exceptions**.

* Client McLain Ward 🛛 🗟 🔂	Provider Sooner Healthcare 💁	Worker Jennifer Alfano	<b>6</b>	Group ID:         427           Filed On:         IVR
* Service Date Respite In-Home 2 12/29/2009	08:53 AM 00:23 ##:## 1     Mileage:	ate Time 2/29/2009 ♥ 09:16 AM	Click here 1 more service(s)	Printer Friendly Show All Claims Total Claims: 1 Total Calculated Amount:
Critical Exceptions		Lines: 1 Total Claims: 1 Total Amount ete   Add Lines Above   Add Lines	: \$10.00 Total Authorized: \$10.00 Below   Move Up   Move Down	Total Authorized Amount: \$10.00 Total Authorized Amount: \$10.00 Total Hours: 00:23
Note:	Add Note			Billing Confirmed Save Delete All Cancel

10. Position the cursor over the **Exceptions icon** to view the pop-up that displays the exceptions for the claim, if there are any.

When business rules are not met, a claim is marked with an exception. Exceptions are classified as Critical or Info.

November 14, 2011





#### Critical Exception example:

* Client McLain Ward 🛛 🔯	÷	Provide Sooner	r Healthcare 🗟		Jer		
* <b>Service</b> Respite In-Home 💁	Date 12/29/2009 ¥	<b>Time</b> 08:53 AM	<b>Amount</b> 00:23	Date ##:## 12/29/2009	Ti		
Position Cursor over icon to display list of exceptions here.       Mileage:         Travel Time:       Total Lines: 1 Total Clair							
Critical Exception	s			Delete   Add Li	nes Abo		
McLain Ward       • Billing has not been confirmed for this claim.         Note       • This claim check in number does not match the authorized number for the client.         • This claim check out number does not match the authorized number for the client.							
	<ul> <li>This claim cheo number for the</li> <li>This claim cheo</li> </ul>	ck in number do: e client. ck out number d	es not match t	he authorized			

Claims with Critical Exceptions cannot be submitted to KMAP/MMIS for payment until the identified problem has been corrected. Critical Exceptions are listed in Appendix A.3.

Info Exception example:

* Client Chris Kappler	Provider Sooner Healthcare 🔄				
* Service	Date	Time	Amount	Date	Time
Respite In-Home 💁	12/29/2009	07:19 AM	00:20 <b>##:#</b>	# 12/29/2009	07:39
1			Tra	Mileage: ivel Time:	
				Total Lines:	1 Total C
Info Exceptions					
Not No n	<ul> <li>This claim claim claim claim</li> </ul>	the client. neck out numbei	does not matc	nt. h the authorized tch the authorized	
Audit Data					

*Info Exceptions* or non-critical exceptions do not prevent a claim from being processed, but serves as a notice of some problem associated with the claim creation which may warrant further investigation (e.g. a check-in from a phone number not associated with the client). These exceptions are informational only and are listed in Appendix A.3



Z



If a worker checks in more than 60 minutes prior to the start time for the scheduled event, the claim indicates an Event Matching Info Exception. For example, if the event was scheduled from 11:00 AM to 11:30 AM and the worker checked in at 9:59 AM, KS AuthentiCare is not able to match the scheduled event to the actual event and an Info Exception results.

11. Click **Cancel** to return to the *Claims* page which lists all of the search results.

Click here 1 more service(s)	Total Claims: 1
	Total Calculated Amount: \$10.00
Total Lines: 1 Total Claims: 1 Total Amount: \$10.00 Total Authorized: \$10.00	Total Authorized Amount: \$10.00
Delete   Add Lines Above   Add Lines Below   Move Up   Move Down	Total Hours: 00:23
	Billing Confirmed
	Save
	Delete All
-	Cancel

#### Additional Functionality on this page:

Click **Printer Friendly** to view the claim in a format that prints well.

Provider Sooner H	ealthcare	2	Worker Jennifer Alfano	4			oup ID: ed On:	427 IVR
	Amount 00:23	Date ##:## 12/29/2009	Time 09:16 AM tal Claims: 1 Total Amou	Click here 1	more service(s) Authorized: \$10.00	Sh To To	inter Friendly ow All Claims tal Claims: tal Calculated tal Authorized	1 Amount: <b>\$10.00</b>
	Add Note		Lines Above   Add Lin	es Below   Move	Up   Move Down		tal Hours: Billing Cou Save Delete J Cance	All





The Claim Acknowledgment page displays.

Prin
Worker Mouse, Mickey (71979)
its Amount
tualNormal 4 ActualNormal \$14.60 horizedNormat AuthorizedNormat14.60
Mouse, Mickey (7 its An

Print the page using **Print**.

12. Click **Done** to return to the *Home* page.

## 12.2 EDITING A CLAIM (CLAIMS CORRECTION)

In most situations, claims are created by workers calling from the client's home. Workers may forget to check in or check out when arriving at or leaving a client's home. They may choose the wrong service in error and fail to correct it while on the phone. In such situations, the provider is able to edit the claim by completing or correcting it.

1. Search for the claim you wish to view according to the instructions in Section 12.1.

The Claim page displays.

* Client McLain Ward	9	Provide Sooner I	r Healthcare 🖁	8		Worker Jennifer Alfano	25	Group ID: Filed On:	427 IVR
* Service Respite In-Home	Date 12/29/2009	Time 08:53 AM Mileage:	Amount 00:23		Date 12/29/2009	Time 09:16 AM	Click here 1 more service(s)	Printer Friendly Show All Claims	
Critical Exceptions		ivel Time:					nt: \$10.00 Total Authorized: \$10.00 Is Below   Move Up   Move Down	Total Claims: Total Calculated Total Authorized	\$10.00
Note:			Add Note	2				Billing Con Save Delete A Cancel	firmed





- 2. Edit the information about the claim as necessary.
- 3. Click **Save** to save your changes OR click **Cancel** to cancel your changes and return to the *Claims* page.

The *Claim Acknowledgement* page displays with a successful save message at the top, if you clicked **Save**.

eate   Reports   Scheduling   Dashboards	Visits   Administ	ration   My	Account   Cu	stom Links   L	ogout	Logged in as	: Suzie@ge
r sa ved ClaimGroup (ID: <u>3931</u> )							
							Pri
Kansas	Provider	Claim Gro Novemb	nowledgement oup ID: 3931 er 10, 2011		Worker		
Mouse, Minie Z (22222222222)			462880000A)		Mouse, Mic	:key (7 1 97 9)	
Claim ID Service	Authorizatio		End	Rates	Units	Amount	
3931 Personal Assistant Services (HCI	DDT1019) 415	Nov 02, 2 09:00:00	011Nov 02, 20 AM 10:00:00 /	AM NormaB.65	500 <sup>ActualNormal</sup> AuthorizedNorm	4 ActualNormal na4 AuthorizedNorm	\$14.60 na\$14.60
Exceptions Critical Billing has not been confirmed Unenrolled Provider Service E: Informational	forthis claim. «ception						

## 12.3 ADDING AN INDIVIDUAL (STANDARD) CLAIM

There are situations where the provider may need to add a claim using the web. For example:

- The worker was unable to use the IVR from the client's home (phone not working, client refused)
- The worker forgot to use the IVR
- The service was not provided in the client's home (such as Adult Day Care)
  - 1. Click Create in the menu bar and select "New Claim".

OR

1. Click **Claim (Standard)** adjacent to "Add New >" in the Claims section of the *Home* page.

November 14, 2011





Claims	
Add New > Claim (Standard) Add New > Claim (Express)	

The *Claim* page displays. It is pre-populated with the name of the Provider of the user currently logged in.

Kansas		nsas AuthentiCare	
Home   Create   Reports   Scho	eduling   Dashboards   Visits   Adminis	tration   My Account   Custom Links   Logout	Logged in as: acr_adm
* Client	* Provider	* Worker	Show All Claims Total Claims: 1
* Service			Total Calculated Amount: \$0.00 Total Authorized Amount:
		Click here 1 ervice(s)	\$0.00
	Total Lines: 1	Total Claims: 1 Total Amount: \$0.00 Total Authorized: \$0.00	Total Units: 0
	Delete Add	1 Lines Above   Add Lines Below   Move Up   Move Down	Save
			Delete All
			Cancel

2. Enter the **Client** ID, full name or partial last name and click the **Looking Glass icon** is to find the client. Then select the client from the list provided.

Provider ard (123456789) oner Healthcare 🗟 rtney (963852741)	Worker	Show All Claims Total Claims: 1
		Total Calculated Amount: \$0.00 Total Authorized Amount:
	Click here 1 more service(s) Total Lines: 1 Total Claims: 1 Total Amount: \$0.00 Total Authorized: \$0.00	\$0.00 Save
	Delete   Add Lines Above   Add Lines Below   Move Up   Move Down	Delete All Cancel

3. Enter the **Service** ID, full name or partial name and click the **Looking Glass icon** for the service. Then select the service from the list provided.

* Client McLain Ward	<b>a</b>	Provider Sooner Healthcare 💁	
* Service re		ortive Restorative (T1019TF) te 8 Plus Hours (S9125)	Circle Total Lines: 1 Total Claims: 1 Total Amounts 0
			Delete   Add Lines Above   Add Lines Below

The following fields display once the service is selected if the service is time based: Date, Time, and Amount.

```
November 14, 2011
```





* Client McLain Ward	•	Provide Sooner	er Healthcare 💁		Worker	
* Service Respite In-Home	Date	Time	Amount	Date ##:##	Time	]
		Mileage: Travel Time:				Click here 1 more service(s)
						Amount: \$0.00 Total Authorized: \$0.00 Lines Below   Move Up   Move Down

#### OR

The following fields display once the service is selected if the service is unit based: Date and Amount.

* Client McLain Ward 🛛 🗟 🚭	Provider Sooner Healthcare 🔄	Worker	Show All Claims Total Claims: 1
* Service Da Skilled Nursing A-E	te Amount     Image: Optimized state		Total Calculated Amount: \$0.00
	Mileage:	Click here 1 more service(s)	Total Authorized Amount: \$0.00
	Travel Time:	Total Lines: 1 Total Claims: 1 Total Amount: \$0.00 Total Authorized: \$0.00	Save Delete All
		Delete   Add Lines Above   Add Lines Below   Move Up   Move Down	Cancel

J.	If the service requires t field displays so that yo				
	* Client	Provider		Worker	
	Paul McCartney 🔄 💁	Sooner Healthcare 🗟		Betsy Jones 🛛 💁	
	* Service Date	Time Amount	Date	Time	
	Personal Care 📓 3/4/2010 💌	11:25 AM 01:00 ##	## 3/4/2010 🖌	12:25 PM	
	Activity Codes: 11 (ex: 3)	,5,8)			
	T	Mileage: 20		Click here	1 more service(s)
		Travel Time: 30			
			Total Lines: 1 Total Cla	aims: 1 Total Amount: \$14.52	Total Authorized: \$0.00
	Critical Exceptions		Delete   Add Lines A	Above   Add Lines Below	Move Up   Move Down

4. Enter the **Worker** ID, full name or partial last name and click the **Looking Glass icon** is to find the worker who performed the service. Then select the worker from the list provided.

November 14, 2011						KS AuthentiCare User M	anual
	Sooner Healthcare 🔄 e Amount Da ##:##		Date			Jack Stedding (27036) Jennifer Alfano (40961)	
	Provider				Worker		





5. Enter the **Date**. This is the date the service was delivered if it was a unit-based service. If it was a time-based service, then this is the date the delivery of the service was started.

If the service is a time-based service, proceed to Step 6. If the service is a unit-based service, proceed to Step 8.

- 6. Enter the **Time** the delivery of the service started. You must include AM or PM in the time entry or use military time. If it is on the hour, it is not necessary to include ":00".
- 7. Enter the **Date** and **Time** the delivery of the service ended.

* Client McLain Ward	÷	Provider Begin Time	ealthcare 🗟		Worker	
* Service Respite In-Home	Date 12/28/2009	▼ 11:00 AM	Amount 01:00	Date ##:## 12/28/2009	▼ 12:00 PM	-
		Mileage: Travel Time:				Click here 1 more service(s)
						nount: \$0.00 Total Authorized: \$0.00 nes Below   Move Up   Move Down

8. Enter the Amount. This is the number of units delivered.

For time-based services the amount is computed by the system based on the start date and time and the end date and time. This step is not required.

* Client McLain Ward 🛛 🚱 🔂	Provider Sooner Healthcare 🗟	Worker Betsy Jones 🛛 🗟	Show All Claims Total Claims: 1		
* Service In-Home Respite 8 Plus Hours 🗟	Date Amount 1/23/2010 Mileage:	Click here 1 more service(s)	Total Calculated Amount: \$0.00 Total Authorized Amount: \$0.00		
	Travel Time:		Total Units: 10		
		Total Lines: 1 Total Claims: 1 Total Amount: \$0.00 Total Authorized: \$0.00	Save		
		Delete   Add Lines Above   Add Lines Below   Move Up   Move Down	Delete All Cancel		

D





9. Click **Save** if you have completed the claim and do not need to group this claim with any other claims. If you are grouping claims, then proceed to Step 12.

* Client McLain Ward	Provider Sooner Healthcare 💁	Worker Jennifer Alfano 🛛 🔯	Show All Claims Total Claims: 1
* Service Date Respite In-Home 🔄 12/28/2009		Date         Time           12/28/2009         ▼         12:00 PM           Click here         1         more service(s)	Total Calculated Amount: \$0.00 Total Authorized Amount: \$0.00
		Total Lines: 1 Total Claims: 1 Total Amount: \$0.00 Total Authorized: \$0.00 Deleta   Add Lines Above   Add Lines Below   Move Up   Move Down	Save Delete All Cancel



The amount for the claim (what displays on the right side of the screen) is not computed until the claim is saved.

The *Claim Acknowledgement* page displays.

							Prin
Kansas							
		Claim Ack	nowledgement				
		Claim Gro	oup ID: 3931				
		Novemb	er 10, 2011				
Client Mouse, Minie Z (2222222222)	Provider General H	ealth Care (·	462880000A)		Worker Mouse,	Mickey (71979)	
Claim ID Service	Authorizati		End	Rates	Units	Amount	
3931 Personal Assistant Services (HCDDT	1019) 415	Nov 02, 2 09:00:00	011Nov 02, 20 AM 10:00:00 A	11NormaB.	6500 <sup>ActualNorma</sup> AuthorizedNo	4 ActualNormal orma4 AuthorizedNor	\$14.60 rma\$14.60
Exceptions Critical Billing has not been confirmed for t Unenrolled Provider Service Excep	nis claim. Ition						
<ul> <li>Informational         <ul> <li>This claim does not have a matchir</li> </ul> </li> </ul>	gevent						
			Total Claims	1 Total Actu	ual Amount: \$14.60	Total Authorized Ar	nount \$14.
						Done	lew Claim

10. Click **Done** or **New Claim**. Click **Done** if you do not need to enter any other new claims. The *Home* page displays after clicking **Done**. Click **New Claim** if you need to enter additional claims. The *Claim* page displays.



November 14, 2011





#### Grouping Claims

11. Click Add Lines Below.

* Client McLain Ward 🛛 🔯	Provider Sooner Healthcare 🗟	Worker Betsy Jones	Show All Claims Total Claims: 1
* Service Date Respite In-Home 2 12/28/2	Time         Amount           1009         11:00 AM         00:30         ##:           Mileage:	Date         Time           ## 12/28/2009         11:30 AM           Click here 1 more service(s)	Total Calculated Amount: \$0.00 Total Authorized Amount: \$0.00
	Travel Time:	Total Lines: 1 Total Claims: 1 Total vnount: \$0.00 Total Authorized: \$0.00 Delete   Add Lines Above Add Lines Below Hove Up   Move Down	Save Delete All Cancel

A new claim opens below the claim you just finished.

- 12. Enter the information for this claim. The claim pre-populates with the same client, provider, and worker of the claim above it; however, this information can be changed if needed. All of the claims in the group do not have to be for the same client and worker. Grouping claims provides a mechanism to view a number of claims on the same screen. Claims cannot be grouped using the IVR or mobile device; they can only be grouped using the web interface.
- 13. Return to Step 10 to proceed with saving all of the claims in the group.

#### 12.4 ADDING MULTIPLE CLAIMS (EXPRESS ENTRY)

There are situations where the provider may need to add a claim using the web for the same client-worker-service combination.

1. Click **Claim (Express)** adjacent to "Add New >" in the Claims section of the *Home* page.

November 14, 2011





Claims	
	<u>Claim (Standard)</u> <u>Claim (Express)</u>

The *Claim* page displays. It is pre-populated with the name of the Provider of the user currently logged in.

* Provider Enid Wo	rker's Co. 🗟		
* Client		* Worker	
* Service			

- 2. Enter the **Client**, **Worker and Service** as for a single claim entry and click **Next**.
- 3. A list with 10 blank rows where you can enter date, time and activity codes will appear.

Provider: Gene	eral Health Care			int Account 1 o	Custom Links   Logout	Logged in as: acr_admi
Client Mou	se, Minie Z		Worken Mouse,	Mickeey	Service: Personal Assist	ant Services
d Rows						
Start Date	Start Time	Duration	End Date	End Time	Activity Codes Note	8
Start Date	Start Time	Duration	End Date	End Time	Activity Codes Note (ex: 3,5,8)	R
Start Date	Start Time	Duration	End Date	End Time	Activity Codes Note	
	Client Mous I Rows Start Date Start Date	Client Mouse, Minie Z I Rows Start Date Start Time Start Date Start Time V	Start Date Start Time Duration	Client Mouse, Minie Z Workert Mouse, I Rowe Start Date Start Time Duration End Date Start Date Start Time Duration End Date Start Date Start Time Duration End Date	Client Mouse, Minie Z Worker: Mouse, Mickeey IRowe Start Date Start Time Duration End Date End Time Start Date Start Time Duration End Date End Time Http://www.internet/inter	Client Mouse, Minie Z Worker: Mouse, Mickeey Service: Personal Assist

4. There is an **Add Rows** button on each screen that allows you to add as many rows as needed.

November 14, 2011





- 5. When you have entered all the data,
  - a. If you click on the **Save and Continue** button, it will save the claims and present the first **Express Claim** screen where you can enter another client, worker and service combination.
  - b. If you click on the **Save and Exit** button, you will return to the **Home** page.

#### 12.5 DELETING A CLAIM

If a claim was added in error, then it should be either edited or deleted. Once the claim is confirmed for billing, it cannot be deleted.

1. Click **Delete All** on the *Claim* page.

* Client McLain Ward 🛛 🔯 🔂	<b>Provider</b> Sooner Health	are 🗟	Worker Jennifer Alfano 🛛 🗟	4	Show All Claims Total Claims: 1
* Service Date Respite In-Home 🔄 12/28/200		ount         Date           00         ##:##         12/28/2009	Time 12:00 PM		Total Calculated Amount: \$0.00 Total Authorized Amount:
	Mileage: Travel Time:		Clic	ck here 1 more service(s)	\$0.00 Save
				t: \$0.00 Total Authorized: \$0.00	Delete All Cancel

The system asks you to confirm the deletion. If you click **OK**, the claim is permanently deleted from the system. If the claim is a group of claims, they will all be permanently deleted. If you click **Cancel**, the claim is not deleted and you are returned to the *Claim* page.



2. Click **OK** to proceed with permanently deleting the claim.

You are returned to the *Home* page which displays a message in the upper left hand corner that the claim was deleted successfully.







## 12.6 CONFIRMING A SINGLE CLAIM FOR BILLING

The function of confirming a single claim can take place in two areas of KS AuthentiCare.

#### Option 1

- 1. Select **Confirm Billing** as the Search Type in the *Claims* section of the *Home* page.
- 2. There is a **Sort By** selection at the bottom of the screen. The default is to display the claims to be confirmed alphabetically by Client's Last Name. However you can also sort by Worker's Last Name, Date of Service, Claim ID, Client ID or Worker ID.
- 3. Click Go!

Claims	
Add New > Add New >	<u>Claim (Standard)</u> <u>Claim (Express)</u>
	O Claim
Search Type:	⊙ Confirm Billing - View
	O Confirm Billing - Bulk
Claim Group ID	
Claim ID:	
Claim Start:	✓
Claim End:	×
Service:	
Authorization ID	
Client:	
Provider:	
Worker:	
CaseManager:	
Representative:	
Procedure Code:	
Travel Time	
Mileage:	
User Option:	
* Sort By:	Member's Last Name 🛛 💌
-	Go! Liear

November 14, 2011





The *Confirm Billing* page displays a maximum of 300 claims not yet confirmed for billing.

Kansas						
Tanoas						
		Confirm Bi	lling			
		November 12	2, 2011			
teck All/Uncheck All						
Approve Silling for Claim (1129)						
Client Mouse, Minie 2 (2222222222)	Provider ARROW?	r HEAD WEST INC	(1000273994	0	Worker Mouse, Micks	tey (\$1629)
Claim ID Service	Authorizatio	n Stert	End	Rates	Unita	Amount
1129 - Personal Assistant Services (HCDOT1019	) 416	Oct 17, 2011 01:00:00 AM	OH:17, 2011 01:15:00 AM	Normal3.6300	ActualNormal 1 AuthorizedNormat	ActualNormal \$3.63 AuthorizedNorma\$3.63
Exceptions Critical o Silling has not been confirmed for this	ı daim.					
<ul> <li>Informational         <ul> <li>This claim does not have a matching of</li> </ul> </li> </ul>	event.					
Approve Billing for Claim (1277)						
Client Mouse, Minie Z (2222222222)	Provider ARROWN	HEAD WEST INC	(1000273994	0	Worker Mouse, Micks	τογ (\$1629)
Claim 10 Service	Authorizatio	n Start	End	Retes	Unita	Amount
1277 Personal Assistant Services (HCDDT1019	) 415	Nov 06, 2011 06:30:00 AM	Nev 06, 2011 06:45:00 AM	Normal3.6300	ActualNormal AuthorizadNormal	ActualNormal \$3.83 AuthorizedNorma\$3.83
Exceptions Critical o Billing has not been confirmed for this	ı daim.					
<ul> <li>Informational         <ul> <li>This doin does not have a matching a</li> <li></li></ul></li></ul>	event.					
Approve Billing for Claim (3933)						
Client Mouse, Minie 2 (2222222222)	Provider ARROWN	, HEAD WEST INC	(1000273994	0	Worker Nouse, Micks	tey (\$1629)

4. Click **Check All/Uncheck All** at the top of the page to select all displayed claims for confirmation.

OR





3. Click the **checkbox** adjacent to the claim you wish to confirm.

Confirm Elling November 12, 2011
Approve Silling for Claim (1129)     Claim Moves, Minit 2, 200227309A, Moves, Minit 2, 20027309A, Moves, Minit 2, 20027300, Moves, Minit 2, 20027300, M
Approve Silling for Claim (1277)      Client Mouse, Minie 2 (20000000000000000000000000000000000

4. Click **Confirm Billing** to confirm all selected claims.



The *Home* page displays with a successfully confirmed billing message.





D





## Option 2:

1. Search for the Claim you wish to view according to the instructions in Section 12.1.

The *Claim* page displays.

Claim Search Results				
* Client Mouse, Minie Z 💁 🔂	* Provider ARROWHEAD WEST INC	* Worker Mouse, Mickeey		Group ID: <b>1277</b> Filed On: <b>Web</b>
* Service Personal Assistant Services 🌄		Amount Date	<b>Time</b> 06:45 AM	Printer Friendly Show All Claims
Critical Exceptions		Clic al Lines: 1 Total Claims: 1 Total Amount ete   Add Lines Above   Add Lines Be	· · · · · · · · · · · · · · · · · · ·	Total Claims: 1 Total Calculated Amount: \$3.63 Total Authorized Amount: \$3.63
Note:	Add Note			Total Units: 1 Total Hours: 00:15 Billing Confirmed
Note Data	×			Delete All Cancel

2. Click the **Billing Confirm** check box and click **Save**.

Claim Search Results					
* Client Mouse, Minie Z 🛛 🙆 🔂	* <b>Provider</b> ARROWHEAD WEST INC 💁	* <b>Worker</b> Mouse, Mickeey		Group ID: Filed On:	1277 Web
* Service Personal Assistant Services 🛃	11/6/2011 O6:30 AM 0	mount Date 10:15 ##:## 11/6/2011 Clic Lines: 1 Total Claims: 1 Total Amount:	Time           06:45 AM           ck here         1           more service(s)           : \$3.63 Total Authorized: \$3.63	Printer Friendly Show All Claims Total Claims: Total Calculated A	
Critical Exceptions	Delete	e   Add Lines Above   Add Lines Be	low   Move Up   Move Down	Total Authorized A	\$3.63 Amount: \$3.63
Note:	Add Note			Total Units: Total Hours: Billing Conf Save	1 00:15 irmed
Note Data	×			Delete Al Cancel	





The *Claim Acknowledgement* page displays with a note that your confirmation saved successfully.

A STATE OF A						
T7						
Kansas						
<b>I</b> MIIBUB						
		Claim Ack	nowledgement			
		Claim Gro	oup ID: 1277			
		Novemb	er 12, 2011			
Client	Provider				Worker	
Mouse, Minie Z (2222222222)		WEST IN	C(10002739	9A)		keey (51629)
Claim ID Service	Authorization		End	Rates	Units	Amount
1277 Personal Assistant Services (HC	CDDT1019) 416	Nov 06, 2 06:30:00	2011Nov 06, 2 AM 06:45:00	011 AMNormal3.63	00ActualNormal AuthorizedNorm	1 ActualNorm al Authorized
Exceptions						
<ul> <li>Informational         <ul> <li>This claim does not have a magnetic does</li></ul></li></ul>	atching event.					
	-					

3. Click **Done** and the *Home* page displays.

## 12.7 CONFIRMING CLAIMS IN BULK

Providers have the option to choose a group of claims to be automatically confirmed by the system after hours but prior to the next submission of claims to KMAP/MMIS. Put in a start and end date for the Date of Service (DOS) to indicate the claims for that date range are to be confirmed in bulk. All claims filed for that date range will be chosen unless you chose a specific Filing Source (Web or IVR). When the bulk confirmation process runs, it will look at the services that are ready for confirmation and confirm those that do not have critical exceptions.

Claims			
Add New > <u>Claim (Standard)</u> Add New > <u>Claim (Express)</u>			
O Claim			
Search Type: O Confirm Billing - View			
• Confirm Billing - Bulk			
Claim Start:	×		
Claim End:	×		
Filing Source	Both		
-	Go! Clear		
	Both Web		
	IVR		





## 12.8 UNCONFIRMING CLAIMS FOR FURTHER EDITING

Occasionally you will need to edit a claim that has been confirmed <u>but not yet submitted</u> to KMAP/MMIS for adjudication. In order to edit it, you must first unconfirm it.

- 1. Open the individual claim.
- 2. Uncheck the **Billing Confirmation** box
- 3. **Save** the claim

The claim can now be edited or even deleted. It must be confirmed again before it can be exported to KMAP/MMIS for adjudication.




# Chapter 13 REPORTING

KS AuthentiCare includes robust reporting capabilities to assist providers and state staff in managing and monitoring clients, workers, schedules and claims. Reports are available 24/7 via the web and information is current as of the time a report is created.

A variety of sort and filter criteria are available to create unique reports reflecting the specific information needed. A user may filter information to produce a report which displays information related to a particular client, worker or service and within date ranges chosen by the user. The information may be sorted to display in an order that is most convenient for the user. The sort and filter options for each report are discussed more fully later in this chapter.

As discussed in earlier chapters, the provider agency/FMS provider can only create reports with information related to the specific provider location (unique Medicaid ID + letter). KDOA staff can create reports on any information related to KDOA clients. SRS staff can create reports on any information related to SRS clients.

# 13.1 CREATING A REPORT

1. Click **Reports** on the Main Menu.



The *Report* page displays.





Report Templates [Delete Selected Templates]	Vie	w Reports	[Refres	h] [Delete Selecte	d Reports]
		Name		Submit Time	<u>Status</u>
Create Reports		Eligible Client Dat	a Listing Report	5/31/2011 10:39 AM	Complet
Authorizations					
AuthentiCare Service Authorizations					
Claim Details					
AuthentiCare Claim Details					
Claim History					
AuthentiCare Claim History					
Calendar					
Scheduled AuthentiCare Calendar Events					
Late and Missed Visits					
Late and Missed Visits for Scheduled AuthentiCare Events					
Worker By Provider					
Worker By Provider Report					
Provider Activity					
Provider Activity Report					
Billing Invoice					
Billing Invoice Report					
Time and Attendance					
Time and Attendance Report					
Exception					
Exception Report					
Overlapped Claim					
Overlapped Claim Report					
Unauthorized Phone Number					
Unauthorized Phone Number Report					
Remittance Advice					
Remittance Advice Report					
Claim Data Listing					
Claim Data Listing Report					
Remittance Data Listing					
Remittance Data Listing Report					
Eligible Client Data Listing					
Eligible Client Data Listing Report					

There are three sections of the Report Page:

- 1. **Report Templates** Users can create templates for reports that are created on a regular basis. For example, there is a need for a report on Late and Missed Visits at the end of each month for all clients, a template can be created with the desired settings. Templates are addressed in more detail in Section 13.4.
- 2. **Create Reports** Seventeen types of reports are available for creation and each can be filtered and sorted to create a unique report to fit the user's needs. Each report name is a hyperlink that allows you to enter your filter and sort criteria and run the report. Proceed to Step 2 for further instructions.
- 3. View Reports Once a report is generated, it appears in the View Reports section of the page. Reports can be saved to the user's local drive for permanent storage and retrieval. If a report is needed at a later date and has been deleted, it can simply be rerun for the same dates. Methods of viewing reports are addressed in detail in Section 13.2.





2. Click on a report name hyperlink from the list provided in the **Create Reports** section of the *Report* page.

Report Templates [Delete Selected Templates]	View Reports	[Refresh]	[Delete Selecte	d Reports]
	□ <u>Name</u>		Submit Time	<u>Status</u>
Create Reports	Eligible Client Data Listin	g Report	5/31/2011 10:39 AM	Complete
Authorizations				
AuthentiCare Service Authorizations				
Claim Details				
AuthentiCare Claim Details				
Claim History				
AuthentiCare Claim History				
Calendar				
Scheduled AuthentiCare Calendar Events				
Late and Missed Visits				
Late and Missed Visits for Scheduled AuthentiCare Events				
Worker By Provider				
Worker By Provider Report				
Provider Activity				
Provider Activity Report				
Billing Invoice				
Billing Invoice Report				
Time and Attendance				
Time and Attendance Report				
Exception				
Exception Report				
Overlapped Claim				
Overlapped Claim Report				
Unauthorized Phone Number				
Unauthorized Phone Number Report				
Remittance Advice				
Remittance Advice Report				
<u>Claim Data Listing</u>				
Claim Data Listing Report				
Remittance Data Listing				
Remittance Data Listing Report				
Eligible Client Data Listing				
Eligible Client Data Listing Report				





The *Authorizations Report* page was chosen for this example. It displays the filter and sort criteria for the report which are unique for the chosen report. This screen varies depending on the type of report chosen in Step 2, refer to Section 13.5 for specific filter and sort information for each type of report.

Authorizations Report * Indicates a required field.
* Report Name: Authorizations Roport
Description:
<u>e</u>
Include Claim Details
* Effective Dates:
Client:
Provider: Oklahama Hama Haalth Provide
Worker:
Case Manager:
Service:
Team Assignment:
Program:
At Rink:
Texonomy Code:
NPI
Disgnosis
External Worker ID
Served by another person
living in the home:
Sort 1:
Sort 2:
Sort 3:
ReportType(x): 2707 Exccl CSV XML
Seve As Templete Run Report Cancel

- 3. Enter a **Report Name**. This automatically defaults to the name of the report selected, but this name should be changed to something more descriptive. For example, if the report is for a single client, the report name can be changed to include the client's name and the date range.
- 4. Enter a **Description**, if desired. This is most helpful in creating templates which are addressed in more detail in Section 13.4.

November 14, 2011





 Choose a date range from the drop-down box for the **Dates** field. This field appears on this page for all of the reports, though it may appear as **Claim Dates** or **Effective Dates** depending on the report selected.



- 6. Enter any other filter criteria desired such as **Client**, **Worker**, **Service** or **Provider**. These criteria are similar for all reports. Entering one of these or a combination of these creates a unique report. If no information is added, the report includes all information for the period selected.
- 7. Select **Sort** criteria as desired. These are similar for all reports. KS AuthentiCare allows selection of up to three sort items.





As an example, the provider may choose to have the report sorted first by Client, then by Service, then by Date. In this example, the Sort fields would be populated as shown below.

Sort 1:	Client	~
Sort 2:	Service	~
Sort 3:	Effective Date Start	~

 Choose the Report Type(s) to indicate the format the report will be displayed. You can choose as many format types as need. If you do not choose, the report will automatically default to PDF <u>except</u> for the two list reports (Claims Data Listing and Remittance Data List) which default to Excel.

ReportType(s): ♥PDF □Excel □CSV □XML

November 14, 2011





#### 9. Click Run Report.

The system returns to the *Report* page. The report appears in the View Reports section of the page with a submitted time and status. The Status can be one of the following:

- Queued the report is in line for processing
- In Progress the report is being created
- Completed the report is ready for viewing



### 13.2 VIEWING A REPORT

Once the Status of the Report has changed to Completed, the report may be viewed in the format(s) selected.

Report Templates [Delete Selected Templates]	View Reports [Refresh] [Delete Selected Reports
	Name         Submit Time         Status
Create Reports	Provider Activity Report 11/9/2011 Completed
Authorizations	Claim Details Report 11/8/2011 Completed 4:44 PM
AuthentiCare Service Authorizations Claim Details AuthentiCare Claim Details	Claim History Report 11/8/2011 Completed 4:18 PM
AuthentiCare Claim Details <u>Claim History</u> AuthentiCare Claim History	Calendar Report 11/8/2011 Completed 4:18 PM

November 14, 2011





- 1. Click one of the icons under the report name to generate the report in the desired format. These icons do not appear until the status of the report is "Completed". The icons are:
  - Click this icon to open the report as an Adobe .pdf file which requires Adobe Reader to view. This format is the most convenient for printing and viewing.



Click this icon to open the report as an Excel spreadsheet.

- Click this icon to open the report as a CVS file. This format may be useful in importing the information to another spreadsheet or database.
- Click this icon to open the report as an XML file. Like the CSV file, this format may be useful in sending information to another source, such as a data warehouse.



The column headings in the View Reports section are hyperlinks that change the sort order of the reports that are displayed in this section. Click the hyperlink once to change the view to ascending order based on the values in the column chosen. Click the hyperlink again to change the view to descending order based on the values in the column chosen.

2. Click **Open** if you wish to open the report in a new window or click **Save** to save the report to a storage location such as your hard drive or a network drive. (You will not see this if you choose the Adobe .pdf option to view your report.)

File Dov	vnload 🛛 🔀					
Do you	Do you want to open or save this file?					
	Name: Calendar_Report_20100128101318.xls Type: Microsoft Office Excel 97-2003 Worksheet, 51.8KB From: ext.fdgs.com					
🗹 Alwa	Open Save Cancel					
2	While files from the Internet can be useful, some files can potentially harm your computer. If you do not trust the source, do not open or save this file. What's the risk?					





The appropriate application starts based on the format you selected and the report is displayed, if you click **Open**. At that point, you may print the report if desired. The following example is from clicking the Adobe .pdf icon.

THWEST K	ANSAS AAA (1	00258959A)					
Worker							
Any							
Client		Service		Start	End	Total Units	Authorizatio
BRUCE,	IAY (00084774411)	Level 2 Attendant Care (H	HCFES5125)	10/01/2011	1 11/30/2011	200	418
Clair	n Claim Start	Claim End	Total Units	Auth Units	Exceptions		
426	10/10/2011 10:11:00 AM	10/10/2011 11:00:00 AM	3	3	ConfirmBillingForCla atching	aim,EventM	
427	10/11/2011 2:00:00 PM	10/11/2011 3:00:00 PM	4	4	ConfirmBillingForCla atching	aim,EventM	
428	10/28/2011 7:38:00 AM		1	1	EventMatching		
430	10/28/2011 9:03:00 AM	10/28/2011 9:06:00 AM			ConfirmBillingForCla atching,MissingActiv		
431	10/28/2011 2:56:00 PM	10/28/2011 2:58:00 PM			ConfirmBillingForCla atching	aim,EventM	
432	10/28/2011 6:49:00 PM	10/28/2011 6:51:00 PM			ConfirmBillingForCla atching	aim,EventM	
433	10/31/2011 8:59:00 AM	10/31/2011 9:07:00 AM	1	1	ConfirmBillingForCla atching	aim,EventM	
438	11/2/2011 12:52:00 PM				EventMatching		
439	11/2/2011 1:19:00 PM				EventMatching		

# 13.3 DELETING A REPORT FROM THE VIEW REPORTS SECTION

1. Click the checkbox to the left of the name of the report you wish to delete.

Report Templates	[Delete Selected Templates]	Vie	ew Reports [Refres	h] [Delete Selec	ted Reports]
			Name	<u>Submit Time</u>	<u>Status</u>
Create Reports			Provider Activity Report	t 11/9/2011 12:06 PM	Completed
Authorizations			Claim Details Report	11/8/2011 4:44 PM	Completed
AuthentiCare Service Authoriz <u>Claim Details</u> AuthentiCare Claim Details	zations		Claim History Report	11/8/2011 4:18 PM	Completed
AuthentiCare Claim Details <u>Claim History</u> AuthentiCare Claim History			Calendar Report	11/8/2011 4:18 PM	Completed
			at a second second second	/ . /	





And the second s	0	box next to <b>Name</b> pop of the reports listed in				
	Vi	iew Reports	[Refresh]	[Delete Selec	ted Reports]	
	E	Name	1	Submit Time	<u>Status</u>	
	5	Events Scheduled for Jennifer Alfano	o today	1/15/2010 3:49 PM	Queued	
	٦	Authorizations Report for January		1/15/2010 3:41 PM	Completed	

2. Click Delete Selected Reports if you want to permanently remove the report.



The *Report* page displays and the report is no longer listed in the **View Reports** section.



### 13.4 USING REPORT TEMPLATES

Instead of creating the same report at the end of each week, month or quarter, providers can save time by creating a template for the report. The **Calendar Report** is being used as an example for this section.





### 13.4.1 CREATING A REPORT TEMPLATE

1. Click on a report name from the list provided in the **Create Reports** section of the *Report* page.

Home       Create       Reports       Visits       Administration       My Account       Custom Links         Report Templates       [Delete Selected Templates]       View Reports         Image: Name       Image: Name       Image: Name         Create Reports       Image: Provider Activit       Image: Name         Authorizations       Claim Details       Image: Claim Details Report         AuthentiCare Claim Details       Image: Claim History       Image: Claim Details Report         AuthentiCare Claim History       Calendar       Image: Claim Details Report
Create Reports       Image: State Stat
Create Reports       Provider Activit         Authorizations       Claim Details Report         AuthentiCare Service Authorizations       Claim Details Report         AuthentiCare Claim Details       Claim History Report         AuthentiCare Claim History       Calendar Report         Calendar       Claim Details Report
Create Reports       Provider Activit         Authorizations       Claim Details Report         AuthentiCare Service Authorizations       Claim Details Report         AuthentiCare Claim Details       Claim History Report         AuthentiCare Claim History       Calendar Report         Calendar       Claim Details Report
Authorizations       Claim Details Right         AuthentiCare Service Authorizations       Claim Details Right         Claim Details       Claim History Right         AuthentiCare Claim Details       Claim History Right         AuthentiCare Claim History       Calendar Report         AuthentiCare Claim History       Calendar Report
Authorizations       Image: Claim Authonizations         AuthentiCare Service Authorizations       Image: Claim History R         AuthentiCare Claim Details       Image: Claim History R         AuthentiCare Claim History       Image: Claim Authonization Report         AuthentiCare Claim History       Image: Claim Authonization Report         AuthentiCare Claim History       Image: Claim Details Report         Calendar       Image: Claim Details Report
Claim Details       Claim History R         AuthentiCare Claim Details       Calendar Report         AuthentiCare Claim History       Calendar Report         AuthentiCare Claim History       Calendar Report         Calendar       Claim Details Report
Claim History     Calendar Report       AuthentiCare Claim History     Claim Details Report       Calendar     Claim Details Report
Calendar Claim Details Re
Scheduled AuthentiCare Calendar Events
Late and Missed Visits
Worker By Provider Authorizations
Provider Activity
Provider Activity Report
Billing Invoice Billing Invoice Report





The *Calendar Report* page for the report displays. This page is where you enter the filter and sort criteria for the report.

	1
* Report Name: Colondo	
Description:	5
* Effective Detext	
Events	
Authorization	
Clients	9
Provider	Cklakema Home Health Provide
Primary Worker:	
Service	
	and the second se
Team Assignment:	
Program	
At Risk:	
Texonomy Code:	
NPI	
Diegnosis	
External Worker LD	
terved by another person living in the home	
Sort 1:	
Sert 2	×
Sert 3:	. N
1	
ReportType(s):	Y PDF Exect GSV XML

- 2. Enter a unique **Report Name**. This automatically defaults to the name of the report selected, but this name should be changed to something more descriptive.
- 3. Enter a **Description**. This identifies the purpose of the report.
- 4. Choose a date range from the drop-down box for the **Effective Dates**. This varies based on the type of report. Refer to Section 13.5.
- 5. Enter any other filter criteria. This varies based on the type of report. Refer to Section 13.5.
- 6. Select **Sort** criteria as desired.
- 7. Select the **Report Type(s)**.
- 8. Click Save as Template.





The *Report* page displays and the template just created is included in the **Report Templates** section.



13.4.2 RUNNING A REPORT FROM A TEMPLATE

1. Click the **Run Report icon** adjacent to the name of the template.



The system returns to the *Report* page. The report appears in the View Reports section of the page with a submitted time and status of "Queued". Proceed to Section 13.2 for further instructions on viewing the report.

Kansas	Kansas AuthentiCare			
Home   Create   Reports	Scheduling Dashboards Visits Administration	My Account   Custom Links   Logout Logged in as: a		
	Report Templates [Delete Selected Templates	View Reports [Refresh] [Delete Selected Reports]		
	Calendar Report	Name Submit Time Status		
	For Minnie Mouse	Calendar Report 11/9/2011 Queued		
	Create Reports	Provider Activity Report 11/9/2011 Completed 12:06 PM		
November 14, 2011		KS AuthentiCare User Manual		





### 13.4.3 EDITING A REPORT TEMPLATE

1. Click the hyperlink on the name of the report.



The Calendar Report page displays.

* Report Name: Colondar Ropa	21
Description:	10 10
* Effective Dates:	~
Events	
Authorization:	
Client:	9
Provider: Ckla)	lama Hama Haalth Pravida
Primary Worker:	2
Service:	8
Team Assignment:	~
Program:	N
At Risk:	N
Texonomy Code:	
NP1:	
Disgnosis;	
External Worker ID	
erved by another person living in the home:	
Sort 1:	
Sers 2	8
Sert 3:	

2. Edit the *Calendar Report* page as desired.





3. Click Save as Template.

The *Report* page displays. In this example, the **Description** was modified and it now displays under the name of the Report Template.



# 13.4.4 DELETING A REPORT TEMPLATE

1. Click the checkbox to the left of the name of the report template you wish to delete.



2. Click **Delete Selected Templates** if you want to permanently remove the report template.



The *Report* page displays and the report template is no longer listed.





# 13.5 REPORT EXAMPLES

#### 13.5.1 AUTHORIZATIONS REPORT WITHOUT CLAIM DETAIL

The Authorizations Report lists all authorizations in KS AuthentiCare for a given time period. The report can be filtered to include only authorizations for a particular client or service. It can be sorted to display the authorizations in a certain order. For example, the provider may choose to see the authorizations sorted by service then by client. The Authorizations Report without Claim Detail does not include the claims associated with the authorizations.

The Authorizations Report as displayed in the screenshot below only requires filter criteria for **Effective Dates** (Effective Dates of the authorization). The options include Current Day, Current Week, Current Month, Current Quarter, Current Year, and Fixed Date. If Fixed Date is chosen, then you must enter the "from" and "to" dates (mm/dd/yyyy). Additional filter criteria include Client, Provider, Worker, Service and Waiver.

Authorizations Report
* Indicates a required field.
* Report Name: Authorizations Ropold
Description:
Include Claim Dataila
* Effective Dates:
Client:
Provider: Oklahoma Home Health Provide
Worker:
Case Manager:
Service:
Team Assignment: N
Program: N
At Risk: N
Texonomy Code:
NPI:
Disgnosis:
External Worker ID
Served by another person living in the home:
Sort 1:
Sort 2:
Sort 3:
ReportType(s): VPDFExcclCDVXML
Seve As Templete Run Report Cencel





The Authorizations Report can also be sorted which means that the information returned on the report is grouped by whatever sort selection is made. For instance if Client is chosen, then all of the authorizations for that client are grouped together. Below is a screenshot of the Sort options available on the Authorizations Report.

Sort 1:	~	]
Sort 2: Sort 3:	Effective Date Start Provider Worker	
Save As Template	Client Case Manager Service	Cancel

An example of the Authorizations Report:

Isas	Aut	hentiCare	e Auth	noriza	tions		Ð
	ISAS AAA (1	00258959A)					
Worker Any							
Client		Service		Start	End	Total Units	Authorizatio
BRUCE, JAY	(00084774411)	Level 2 Attendant Care (H	ICFES5125)	10/01/2011	11/30/2011	200	418
Claim	Claim Start	Claim End	Total Units	Auth Units	Exceptions		
426	10/10/2011 10:11:00 AM	10/10/2011 11:00:00 AM	3	3	ConfirmBillingF atching	orClaim,EventM	
427	10/11/2011 2:00:00 PM	10/11/2011 3:00:00 PM	4	4	ConfirmBillingF atching	orClaim,EventM	
428	10/28/2011 7:38:00 AM		1	1	EventMatching		
430	10/28/2011 9:03:00 AM	10/28/2011 9:06:00 AM			ConfirmBillingF atching,Missing	orClaim,EventM ActivityCode	
431	10/28/2011 2:56:00 PM	10/28/2011 2:58:00 PM			ConfirmBillingF atching	orClaim,EventM	
432	10/28/2011 6:49:00 PM	10/28/2011 6:51:00 PM			ConfirmBillingF atching	orClaim,EventM	
433	10/31/2011 8:59:00 AM	10/31/2011 9:07:00 AM	1	1	ConfirmBillingF atching	orClaim,EventM	
438	11/2/2011 12:52:00 PM				EventMatching		
439	11/2/2011 1:19:00 PM				EventMatching		
Number of Cl	laims: 9						
Authorized U	nits Remaining: 19	91					

### 13.5.2 AUTHORIZATIONS REPORT WITH CLAIM DETAIL

The Authorizations Report with Claim Detail is the same as the Authorizations Report described in Section 13.5.1; however, in addition to the authorizations, any claims associated with those authorizations are also displayed. This report provides information on the number of units remaining in the authorization based on the number of units for which there are claims.





The filter and sort criteria are the same as the Authorizations Report without Claim Detail, but you must check the **Include Claim Details** checkbox.

Authorization * Indicates a requ		
* Report Name:	Authorizations Report	
Description:		< >
	Include Claim Details	3
* Effective Da	tes: 👻	
Clie	ent:	
Provi	der:	
Wor	ker:	

### 13.5.3 CLAIM DETAILS REPORT

The Claim Details Report lists all claims in KS AuthentiCare for the time period specified. The report can be filtered to include only certain types of claims (for example, claims that have been exported for billing) or only claims for a particular client, worker and/or service. It can also be sorted to display the claims in a specific order.

The Claim Details Report as displayed in the screenshot below has several filter criteria. **Claim Type**, **Claim Dates**, and **Group By** are all required when running the report.

Claim Details Report * Indicates a required field.	
* Indicates a required held.	
* Report Name: Claim Dotails R	
Description:	8
* Claim Type:	<u> </u>
* Claim Dates:	N
Client:	
Provider: Oklah	oma Home Health Provide
Worker:	
Case Manager:	
Service:	
* Group By:	<u></u>
Team Assignment:	N
Program:	N
At Risk:	N
Texonomy Code:	
NPI:	
Disgnosis:	
External Worker ID	
Served by another person living in the home:	
Sort 1: Sort 2:	
Sort 3:	
ReportType(s): PC	P Bacci CSV XML
Seve As Templete Run Ri	sport Cencel

November 14, 2011





Select the **Claim Type** from All Claims, Exported Claims Only, Non-Exported Claims Only or Specific Claim. If Specific Claim is chosen, you must supply a Claim number for the claim on which you wish to report.



Select the **Claim Dates** from Current Day, Current Week, Current Month, Current Quarter, Current Year, Last Week, Last Month, or Fixed Date Range. If you select Fixed Date Range, you must specify a "from" and "to" date (mm/dd/yyyy).

* Claim Dates:	×
Client:	Fixed Date Range
Provider:	Current Day Current Week
Worker:	Current Month Current Quarter
Case Manager:	Current Year Last Week
Service:	Last Month

Select **Group By** so the claims listed are grouped according to your selection. The options include Client, Provider, Exception (Client), and Exception (Provider).

* Group By:	~
Team Assignment:	Client Case Manager Provider Exception (Client) Exception (Provider)

Additional filter criteria include Client, Provider, Worker, Service and Waiver.

If desired, you may also select sort criteria which include Start Date, Provider, Client, Worker, and Service. This determines the order of the data within the group selected in Group By.

Sort 1:	
Sort 2: Sort 3: Diagnosis:	Start Date Provider Worker Client Case Manager Service





An example of a Claim Details report:

Kans	as	Auth	entiC	are C	laim	Det	ails	<b>i</b>			
PHILLIPS	, BRANDON (000674	68911)		Total Numb	per of Claim	s: 1	т	otal Amoun	t: \$ 14.92		
Provider:	CENTRAL PLAINS AAA	(100260949A)									
Claim	Worker	Service	Service Type	Claim Start	Claim End	Total Units	Auth Units	Total Amount	Auth Amount	Export Date	Exceptions
447	Braskens, BIII (86925)	Level 2 Attendant Care (HCFES5125)	TimeBased	11/07/2011 08:05 AM	11/07/2011 09:12 AM	4	4	14.92	14.92		EventMatching
Number of C	Claims: 1										
Total Amou	nt: \$ 14.92										

# 13.5.4 CLAIM HISTORY REPORT

The Claim History Report lists the detail of changes made to a claim or group of claims for auditing purposes. For example, a claim was confirmed for billing and there is a need to know who confirmed it.

The Claim History Report as displayed in the screenshot below only requires a filter for **Claim Dates** which include Current Day, Current Week, Current Month, Current Quarter, Current Year, Last Week, Last Month, or Fixed Date Range. If you select Fixed Date Range, you must specify a "from" and "to" date (mm/dd/yyyy).

Claim History Report * Indicates a required field.
* Report Name: Claim Hallory Roport
Description:
Claim:
* Claim Dates:
Client:
Provider: Oklahoma Homa Haalth Provide
Worker:
Case Manager:
Service:
Team Assignment:
Program:
At Risk: N
Texonomy Code:
NPI
Disgnosis:
External Worker ID
Served by another person living in the home:
Sort 1:
Sort 2:
Sort 3:
ReportType(s): 2707 _ Stoci _ CSV _ XML
Seve As Templete Run Report Cencel





Additional filter criteria include Claim (to report on a specific claim), Client, Provider, Worker, Service and Waiver.

If desired, you may also select sort criteria which include Start Date, Provider, Client, Worker, and Service.

Sort 1:	· · · · · · · · · · · · · · · · · · ·
Sort 3:	Start Date Provider Worker Client Case Manager Service

An example of the Claim History report:

Claim: 447 Client: F	HILLIPS, BRANDO			UthentiCare Claim History Provider: CENTRAL PLAINS AAA (100260949A) Worker: Braskens, Bill					(86925)
Last Update	Updated By	Service	Service Type	Claim Start	Claim End	Actual Units	Auth Units	Actual Amt	Auth Amt Exceptions
11/07/2011 08:05 AM	acr_admin@acr.com	Level 2 Attendant Care (HCFES5125)	TimeBased	11/07/2011 08:05 AM					EventMatching
11/07/2011 09:07 AM	acr_admin@acr.com	Level 2 Attendant Care (HCFES5125)	TimeBased	11/07/2011 08:05 AM	11/07/2011 09:02 AM	4	4	14.92	14.92 ConfirmBillingForClaim, EventMatching
11/07/2011 09:14 AM	acr_admin@acr.com	Level 2 Attendant Care (HCFES5125)	TimeBased	11/07/2011 08:05 AM	11/07/2011 09:12 AM	4	4	14.92	14.92 ConfirmBillingForClaim, EventMatching
11/07/2011 09:14 AM	acr_admin@acr.com	Level 2 Attendant Care	TimeBased	11/07/2011 08:05 AM	11/07/2011 09:12 AM	4	4	14.92	14.92 EventMatching

# 13.5.5 CALENDAR REPORT

The Calendar Report lists all scheduled events for a selected time period. The report can be filtered to include only events related to a particular client, worker or service. The report can be generated by the day, by the week or by the month.





The Calendar Report as displayed in the screenshot below only requires a filter criteria for **Effective Dates** which include Current Day, Current Week, Current Month, Current Quarter, Current Year, Last Week, Last Month, or Fixed Date Range. If you select Fixed Date Range, you must specify a "from" and "to" date (mm/dd/yyyy).

* Report Name: Colondar Report	· · · · · · · · · · · · · · · · · · ·
Description:	4
* Effective Dates:	~
Events	
Authorization:	
Clients	2
Provider: Oklaho	ma Hama Haaldi Pravida
Primary Worker:	2
Service:	19
Team Assignment:	×
Program:	~
At Risk:	N
Taxonomy Code :	
NPI:	
Disgnosis:	
External Worker ID	
ierved by another person living in the home:	
Sort 1:	×
Bert 2	. N
Seri bi	

Additional filter criteria include Event (to report on a specific event), Authorization (to report on the events scheduled for a specific authorization), Client, Provider, Primary Worker and Service.

If desired, you may also select sort criteria which include Scheduled Start, Provider, Client, Worker, Event, Authorization, and Service.

Sort 1:	~	
Sort 2:	Scheduled Start	
SOPE 3:	Provider Worker Client	
Diagn	Event Authorization	
Savo As Tomplato	Service	ancol

November 14, 2011





An example of the Calendar Report:

	4	AuthentiCare	Ð		
(86925)					Event Total: 1
Scheduled Start	Scheduled End	Client	Provider	Service	Authorization ID
11/3/2011 9:10:00 AM	11/3/2011 9:40:00 AM	STUBBS, DREW (00082797611) 123 MAIN ST	CENTRAL PLAINS AAA (100260949A)	Level 2 Attendant Care (HCFES5125)	423
		WICHITA, KS 672163045			
		000000000			
	Scheduled Start	(86925) Scheduled Start Scheduled End	(86925)         Client           11/3/2011 9:10:00 AM         11/3/2011 9:40:00 AM         STUBBS, DREW (00082797611)           123 MAIN ST         WICHITA, KS 672163045	Scheduled Start         Scheduled End         Client         Provider           11/3/2011 9:10:00 AM         11/3/2011 9:40:00 AM         STUBBS, DREW (00062797611)         CENTRAL PLAINS AAA (100260949A)           123 MAIN ST         WICHITA, KS 672163045         WICHITA, KS 672163045	Image: With the start         Scheduled End         Client         Provider         Service           11/3/2011 9:10:00 AM         11/3/2011 9:40:00 AM         STUBBS, DREW (00082797611)         CENTRAL PLAINS AAA (100260949A)         Level 2 Attendant Care (HOFESS125)           123 MAIN ST         WICHITA, KS 672163045         WICHITA, KS 672163045

### 13.5.6 LATE AND MISSED VISITS REPORT

The Late and Missed Visits Report lists all late and missed visits for a selected time period. The report can be filtered to display information relating to a particular client, worker, service or event.

The Late and Missed Visits Report as displayed in the screenshot below only requires filter criteria for **Effective Dates** which include Current Day, Current Week, Current Month, Current Quarter, Current Year, Last Week, Last Month, or Fixed Date Range. If you select Fixed Date Range, you must specify a "from" and "to" date (mm/dd/yyyy).

* Report Name: Late and Mass	d Valla Report
Description:	(i) (i)
* Effective Dates:	
Eventi	
Statua	×
Authorization:	
Client:	2
Provider: Oklah	ama Hama Haaldi Pravid
Primary Worker:	19
Service:	12
Team Assignment:	~
Program:	×
At Risk:	N
Taxonomy Code:	
NPI:	
Diegnosis:	
External Worker ID	
rved by another person living in the home:	
Sort 1:	
Seri 1.	×
2011 21	N

November 14, 2011





Additional filter criteria include Event (to report on a specific event), Status (Late, Missed or Completed Late), Authorization, Client, Provider, Primary Worker and Service.

If desired, you may also select sort criteria which include Scheduled Start, Provider, Client, Worker, Event, Authorization, and Service.



An example of the Late and Missed Visits Report:

Kansas AuthentiCare Schedule									Ð				
ST	<b>FUBB</b>	S, DRE	W (0008279	)7611)								Event Total:	1
F	Provide	er: CEN	TRAL PLAINS	AAA (100260949A	)								
	Event	Status	Worker	Service	Service Type	Threshold (minutes)	Schedule Start	Schedule End	Claim ID	Claim Start	Email Sent	Acknowledged By	Missed Visit Code
	417	Missed	Braskens, BII (86925)	Level 2 Attendant Care (HCFES5125)	TimeBased		11/03/2011 09:10 AM	11/03/2011 09:40 AM			11/03/2011 11:10 AM	central@aol.com	3
	lumber (	of Events:	1										

### 13.5.7 EXCEPTION REPORT

Exceptions are used to readily identify claims that do not meet the business rules established for the program. Exceptions can be informational to alert the user that a criterion was not met, i.e. check in phone number does not match authorized, or can be critical which prevents the claim from being exported to KMAP/MMIS for adjudication, i.e. no authorization for service. The Exception Report is structured to identify exceptions for a single client or for multiple clients with the same exception.





The Exception Report as displayed in the screenshot below has several filter criteria. **Claim Type**, **Claim Dates**, and **Exception** are all required when running the report. Select from the Exception list which exceptions you want returned in the report. Hold down the Ctrl key to select more than one type of Exception.

Exception Report * Indicates a required field.							
* Report Name:	Exception Report						
Description:		8					
* Claim Type:		M					
* Claim Dates:	×						
Client:							
Provider:	Oklahoma Home Heal	th Provide					
Worker:							
Case Manager:							
	Al Services						
	Respice In-Home Advanced Supportive/Rest						
Service:	RN Assessment/Evaluatio						
	Case Management - 5						
	Case Management - VR	۵					
	Al Exceptions						
	Authorize	-					
* Exception:	Calculate	-					
- exception:	AuthExhaustedOn AuthExhaustedZefore						
	WorkerSigbility	-					
Sort	1:	N					
Sort	2:	N					
Sort	3:	S					
ReportType(s):	ReportType(s): Y POF Stoci Stoci Sty						
Sever An Terrolet	te Run Report	Cencel					
and the second sec	ter interport	Control					

Select the **Claim Type** from All Claims, Exported Claims Only, Non-Exported Claims Only or Specific Claim. If Specific Claim is chosen, you must supply a Claim number for the claim on which you wish to report.

* Claim Type:	All Claims	~
* Claim Dates:	All Claims	
	Exported Claims Only Non-Exported Claims Only	
Provider:	Specific Claim	

Select the **Claim Dates** from Current Day, Current Week, Current Month, Current Quarter, Current Year, Last Week, Last Month, or Fixed Date Range. If you select Fixed Date Range, you must specify a "from" and "to" date (mm/dd/yyyy).

November 14, 2011





* Claim Dates:	~
Client:	Fixed Date Range
Provider:	Current Day Current Week
Worker:	Current Month Current Quarter
Case Manager:	Current Year Last Week
Service:	Last Month

Select the **Exception** from the list of information and critical exceptions or choose All Exceptions.

	All Exceptions	^
	Authorize	
Exceptions	Calculate	_
Exception:	AuthExhaustedOn	
	AuthExhaustedBefore	
	WorkerEligibility	~

Additional filter criteria include Client, Provider, Worker and Service.

If desired, you may also select sort criteria which include Service Date, Check In Time, Claim Number, Client, Worker ID, Worker Name, and Service. This determines the order of the data.



An example of the Exception Report:





Kans				Exception Report								-		Ð
Report Da	te: November	10, 2011 08:4	4:11 AM						Filtered By: Exception	Date Range	,Claim Type,	Provider	ID, Ser	vice,
Date Rang	ge: 2011-11-07	to 2011-11-1	3					:	Sort by:					
Provider k	d: 100260949A			Cla	aim Exceptio	n Record Return	ned : 7		Case Mana	ger ld:				
Worker Id:	:			Claim Type: All Claims Service: All										
Client ID:									Exception:	All				
Provider Name	CENTRAL PLA	NS AAA (100260	1949A)	Number Of Claims With Exceptions: 7					Number of Exceptions: 7					
Exception	Claim Number	Client ID	Client Name	CaseManager Name	Worker ID	Worker Name	Check in Time	Check in Number	Check Out Time	Check Out Number	Actual Date Of Service	Actual Unit	Auth Unit	Service
E1	447	00067468911	PHILLIPS, BRANDON		86925	Braskens, Bill	8:05AM		9:12AM		11/07/2011	4	4	HCFES5125
E1, G1, G2	451	00067468911	PHILLIPS, BRANDON		86925	Braskens, Bill	6:30AM	5135550922	7:15AM	5138904454	11/07/2011	3	3	HCFES5125
C1, E1	459	00082797611	STUBBS, DREW		86925	Braskens, Bill	10:09AM		10:28AM		11/09/2011	1	1	HCFES5125
C1, E1	460	00082797611	STUBBS, DREW		86925	Braskens, Bill	4:30PM		6:15PM		11/08/2011	7	7	HCFES5125
A1, C1, E1, I4, I5	461	00082797611	STUBBS, DREW		86925	Braskens, Bill	11:30PM		5:30AM		11/07/2011	0	0	HCDDH0045
	463	00082797611	STUBBS, DREW		86925	Braskens, Bill	2:40PM		6:01PM		11/07/2011	0	0	HCFET2025
A1, C1, E1														



The exceptions are grouped by claim number. Claim number is equivalent to Claim ID. For the screenshot above, Claim Number 459 has an Event Matching exception and a Billing Confirmation exception.

### 13.5.8 UNAUTHORIZED PHONE NUMBER REPORT

The Unauthorized Phone Number Report will provide ready access to a list of calls that were made from a phone other than the phone number associated with the client. These calls result in a claim with an unauthorized phone number exception. This report serves as an administrative tool allowing the Provider or State Administrative User to:

- Identify workers making calls from outside the home
- Identify phone numbers that have changed and need updating in KS AuthentiCare





The Unauthorized Phone Number Report as displayed in the screenshot below has several filter criteria. **Claim Type**, **Claim Dates**, and **Exception** are all required when running the report.

Unauthorized Phone Number Report * Indicates a required field.						
* Report Name:	Una utherize di Phonie Number Report					
Description:	8 9					
* Claim Type:	<u></u>					
* Claim Dates:	~					
Client:						
Provider: (	Sklahoma Home Health Provide					
Worker: [						
Case Manager:	2					
Service:	All Services Respire In-Nome Advanced Supportive Restorative RN Assessment Evaluation Case Usingement - 5 Case Nangement - VR					
	Al Caina					
	Unsutherized Calls Only					
* Exception:						
Sort 3						
Sort 2	×					
Sort 3	s 🔊					
ReportType(s):	ReportType(s): Y FOF Excol CDV XML					
Seve As Templet	Run Report Cencel					

Select the **Claim Type** from All Claims, Exported Claims Only, Non-Exported Claims Only or Specific Claim. If Specific Claim is chosen, you must supply a Claim number for the claim on which you wish to report.

* Claim Type:	All Claims 💌
toll pr	
* Claim Dates:	All Claims
Client:	Exported Claims Only
client:	Non-Exported Claims Only
Provider:	Specific Claim
Provider.	





Select the **Claim Dates** from Current Day, Current Week, Current Month, Current Quarter, Current Year, Last Week, Last Month, or Fixed Date Range. If you select Fixed Date Range, you must specify a "from" and "to" date (mm/dd/yyyy).

* Claim Dates:	~
Client:	Fixed Date Range
Provider:	Current Day Current Week
Worker:	Current Month Current Quarter
Case Manager:	Current Year Last Week
Service:	Last Month

Select "Unauthorized Calls Only" for **Exception**. If you select "All Claims", all claims with exceptions will be returned; not just the ones with unauthorized phone number exceptions.

All Claims
Unauthorized Calls Only

Additional filter criteria include Client, Provider, Worker and Service.

If desired, you may also select sort criteria which include Service Date, Check In Time, Claim Number, Client, Worker ID, Worker Name, and Service. This determines the order of the data.

Sort 1:	×	
Sort 2: Sort 3:	Service Date Check In Time Claim Number	
Save As Template	Client Worker ID	Cancel

An example of the Unauthorized Phone Number Report:

Ka	insas		ı	Jnaut	horize	d Ph	one	Nur	nbe	er Rej	port			Page 1 o	£1
Report	Date: Nover	mber 10, 2011 08:53 11-07 to 2011-11-07	:11 AM					Ex	tered By ception		ge,Claim	Type, I	Provider ID, S	iervice,	
	er ld: 100260				Total Record	s: 1				ager Id:					
Worker	r Id:				Claim Type: \$	Specific Cla	aim - 451	Se	rvice: A	II					
Client I	ID:							Ex	ception	:All					
Provider k	d : 100260949A	i i i i i i i i i i i i i i i i i i i	Provider	Name: CENTRA	AL PLAINS AAA										
Worker Id	: 86825		Worker N	lame : Brackens	s, BIII										
Claim Number	Client ID	Client Name	SVC	Exceptions		Date of Service	Check In	Check Or	ut Units	Cheok in Phone	Matches Name	ld /	Cheok Out Phone	Matohes Name	ld /
451	00067468911	PHILLIPS, BRANDON	HCFE8 5125	E1, G1, G2		11/7/2011	6:30AM	7:15AM	3	5135550922	Not Found	1	5138904454	Not Found	4

November 14, 2011





### 13.5.9 BILLING INVOICE REPORT

The Billing Invoice report gives a list of claims for each service date, along with the KMAP/MMIS billing status and amount. With this report, providers have documented what was submitted to KMAP/MMIS each day and then monitor the Remittance Advice to ensure that each claim was adjudicated as expected.

The Billing Invoice Report as displayed in the screenshot below has several filter criteria. **Claim Type**, **Claim Dates**, and **Exception** are all required when running the report.

Billing Invoice * Indicates a req	
* Report Name:	Siling Invoia: Report
Description:	8
* Claim Type:	<u></u>
* Claim Dates:	~
Client:	
Provider: (	Oklahoma Home Health Provide
Worker:	
Case Manager:	
Service:	Al Sevices  Resple In-Home Rounces Supportive Restantive RN Assessment Evaluation Case Management - 5 Case Management - VR.
* Exception:	Al Exceptions  Authoritie Calculate Authoritie Calculate AuthoritieutieoOn AuthoritieutieoEn WorkenEigbity
Sort	1:
Sort 3	2:
Sort :	5: 💊
	V POP - Excel - CSV - XML
Seve As Templet	e Run Report Cencel

Select the **Claim Type** from All Claims, Exported Claims Only, Non-Exported Claims Only or Specific Claim. If Specific Claim is chosen, you must supply a Claim number for the claim on which you wish to report.

* Claim Type:	All Claims 💌
* Claim Dates:	All Claims
Client:	Exported Claims Only Non-Exported Claims Only
Provider:	Specific Claim





Select the **Claim Dates** from Current Day, Current Week, Current Month, Current Quarter, Current Year, Last Week, Last Month, or Fixed Date Range. If you select Fixed Date Range, you must specify a "from" and "to" date (mm/dd/yyyy).



Select the **Exception** from the list of information and critical exceptions or choose All Exceptions.



Additional filter criteria include Client, Provider, Worker and Service.

If desired, you may also select sort criteria which include Service Date, Check In Time, Claim Number, Client, Worker ID, Worker Name, and Service. This determines the order of the data.

Sort 1:	×	
Sort 2:	Service Date	
Sort 3:	Check In Time	
	Claim Number Client	
Save As Template	Worker ID	Cancel
Save As Template	Worker Name Service	Jancer

An example of the Billing Invoice Report:





K	ansas				E	Billing	Invo	ice R	epo	ort				G
Report Date: April 05 2010 E DE 59 AM Date Range: 115/2010 to 3/15/2010 Provider Id: Worker Id:				Total Records Returned: 10 Claim Type: Exported Claims Only				Fitered By: Date Range,Claim Type, Provider ID, Service, Exception Sort by Case Manager Id. Service: All						
Client IC	2:									Excep	ition:All			
Provider					Provider	Name: Booner He	atton							
Data Of 8 Claim Number	Clert Name	Medicale	CINE	Calm Matue	Gervice	Worker Name	Check In	Check Out	Actual	Auto	Pair	Payment Cate	Export	Exceptore
512	Ward McLain	123456789	125458789	info@xqr8	11019	Setay Jones	1:00PM	2:00PM	4		5.00		-	Eventión
113	Adrie, Milly	10767654321	967654321	Info@xqr8	T1000	Setay Jones	215PM	3.15PM	4	4	0.00			Event/Ith
							3	Daty Tolare 1			8.65			1
Claim Claim Number	Cleft Name	Nedtard Nuttier	Cherto	Cure Sulue	3979189 111005	Wolker Natio	Check In	Check Out	Actual	Auth	Para	Payment Date	Export Data	Exceptore
	11000,000000	- and and	1.0.000		peare	lecalized	_	Daily Totara 2	COLUMN 1			_	-	1
Dela Of 8 Claim Rumbar 1 1	entra ELOXORIA Client Name	Medicard Rottbar Jat7114221	Cilent 10	Calm Balan	Bervice Troos	Worker Name	Check In 1 SCPM	Check Out 2 SQFW	4	Auth Set	Paid Amount 1.00	Payment	Expan	Exceptions Exercision
								Delly Totals :			8.85			1
Data Or 8 Claim Number	ervice: #20000010 Client Name	Notices Renter	Civit ID	Ciere Batue	tervice	Worker Name	Check In	Check Out	Actual	Auth	Part	Payment	Export Data	Exceptors
818	Aana, Multy	967674321	967654321	InfoExants	T1018	Setay Jones	2-55AM	11:25AM	4	4.	0.00		-	EventMits, Cicinitalian, Cicilianian
818	Aate, Moly	887654321	987654321	infoExcela	11008	Snet Johnson	11:DOAM	12:00PM	4		0.00			EventMith, CkinNoMatch, WorkServ
							0.00	Daily Totain 1	000000		8.85			

### 13.5.10 TIME AND ATTENDANCE REPORT

The Time and Attendance Report is a useful tool for the providers who need to know the time billed by a selected worker for a specified time period. It can be used to identify workers who report an unusually high number of hours worked as that could be considered a risk for quality of care issues or for providers to use to compare revenue generated by one worker with another.





The Time and Attendance Report as displayed in the screenshot below has several filter criteria. **Claim Type**, **Claim Dates**, and **Exception** are all required when running the report.

	Time and Attendance Report * Indicates a required field.						
* Report Name:	Time and Attendiance Report						
Description:	8						
* Claim Type:	<u> </u>						
* Claim Dates:	<b>v</b>						
Client:							
Provider: (	Oklahoma Home Health Provide						
Worker:							
Case Manager:							
Service:	Al Services Respira In-Korne Advanced Supportive Restorative RN Assessment Svaluation Case Management - VR Case Management - VR						
* Exception:	Al Exceptions  Authorize Calculate AuthorizeCin						
	AuthExhaustedZefore WorkerEligbility						
Sort	s						
Sert : Sert :							
	V POF Excel CSV 2XML						

Select the **Claim Type** from All Claims, Exported Claims Only, Non-Exported Claims Only or Specific Claim. If Specific Claim is chosen, you must supply a Claim number for the claim on which you wish to report.

-		
* Claim Type:	All Claims	~
* Claim Dates:	All Claims	
Client:	Exported Claims Only Non-Exported Claims Only	
Provider:	Specific Claim	

Select the **Claim Dates** from Current Day, Current Week, Current Month, Current Quarter, Current Year, Last Week, Last Month, or Fixed Date Range. If you select Fixed Date Range, you must specify a "from" and "to" date (mm/dd/yyyy).





* Claim Dates:	<b>~</b>
Client:	Fixed Date Range
Provider:	Current Day Current Week
Worker:	Current Month Current Ouarter
Case Manager:	Current Year Last Week
Service:	Last Month

Select the **Exception** from the list of information and critical exceptions or choose All Exceptions.

	All Exceptions	^
* Exception:	Authorize Calculate AuthExhaustedOn AuthExhaustedBefore	
	WorkerEligibility	~

Additional filter criteria include Client, Provider, Worker and Service.

If desired, you may also select sort criteria which include Service Date, Check In Time, Claim Number, Client, Worker ID, Worker Name, and Service. This determines the order of the data.



An example of the Time and Attendance Report:





<b>K</b> ans	sas				Time	and A	tten	danc	e Re	еро	rt			2
Report Date:         November 10, 2011 06:53:39 AM         Total Records Return           Date Range:         2011-11-07 to 2011-11-13         Cialm Type: All Cial           Provider Id:         100260949A         Cialm Type: All Cial						Sort by:						D, Service, Exception		
Client ID:									Exception	n:All				
	: 100250949A				Name: CENTRAL PLA	UNS AAA								
Worker Id : Claim Number	Client ID	Client Name	Date of Service		ame : Braskens, Bill Worker Name	Worker ID	Check In	Check Out	Actual Unit	Auth Unit	Actual Amount	Auth Amount	Export Date	Exceptions
447	00067468911	PHILLIPS, BRANDON	11/07/2011	HCFE8 5125	Breskens, Bill	86925	8:05AM	9.12AM	4	4	14.92	14.92		E1
451	00067468911	PHILLIPS, BRANDON	11/07/2011	HCFE8 5125	Braskens, Bill	86925	6:30AM	7:15AM	3	3	11.19	11.19		E1, G1, G2
459	00082797611	STUBBS, DREW	11/09/2011	HCFE8 5125	Braskens, Bill	86925	10:09AM	10:28AM	1	1	3.73	3.73		C1, E1
480	00082797611	STUBBS, DREW	11/08/2011	HCFE8 5125	Braskens, Bill	86925	4:30PM	6:15PM	7	7	28.11	28.11		C1, E1
461	00082797611	STUBBS, DREW	11/07/2011	HCDDH 0045	Braskens, Bill	86925	11:30PM	5:30AM	0	0	0.00	0.00		A1, C1, E1, I4, I5
463	00082797611	STUBBS, DREW	11/07/2011	HCFET 2025	Braskens, Bill	86925	2:40PM	6.01PM	0	0	0.00	0.00		A1, C1, E1
465	00082797611	STUBBS, DREW	11/07/2011	HCFET 1001	Braskens, Bill	86925	12:00AM		1	0	0.00	0.00		A1, C1, E1
								Vorker Total : ovider Total :		15	55.95	55.95		

### 13.5.11 OVERLAPPED CLAIM REPORT

The Overlapping Time Report is useful in identifying quality concerns and/or overpayments. Under usual circumstances, workers should complete care for one client before moving on to provide care to another client. Monitoring the Overlapped Claim Report enables the user to identify clients whose care may be compromised as well as workers that may have forgotten to check out from one service before beginning to provide another service to the same or another client. This report is also helpful in determining patterns for specific workers that may need targeted training/retraining or reminders of program requirements and expectations.





The Overlapped Claim Report as displayed in the screenshot below has several filter criteria. **Claim Type** and **Claim Dates** are required when running the report.

Overlapped Cl	aim Report
* Indicates a reg	uired field.
* Report Name:	Overlap ped Claim Reipert
	8
Description:	
A Claim Type:	
* Claim Dates:	<u>×</u>
Client:	
and the second	Sklahoma Home Health Provide
	~
Worker:	
Case Manager:	
-	
Overlap:	<u>All N</u>
	Al Services
	Respice In-Home
Service:	Advanced Suggerive Restorative 🗧
Service:	RN Assessment/Evaluation Case Management - S
	Case Management - S Case Management - VR
Sort 3	L: V
Sort	
Sort	s: N
Recent Trees (a):	V ROF Exect CSV XML
weborer Abe(a):	Contraction Contraction
Seve As Templet	Run Report Cencel

Select the **Claim Type** from All Claims, Exported Claims Only, Non-Exported Claims Only or Specific Claim. If Specific Claim is chosen, you must supply a Claim number for the claim on which you wish to report.

* Claim Type:	All Claims 💌
* Claim Dates:	All Claims
Client:	Exported Claims Only Non-Exported Claims Only
Provider:	Specific Claim

Select the **Claim Dates** from Current Day, Current Week, Current Month, Current Quarter, Current Year, Last Week, Last Month, or Fixed Date Range. If you select Fixed Date Range, you must specify a "from" and "to" date (mm/dd/yyyy).

* Claim Dates:	×
Client:	Fixed Date Range
Provider:	Current Day Current Week
Worker:	Current Month Current Quarter
Case Manager:	Current Year Last Week
Service:	Last Month





If you wish, you may change the selection for **Overlap**. This automatically defaults to "All", but you can change it to "Clients" or "Services".

Overlap:	All	<b>~</b>	
	All		
	Clients	jement - VR	~
	Services	re	_

Additional filter criteria include Client, Provider, Worker and Service.

If desired, you may also select sort criteria which include Service Date, Check In Time, Claim Number, Client, Worker ID, Worker Name, and Service. This determines the order of the data.

Sort 1:	×	
Sort 2:	Service Date	
Sort 3:	Check In Time Claim Number	
	Client	
Save As Template	Worker ID Worker Name	Cancel
	Service	

An example of the Overlapped Claim Report:

Kansas		Overlapped Claim Report					t			
Report Date: April 07 2010 11.31.32 AM Service Date Range: 1/1/2010 to 4/6/2010 Provider Id: 472 Worker Id:						Filtered By: Date Range,Claim Type, Provider ID, S Sort by:				
		Total OverLaps Returned: 1 Claim Type: All Claims				Case Manager Id: Service: All Exception:				
										Dient Id.
Provider Id : 472 Worker Id : 85490										Provider Name: Sconer Heathcare Worker Name : Chris Kappler
Claim Mumber	Client ID	Client Name	RVC.	Date of Service	Check In Tame	Check Out Time	Check In Places	Check Out Phone	Check-In Ph. Owner	Check-Out Ph Owner
617	676543219	Weekley, Bolo	T1019	03/25/2010	1.50PM	2.05PM	9185948101			
618	963852741	McCattrey, Paul	T1016	03/26/2010	1.00PM	2:00PM				

#### 13.5.12 WORKERS BY PROVIDER REPORT

The Workers by Provider Report is used for monitoring purposes or can be used by providers to determine workers that are currently employed to provide care. Other uses of the report include determining worker to client ratios or validating that all workers employed by the provider are registered in the system.




The Workers By Provider Report as displayed in the screenshot below has filter criteria of **Worker Status**. Select from "All", "Active", or "Inactive".

Worker By Provider Report * Indicates a required field. * Report Name: Worker By Provider Report Description:
Worker Start Date Range: S Provider: Oklahama Hama HaalO Provida@ Worker: S
Al Services Respite In-Home Advanced Supports Restorative Service: RN Assessment (Subation Case Management - S Case Management - VR
* Worker Status: All
Sort 1:
ReportType(s): V pop dexect CSV d XML
Seve As Templete Run Report Cancel

Additional filter criteria include Worker Start Date Range, Provider, Worker and Service.

If desired, you may also select sort criteria which include Worker Name, Worker ID, Worker Status, and Start Date. This determines the order of the data.

Sort 1:	×
Sort 3:	Worker Name Worker ID Worker Status Start Date





An example of the Workers by Provider Report:

Kansas	Workers b	y Provid	ler Re	port
Report Date: November 07, 2011 09:4	3:21 AM			l By: Worker Start Date Range, Provider ID, Status, Service
Provider ID: 100258959A			Sort by	:
Worker ID:			Worker	Status: All
	I otal Recon	ds Returned: 2	Service	e All
Provider Name : SOUTHWEST KANSAS AA	Provider ID : 1002	58959A		ce: ;FES5125UD,HCFES5130,HCFES5101,HCFES5160 CFET1001,HCFET2025,HCFES5135,HCFES5135UD
Worker Id Worker \$\$N Worker	lame Start Date	Termination Date	Sanctions	Worker Service
01213 Choo, Si	Soo			HCFES5125,HCFES5130
28385 Slammy,	Super			HCFES5125,HCFES5125UD,HCFES5130

#### 13.5.13 PROVIDER ACTIVITY REPORT

A report that lists, by worker, all services performed during a given time period and the total dollars billed to KMAP/MMIS, again by worker. The Provider Activity Report is a useful tool for State monitoring or for the providers who need to know the revenue billed by a selected worker for a specified time period. It can be used to identify workers who report an unusually high number of hours worked as that could be considered a risk for quality of care issues or for providers to use to compare revenue generated by one worker over another.





The Provider Activity Report as displayed in the screenshot below has several filter criteria. **Claim Type, Claim Dates,** and **Exception** are required when running the report.

Provider Activi * Indicates a req	
* Report Name:	Provide r Adivity Report
Description:	8
* Claim Type:	
* Claim Dates:	×
Client:	
Provider: (	Oklahoma Home Health Provide
Worker:	12
Case Manager:	2
Service:	All Services Respite In-Nome Advanced Supportive Restantive RN Assessment Subjection Case Usingement - 5 Case Management - VIR
* Exception:	Al Exceptions  Authorse Catalates Authorse AuthorseuterCon AuthScheuterCon WorkerEigbilty
Sort : Sort : Sort :	2:
ReportType(s):	Y PDF Exect CSV □XML
Seve As Templet	e Run Report Concel

Select the **Claim Type** from All Claims, Exported Claims Only, Non-Exported Claims Only or Specific Claim. If Specific Claim is chosen, you must supply a Claim number for the claim on which you wish to report.

* Claim Type:	All Claims
* Claim Dates:	
* Claim Dates:	All Claims
	Exported Claims Only Non-Exported Claims Only
Client:	Non Exported Claims Only
Provider:	Specific Claim
Provider:	





Select the **Claim Dates** from Current Day, Current Week, Current Month, Current Quarter, Current Year, Last Week, Last Month, or Fixed Date Range. If you select Fixed Date Range, you must specify a "from" and "to" date (mm/dd/yyyy).



Select the **Exception** from the list of information and critical exceptions or choose All Exceptions.



Additional filter criteria include Client, Provider, Worker and Service.

If desired, you may also select sort criteria which include Service Date, Check In Time, Claim Number, Client, Worker ID, Worker Name, and Service. This determines the order of the data.

Sort 1	:	
Sort 2	Service Date	
Sort 3	Check In Time	
	Claim Number — Client	
Cours de Tonnelat	Manhan TD	2- u - e l
Save As Template	Worker Name Service	Cancel





An example of the Provider Activity Report:

Kan	sas			Pro	vide	r Act	ivity	Repo	ort							Page 1 of 1
Report Da	ate: Novemb ge: 2011-11 ld: f:	er 07, 2011 09:4 -07 to 2011-11-0					cords Return /pe: All Clain			Flitered I Sort by: Case Ma Service: Exceptio	anager I All		alm Type,	Service,	Exception	
Provider Id	: 100260949A				Provider I	Name: CENT	RAL PLAINS A	LAA								
Worker id :	86925				Worker Na	me : Brask	ms, Bill									
Claim Number	Client ID	Client Name	Claim Status	Date of Service	Service	Activity Code(s)	Worker ID	Check In	Check Out	Actual Unit	Auth Unit	Paid Amount	Payment Date	Export Date	Exceptions	
447	00067468911	PHILLIPS, BRANDON	InfoExcpts	11/07/2011	HCFE8 5125	71,73,76	86925	8:05AM	9:12AM	4	4	0.00			E1	
								W	orker Total :	4	4	0.00				1
								Pro	vider Total :	4	4	0.00				1

#### 13.5.14 REMITTANCE ADVICE REPORT

Providers need to be able to balance their accounting records to ensure that payment was received for a service that was provided and billed. This report provides remittance advice reporting on the KS AuthentiCare web, so that providers can examine paid claims, and understand check amounts. It is only available if the provider uploads the 835 electronic remittance advice into KS AuthentiCare. If the provider does not upload the 835, the system has no record of payment, only of claims submitted for payment.

- In contrast to the Provider Activity report, the Remittance report is oriented around the KMAP/MMIS Claim number and not the AuthentiCare claim number, and around payment date, not claim date. The intent is to support drill down of a payment received to the individual claims included in it. Claim reports should still be used to research the payment status of an individual claim.
- The report uses a filter page variation that is specific to the Remittance Advice report. Note that the date range selected is the KMAP/MMIS payment or processing date, not the date of service.
- The Remittance report offers the similar sort and filter criteria as the other reports, but some criteria that are not relevant are omitted.
- Report totals are provided at the end. If a grouping is selected, group subtotals are provided at the end of each group.





The Remittance Advice Report as displayed in the screenshot below has filter criteria of **Payment Dates** and **Group By**.

Remittance Advice Report * Indicates a required field.
* Report Name: Remitiance Advice Report  Description:
* Payment Dates:
Sort 1:
Seve As Templete Run Report Concel

Select the **Payment Dates** from Current Day, Current Week, Current Month, Current Quarter, Current Year, Last Week, Last Month, or Fixed Date Range. If you select Fixed Date Range, you must specify a "from" and "to" date (mm/dd/yyyy).

* Payment Dates:		*
Client:	Fixed Date Range	
Provider:	Current Day	
Provider:	Current week	
Worker:	Current Month	
worker.	Current Quarter	
Case Manager:	Current Year	
case manager.	Last Week	
Comico	Last Month	_

Select Group By from Client, Payee Provider, Worker and no grouping.

* Group By:	✓	
Sort 1: Sort 2:	Client Case Manager Payee Provider Worker no grouping	> >

Additional filter criteria include Client, Provider, Worker and Service.

KS AuthentiCare User Manual





If desired, you may also select sort criteria which include Service Date, Provider, Client, Worker and Service. This determines the order of the data.

Sort 1:	×	
Sort 2:		
501C 2.	Service Date	
Sort 3:	Client	
	Worker	
	Provider	
Save As Template	Service	Cancel
Save As Template	Case Manager	Jancer

An example of the Remittance Advice Report:

Kan	sas		F	Remi	ttanc	e Ad	vice	Rep	ort			Ð
Date Range: -							Filt	tered By:	Work	er Start Dat	e Range	
Provider Id:							So	rted by:				
Worker ID:							Ca	se Manage	er ID:			
							Se	rvices:				
Provider: Home Servic	xes1 ()	Media	oaid ID:987654	321								
MMIS Claim #	Claim	Date of	Client Name	Medicad	Worker ID	Procedure	Charles	Total	Total	Submit to	MMIS	MMIS
mmis claim w	#	Service	Chenchame	Number	WORKER ID	Code	EFT	Billed(\$)	Payment (\$)	MMIS	Paymen t Date	
0817400375811300A		20080630	Patton,Willia m	987654321	987654321	G9012	5985881	45.50	45.50		2008070 3	P
0817400375811300A		20060630	Patton,Willia m	987654321	987654321	T1019	5985881	124.00	124.00		2008070 3	P
0817400375811300A		20060630	Patton,Willia m	987654321	987654321	T1019	5985881	124.00	124.00		2006070 3	P
0817400375811300A		20060630	Patton,Willia m	987654321	987654321	LTC10	5985881	40.00	40.00		2008070 3	P
0817400375811300A		20060630	Patton,Willia m	987654321	987654321	LTC10	5985881	40.00	40.00		2006070 3	P
0817400375811300A		20080630	Patton,Willia m	987654321	987654321	X0273	5985881	13.00	13.00		2008070 3	Р
0817400375811300A		20080630	Patton,Willia m	987654321	987654321	S5165	5985881	1200.00	1200.00		2006070 3	
0817400375811300A		20080630	Patton,Willia m	987654321	987654321	T1019	5985881	124.00	124.00		2008070 3	P

## 13.5.15 CLAIM DATA LISTING REPORT

The KS AuthentiCare List Report gives a provider the ability to download claims data as needed for use in the back-end systems. As with the other KS AuthentiCare reports, the provider must select report criteria on the criteria pages.





* Report Name: C	aim Data Listing Report
Description:	8
	SI
* Claim Type:	
Cianti Pipe.	+ At least one of the date ranges must be select
+ Service Dates:	~
+ Silling Dates:	
+ Payment Dates:	¥
+ Update Dates:	¥
Claim Creation Dates:	¥
Client:	8
Provider:	Oklahoma Home Health Provide
Worker:	
Case Manager:	
	Al Services
	Resple In-Home Advanced Supportive/Restorative
Service:	RN Assessment/Evaluation Case Management - S
	Case Management - VR
Sor	t1: V
Sor	t 2: V
Ser	rt 3:
ReportType(s):	✓ Exect PDF CSV □XML

The Claim Data Listing Report is a report that lists, by provider and worker, all services performed during a given time period and the total dollars billed to and paid by KMAP/MMIS (if the provider uploads the 835 remittance advice into KS AuthentiCare).

- The report has a column for external worker ID for the provider-specific worker ID if used by the provider.
- The report also has a column for "claim create date" and the report can be sorted by that date.

This report is a useful tool for State monitoring or for the provider who needs to know the services delivered by their workers for a specified time period. The list report is a very simple format with a row of column headings followed by a list of data rows so that it is easily integrated with other back office systems.

#### 13.5.16 REMITTANCE DATA LIST REPORT

The Remittance Data List Report provides remittance advice reporting on the KS AuthentiCare web, so that the provider can examine paid claims, and understand check amounts.





Remittance Data L	isting Report	
* Indicates a required	field.	
* Report Name: Rom	Conc. Dele Latir	an Ricci ant
Description:		3
		1
	-	-
* Payment Dates:		M
Client:		
Provider:	ALL 😽	
Provider Medicaid Id:	ALL V	
	W	-
Worker:		
Case Manager:		
	Al Services	
	Respte In-Home	
	Advanced Support	
Service:	RN Assessment/E	
	Case Management Case Management	
Sort 1:		N
Sort 2:		<b>N</b>
Sort 3:		
ReportType(s):	Exect Dor	CSV XML
Save As Templete	Run Report	Cencel
	ine interport.	

The Remittance Data List report is oriented around the KMAP/MMIS claim number and not the KS AuthentiCare claim number, and around payment date, not claim date. The intent is to support drill down of a payment received to the individual claims included in it. Claim reports should still be used to research the payment status of an individual claim.

The Remittance report offers the similar sort and filter criteria as the other provider reports, but some criteria that are not relevant are omitted. **Note** that the date range selected is the MMIS payment or processing date, not the date of service.

V			Remittan	ce Data List	ing Repe	fric		$\bigcirc$																
Kansas																								
Report Date:	5/6/2010	Report Time:	3:32 PN	1																				
Rour Returned:	3																							
Payment Dates:	From	5/5/2010	Te	: 5/5/201	)																			
Client Id:																								
Pravider Id:																								
Pravider Medicaid Id:																								
Warker Id:	2378																							
Care Manager Id:																								
Services																								
		Provider	AuthentiC	Payse	Payse	Rondoring	-					Care	Care		Praced						Lart		Check	HHIS
	Sabaitte	Hedicaid		Pravidar	Pravid	Pravider	Pravidar	Warker	Warker			Heneger	Heneys			Date of	Authoriz	Paid	Billad	Paid		Paymont	ar EFT	
MMIS Claim #	r Claim #	14	14	Hame	er 14	Hame	14	Heme	14	Client Heme	Client Id	Hame	+ 14	Service		Service	ed Unite		Amanat		Date	Dete	Humber	
10100000456785A		235456122F			2378	Bert Healthcare		Smith, Jac	2378	Parriz, Mary	600234561	Cax, Wilma	2376	T1019	T1019	4/28/2010		7	7 \$21.00	\$21.00	4/30/2010	5/5/2010		
10100000456788A				Best Healthcare			2378	Smith, Jac	2378	Wilcox, Todd	606324561	Janer, Fred	4462	T1019TF	T1019TF	4/28/2010		6	6 \$18.00		4/30/2010	5/5/2010		
10100000456801A				Bert Healthcore		Bert Healthcare		Smith, Jac	2378	Begine, Bill	100234964		2376	T1019	T1019	4/28/2010			6 \$18.00		4/30/2010			

The Remittance Data List Report is only available in CSV and Excel formats.

## 13.5.17 ELIGIBLE CLIENT REPORT

This report shows any clients that the provider has authorization to provide services for or has claims. The report provides most of the data elements shown in the client record.

November 14, 2011

KS AuthentiCare User Manual





Eligible Client Data Listing Report <sup>4</sup> Indicates a required field.	
* Report Name: Eight Cliont Cats Using Report	
Description:	
* Eligibility Dates:	
Client:	
Provider: Grants Oklahoma Provider	
Worker:	
Case Manager:	
Al Services Respite In-Home Advances Supportive/Restorative Service: R/Assessment/Svaluation Case Management - S Case Management - VR	(III) (III)
Team Assignment: 🛛 💙	
Program: 🛛 👷	
At Rink: 🛛 😽	
Texonomy Code:	
NPI:	
Diegnosis:	
External Worker ID:	
Served by another person living in the home:	
The second se	_
Sert 1: Client Last Name V Sert 2: Client First Name V	
Sort 3: Medicaid 10	
ReportType(s): ♥ Exccl POP CSV [XML	
Seve As Templete Run Report Cencel	





## Chapter 14 ONGOING USER SUPPORT AND TRAINING

By design, this User Manual can help you research any system issues that you encounter because it fully explains all the functions you use in KS AuthentiCare. As with most documents of this type, searching by topic is often the first step.

- 1. If you are unable to solve your problem by using this manual, you can contact First Data Client Support services at 1-800-441-4667, Option 6 or <u>clientsupport@firstdata.com</u> for assistance.
- 2. For training your staff, a training website and training IVR are available 24/7 and mirror the production web and IVR.
  - The url for the Training Website is <a href="https://ext.fdgs.com/kansas">https://ext.fdgs.com/kansas</a>
  - The toll-free phone number for the Training IVR is (866) 388-2367; application identification#: 60.
- 3. For user support regarding authorizations, the client's plan of care or policies of KDOA or SRS, contact <u>KSAuthentiCare@aging.ks.gov</u>.





# APPENDIX A.1: SERVICE CODES

# **KDOA – Frail Elderly Waiver**

Service Code	Service Name	Unit Definition	Activity Code Required
HCFES5125	Level 2 Attendant Care	15 minutes	Y
HCFES5125UD	Self-Directed Attendant Care	15 minutes	Y
HCFES5130	Level 1 Attendant Care	15 minutes	Y
HCFES5101	Adult Day Care	Time Based *	Ν
HCFES5160	Personal Emergency Response – Install	Per Visit	Ν
HCFES5190	Wellness Monitoring	Per Visit	Ν
HCFET1001	Nurse Evaluation Visit	Per Visit	Ν
HCFET2025	Sleep Cycle Support	6 – 12 hours	Ν
HCFES5135	Provider Directed Comprehensive Support	15 minutes	Ν
HCFES5135UD	Self Directed Comprehensive Support	15 minutes	Ν

\* Maximum 2 visits per day

- If the length of the Adult Day Care is 0-59 minutes, the claim calculates 0 units.
- If the length of the Adult Day Care is 1-5 hours (60 300 minutes), the claim calculates 1 unit.
- If the length of the Adult Care is more than 301 minutes in a 24 hour period from check in, the claim calculates 2 units.





## SRS – MR/DD Waiver

Service Code	Service Name	Unit Definition	Activity Code Required
HCDDT1019	Personal Assistant Services	15 minutes	Y
HCDDT2025	Sleep Cycle Support	Per Visit	Ν
HCDDH0045	Overnight Respite	Per Visit	Ν
HCDDT1000	LPN Specialized Medical Care	15 minutes	Y
HCDDT1000TD	RN Specialized Medical Care	15 minutes	Y

# **SRS – Physical Disability Waiver**

Service Code	Service Name	Unit Definition	Activity Code Required
HCPDS5126U6	Personal Services	1 hour	Y
HCPDT2025	Sleep Cycle Support	6+ hours	Ν

# SRS – Traumatic Brain Injury Waiver

Service Code	Service Name	Unit Definition	Activity Code Required			
HCTBIS5126UB	Personal Services	1 hour	Y			
HCTBIT2025	Sleep Cycle Support	6+ hours	Ν			

# SRS – Technology Assisted Waiver

Service Code	Service Name	Unit Definition	Activity Code Required
HCTAT1019	Personal Service Attendant	15 minutes	Y

November 14, 2011





# APPENDIX A.2: ACTIVITY CODES FOR SELECTED SERVICES

## ACTIVITY CODES FOR ATTENDANT CARE SERVICES LEVEL I, LEVEL II, AND SELF-DIRECTED IN THE FRAIL ELDERLY WAIVER

Service	Activity	Code
S5125, S5125 UD & S5130	Bathing/Grooming	70
S5125, S5125 UD & S5130	Dressing/Undressing	71
S5125, S5125 UD & S5130	Toileting	72
S5125, S5125 UD & S5130	Mobility	73
S5125, S5125 UD & S5130	Eating	74
S5125, S5125 UD & S5130	Meal Preparation	75
S5125, S5125 UD & S5130	Shopping	76
S5125, S5125 UD & S5130	Accompanying to Medical Appointment	77
S5125, S5125 UD & S5130	Laundry/Housekeeping	78
S5125, S5125 UD & S5130	Management of Meds/Treatments	79





## ACTIVITY CODES FOR PERSONAL ASSISTANT SERVICES (T1019) IN THE MR/DD WAIVER AND PERSONAL SERVICES (S5126) IN THE PD AND TBI WAIVERS

Service	Activity	Code
T1019 & S5126	Bathing	11
T1019 & S5126	Dressing	12
T1019 & S5126	Oral Hygiene	13
T1019 & S5126	Hair Care	14
T1019 & S5126	Skin Care	15
T1019 & S5126	Nail Care	16
T1019 & S5126	Shaving	17
T1019 & S5126	Prosthetic/Orthotic Assistance	18
T1019 & S5126	Toileting	19
T1019 & S5126	Transfer	20
T1019 & S5126	Walking/Mobility	21
T1019 & S5126	Wheelchair Maneuvering	22
T1019 & S5126	Eating	23
T1019 & S5126	Meal Planning/Preparation/Clean-Up	24
T1019 & S5126	Shopping and errands	25
T1019 & S5126	Medications/Treatments	26
T1019 & S5126	Transportation	27
T1019 & S5126	Use of Telephone	28
T1019 & S5126	Laundry	29
T1019 & S5126	Housekeeping	30
T1019 & S5126	Minor Sewing/Mending	31
T1019 & S5126	Exercises/range of motion activities	32
T1019 & S5126	Other Health Maintenance Activities	33
T1019 & S5126	Assistance in the community	34





#### ACTIVITY CODES FOR LPN (T1000) AND RN (T1000TD) SPECIALIZED MEDICAL CARE SERVICES MR/DD WAIVER

Service	Activity	Code
T1000 & T1000TD	Special Diet/Oral Feeding	80
T1000 & T1000TD	NG- tube	81
T1000 & T1000TD	G-tube	82
T1000 & T1000TD	Stoma/Ostomy Care	83
T1000 & T1000TD	Wound Care	84
T1000 & T1000TD	Decubitus Care	85
T1000 & T1000TD	Burn Care	86
T1000 & T1000TD	Complex Dressings	87
T1000 & T1000TD	Lifting/Transferring	88
T1000 & T1000TD	IV Therapy	89
T1000 & T1000TD	Central Line care	90
T1000 & T1000TD	Assessment	91
T1000 & T1000TD	Tracheostomy Care	92
T1000 & T1000TD	Oxygen	93
T1000 & T1000TD	Bi Pap/C Pap care	94
T1000 & T1000TD	Ventilator care	95
T1000 & T1000TD	Other Respiratory care	96
T1000 & T1000TD	Routine Medication Administration	97
T1000 & T1000TD	Injectable Medication Administration	98





#### ACTIVITY CODES FOR PERSONAL CARE ATTENDANT

#### SRS – TECHNOLOGY ASSISTED WAIVER

Service	Activity*	Includes	Code
T1019	Lifting	Lifting/Body Mechanics/Transfer/Position	41
T1019	Personal Care	Dressing/Bathing/Hair/Oral/Skin/Nail	42
T1019	Nutrition	Diet/Nutrition Prep/Clean-up	43
T1019	Toileting	Toileting/Diapering/Personal Adjustment	44
T1019	Housekeeping	Housekeeping/Laundry	45
T1019	Ambulation	Ambulation Technique Assistance	46
T1019	Medication	Medication Administration	47
T1019	Oxygen	Oxygen Administration	48
T1019	CPR/First Aid	SAME	49
T1019	Emergency	Emergency Procedures	50
T1019	Tracheotomy	SAME	51
	Care		
T1019	Seizure Control	SAME	52
T1019	Infection Control	SAME	53
T1019	Suction	Suction Machine Use	54
T1019	Glucometer	Glucometer Use (blood sugar monitoring)	55
T1019	Vital Signs	Vital Sign Monitoring (Temp, BP, Pulse, Pulse Ox, Resp)	56
T1019	NG	NG/GT/NJ Feeding and Care	57
T1019	Catheter	Catheter Care/Recording Input & Output	58
T1019	Enema	Enema/Suppository Insertion	59
T1019	Range of Motion	Range of Motion Exercises	60
T1019	Documentation	Documentation/Record Keeping	61
T1019	Recreation	Recreation/Socialization	62
T1019	Transportation	SAME	63
T1019	Hearing	Hearing Impaired Assistance	64
T1019	Visual	Visually Impaired Assistance	65
T1019	Communication	Communication Technique Assistance	66
T1019	Behavior Modification	Behavior Modification Technique Assistance	67
T1019	Other	SAME	68

\* Phrase stated on the IVR/listed on the web claim page.





## APPENDIX A.3: CLAIM EXCEPTION CODES

#### **CRITICAL EXCEPTIONS**

Claims with Critical Exceptions cannot be submitted to KMAP/MMIS for payment until the identified problem has been corrected.

Code	Exception	Definition	What do I need to do?
A1	Authorize	There is no authorization in KS AuthentiCare for the service, date, client, and/or provider.	Wait for the file to be loaded from KMAP/MMIS. KS AuthentiCare will automatically recalculate when a valid authorization is entered.
C5	Calculate	All data needed to calculate the claim is not available. Usually means that the check- in time or check-out time is missing. If Authorize exception occurs, it will trigger the Calculate exception as well.	Verify what information is missing. If the check in or out time is missing, see Section 12.2 for claims correction instructions.
A2	Authorization Exhausted on Claim	Indicates the authorization was exhausted (authorized units used up) <u>while</u> this claim was being calculated.	There are not enough authorized units to cover the claim. If additional authorized units are entered from KMAP/MMIS, the system will recalculate and remove this exception. If no more units are available, the provider may edit the claim (change the check in or check out time) to match the authorized units so the claim can be submitted for payment.





Code	Exception	Definition	What do I need to do?
A3	Authorization Exhausted Before Claim	All authorized units were used <u>before</u> this claim was calculated.	Contact the Case Manager to see if additional units can be added. This information will be reflected in KS AuthentiCare in the next data file transfer. KS AuthentiCare will automatically recalculate when new authorization information is entered.
12	Ineligible Worker	The worker is not eligible to deliver services based on his/her begin and end dates or active/inactive/suspend status.	If you verify that this information is correct, the claim cannot be submitted. If the information is incorrect and the provider corrects it, KS AuthentiCare will automatically recalculate if the worker's eligibility status changes.
11	Ineligible Provider	The provider is designated as inactive.	If you verify that this information is correct, the claim cannot be submitted. If corrected information is provided by file transfer from KMAP/MMIS, KS AuthentiCare will automatically recalculate.
C1	Billing Confirmation	The claim has not been confirmed for billing.	Complete billing confirmation. See Section 12.6 and 12.7 for instructions.
C2	Duplicate Claim	The claim data already exists in KS AuthentiCare system.	As a duplicate, it cannot be submitted to KMAP/MMIS. Refer to Section 12.2, claims corrections, for the steps needed to have this claim deleted.





Code	Exception	Definition	What do I need to do?
C3	Missing Activity Codes	The claim is for a service that requires the selection of an activity code but no code was entered.	Add appropriate activity code(s).
14	Unenrolled Provider Service	The claim has been filed for a service that this provider does not provide.	This only occurs with claims entered via the web if the incorrect service is erroneously entered. To correct this claim exception, either correct the service entered for the claim or, when the provider's authorized service list is updated from KMAP/MMIS, KS AuthentiCare will automatically recalculate the claim.

## INFORMATIONAL (NON-CRITICAL) EXCEPTIONS

*Informational Exceptions* or non-Critical exceptions do not prevent a claim from being processed, but serves as a notice of some problem associated with the claim creation which may warrant further investigation.

Code	Exception	Definition	What do I need to do?
E1	Event Matching	The does not match an event scheduled in the system.	No action required. You cannot enter an event for a date in the past.
G1	Unauthorized phone number – No Match – Check in	The claim was filed by checking in from a phone number that does not match the phone number on record for the client.	No action required. You may want to check with the worker to understand why the client's phone was not used. Double check the client's phone number to make sure it is correct.





Code	Exception	Definition	What do I need to do?
G2	Unauthorized phone number – No match – Check Out	The claim was filed by checking out from a phone number that does not match the phone number on record for the client.	No action required. You may want to check with the worker to understand why the client's phone was not used. Double check the client's phone number to make sure it is correct.
G3	Unauthorized phone number – Other Match – Check in	The claim was filed by checking in from a phone number that exists in the system, but isn't the phone number on record for the client named in the claim.	No action required. You may want to check with the worker to understand why the client's phone was not used. You may also want to see what other phone number in KS AuthentiCare this matches (such as another client's home). Double check the client's phone number to make sure it is correct.
G4	Unauthorized phone number – Other Match – Check out	The claim was filed by checking out from a phone number that exists in the system, but isn't the phone number on record for the client for which the claim is being filed.	No action required. You may want to check with the worker to understand why the client's phone was not used. You may also want to see what other phone number in KS AuthentiCare this matches (such as another client's home). Double check the client's phone number to make sure it is correct.





Code	Exception	Definition	What do I need to do?
15	Unenrolled Worker Service	The claim is for a service the worker is not authorized to provide.	No action required. Verify that the service on the claim is correct. If it is, then verify the services entered for the worker on the Worker page and modify as needed. Refer to Section 6.3 for further information.

KS AuthentiCare User Manual





## APPENDIX A.4: DESCRIPTION OF PROVIDER SUB-ROLES

Each provider has a broad Administrator role but there are relevant sub-roles that can be assigned to staff members with specific functions related to KS AuthentiCare and their day-to-day responsibilities.

Name	Rights	
KS_Administrator	Rights to do all functions for that provider except those functions restricted to First Data (add, edit, delete services; add/edit/delete authorizations and delete providers). Can view the Provider and Worker Dashboards. [NOTE: First Data assigns a log in and initial password for the first Administrator for the provider who can then add/manage other users (including other administrators)]	
KS_AdminAssistant		
KS_Payroll/Billing	Activities associated with billing KMAP/MMIS and using KS AuthentiCare information for employee payroll. Includes adding, editing, deleting claims as well as confirm billing. This role has primary responsibility for resolving claims with critical exceptions. Can view the Provider and Worker Dashboards.	
KS_Human	Activities associated with managing workers – adding, editing,	
Resources KS_Payroll/Billing/ Human Resources	and deleting workers and the Worker by Provider Report. Combination of Payroll/ Billing and Human Resources roles	
KS_Scheduler / Coordinator	which may be more appropriate for smaller providers Activities necessary to schedule visits for clients. Includes view and search of clients, workers, authorizations and services as these are needed to accomplish the tasks. This role will acknowledge missed visits and run Late and Missed Visit and Provider Activity Reports. This role can also view the Worker Dashboard.	
KS_Claims Mgt 1	Can add, edit and delete claims	
KS_Claims Mgt 2	Can add, edit, delete and confirm claims for submission to KMAP/MMIS	
KS_Intake & Referral	Ability to edit client information	





KS AuthentiCare User Manual